

CONSTITUTIONAL COURTS AFTER THE ARAB SPRING

Appointment mechanisms and relative judicial independence



the center for **constitutional** transitions at nyu law

Constitutional Courts after the Arab Spring: Appointment mechanisms and relative judicial independence

Center for Constitutional Transitions at NYU Law International Institute for Democracy and Electoral Assistance

Sujit Choudhry Cecelia Goetz Professor of Law, NYU School of Law Faculty Director, Center for Constitutional Transitions at NYU Law

Katherine Glenn Bass Adjunct Professor, NYU School of Law Director of Clinical Programming, Center for Constitutional Transitions at NYU Law with Bieta Andemariam, Sandra Azima, Kuntal Cholera, Lauren Katz, Cenobar Parker, Johanna Peet and Katharina Rall

Copyright 2014 Center for Constitutional Transitions and International IDEA

The electronic version of this publication (excluding the cover photos) is available under a Creative Commons License (CCI) – Creative Commons Attribute-Non Commercial-Share Alike 3.0 Licence.

International IDEA publications are independent of specific national or political interests. Views expressed in this publication do not necessarily represent the views of International IDEA, its Board or its Council members.

ISBN: 978-91-87729-40-9



The International Institute for Democracy and Electoral Assistance (International IDEA) is an intergovernmental organization with 28 member states that supports sustainable democracy worldwide. International IDEA's mission is to support sustainable democratic change by providing comparative knowledge, assisting in democratic reform, and influencing policies and politics.

International IDEA produces comparative knowledge in its key areas of expertise: electoral processes, constitution building, political participation and representation, and democracy and development, as well as on democracy as it relates to gender, diversity, and conflict and security.

IDEA's work is non-prescriptive; it takes an impartial and collaborative approach to democracy cooperation, emphasizing diversity in democracy, equal political participation, representation of women and men in politics and decision making, and helping to enhance the political will required for change.

IDEA brings together a wide range of political entities and opinion leaders. By convening seminars, conferences and capacity-building workshops, **IDEA** facilitates the exchange of knowledge and experience at global, regional and national levels.

International IDEA is a Permanent Observer to the United Nations. For more information, please visit <u>www.idea.int</u>.



the center for **constitutional** transitions at nyu law

The Center for Constitutional Transitions at NYU Law (Constitutional Transitions) generates and mobilizes knowledge in support of constitution building.

Agenda-Setting Research: Constitutional Transitions generates knowledge by identifying issues of critical importance to the success of constitutional transitions, where a lack of adequate, up-to-date research impedes the effectiveness of technical assistance for constitution building. Constitutional Transitions assembles and leads international networks of experts to complete thematic research projects that offer evidence-based policy options to practitioners.

Constitutional Transitions Clinic: Constitutional Transitions mobilizes knowledge through an innovative clinical programme that provides 'back office' research support to constitutional advisers in the field and deploys faculty experts and field researchers for support on the ground. The Clinic meets existing field missions' needs for comprehensive research, dramatically enhancing their effectiveness and efficiency in their role as policy advisers and actors.

The **Constitutional Transitions Clinic's** client for 2012–14 is the West Asia and North Africa Office of **International IDEA**, which it has supported with over **40 student researchers from 11 countries** based at NYU and stationed in Beirut, Cairo and Tunis. For more information, please visit <u>www.constitutionaltransitions.org</u>.

About this report

The Constitutional Transitions Clinic 'back office' is preparing a series of thematic, comparative research reports on issues in constitutional design that have arisen in the Middle East and North Africa. Zaid Al-Ali, Senior Adviser on Constitution Building at International IDEA, has acted as an adviser on these reports and has overseen International IDEA's participation in the report-drafting process. These reports will be jointly published by Constitutional Transitions and International IDEA in English and Arabic, and will be used as engagement tools in support of constitution-building activities in the region (e.g. in Libya, Tunisia and Yemen). The forthcoming reports are:

- Constitutional Courts after the Arab Spring: Appointment Mechanisms and Relative Judicial Independence (Spring 2014)
- Semi-Presidentialism as Power Sharing: Constitutional reform after the Arab Spring (Spring 2014)
- Political Party Finance Regulation: Constitutional reform after the Arab Spring (Spring 2014)
- Anti-Corruption: Constitutional Frameworks for the Middle East and North Africa (Fall 2014)
- Decentralization in Unitary States: Constitutional Frameworks for the Middle East and North Africa (Fall 2014)
- Oil and Natural Gas: Constitutional Frameworks for the Middle East and North Africa (Fall 2014)

The reports will be available at <u>www.constitutionaltransitions.org</u> and <u>www.idea.int</u>. An Arabic translation of the reports is forthcoming. For more information, please visit <u>www.constitutionaltransitions.org</u>.

Acknowledgements

The authors of this report wish to thank Anis Abidi (International IDEA), Aslı Bâli (UCLA School of Law), Nathan Brown (George Washington University), Jennifer Canose (NYU School of Law), Haider Hamoudi (University of Pittsburgh School of Law), Mattias Kumm (NYU School of Law), Ahmed Ouerfelli (Office of the President of Tunisia), Pasquale Pasquino (NYU), Intisar Rabb (NYU School of Law), Chafik Sarsar (University of Tunis–El Manar), Christoph Schoenberger (University of Konstanz) and Richard Stacey (NYU School of Law). The views expressed in this report, and any errors it contains, are the responsibility of the authors alone.

Many of the ideas that are set out in this report were tested in a conference that took place on 8 April 2013 at the University of Tunis–El Manar, which was attended by Tunisian academics, postgraduate law students, members of the Tunisian Constituent Assembly and legal advisers to the Tunisian state. We are grateful for the helpful suggestions offered by the conference participants.

This report has been prepared by the Center for Constitutional Transitions at NYU Law, affiliated with the New York University School of Law, but does not purport to present the school's institutional views, if any. All English-language translations of documents referred to in this report are unofficial, unless otherwise noted.

Contents

About this report	4
Acknowledgements	5
Contents	6
Executive summary	9
The legislative supermajority model	.10
The judicial council model	.11
The judiciary-executive model	.11
The multi-constituency model	.12
Qualifications	. 13
Removal	. 13
1 Introduction	.15
1.1 Constitutional review in the Middle East-North Africa region	
1.1.1 New courts in new democracies	
1.1.2 Existing constitutional courts in transition	
1.1.3 New constitutional courts under existing governments	
1.2 Functions of a constitutional court	.19
1.2.1 Upholding the rule of law	21
1.2.2 Protecting individual rights	21
1.2.3 Providing a forum for the arbitration of disputes among political parties,	
branches of government and government officials	21
1.2.4 Separation of powers	
1.2.5 Enforcing the constitutional pact	22
1.2.6 Political insurance	22
1.2.7 Symbolic value	22
1.3 Constitutional courts and politically contested cases	
1.3.1 Review of electoral laws and elections	
1.3.2 Political party regulation	
1.3.3 Enforcing the separation of powers	
1.3.4 Post-authoritarian legal reform	
1.3.5 Overseeing constitutional amendment procedures	
1.4 Relative judicial independence and political investment	
1.5 The importance of constitutional court appointments	.28
2 Constitutional court appointments: MENA region challenges	.29
2.1 Threats to the court's independence	.29
2.1.1 Judicial appointments dominated by the executive	29
2.1.2 Executive manipulation of court composition	30
2.2 Threats to the court's legitimacy: locating relative judicial independence	.31

3 Principles of constitutional design	32
3.1 The appointments process should encourage widespread participation from	
different political constituencies	32
3.2 Different actors should oversee appointment and removal processes	32
3.3 Qualifications should be set to select judges with a high level of legal expertise	.33
4 Judicial appointments: design options	
4.1 The legislative supermajority model: Germany	34
4.1.1 Germany: historical and political context	. 35
4.1.2 Germany: appointments procedure	. 36
4.1.3 Germany: appointments process in practice	. 38
4.1.4 The legislative supermajority model: key constitutional considerations	
(medium level of political investment)	
4.2 The judicial council model: South Africa	45
4.2.1 South Africa: historical and political context	
4.2.2 South Africa: appointments procedure	
4.2.3 South Africa: appointments process in practice	. 50
4.2.4 The judicial council model: key constitutional considerations (high level of	
political investment)	
4.3 The judiciary-executive model: Egypt and Iraq	
4.3.1 Egypt	
4.3.1.1 Egypt: historical and political context	
4.3.1.2 Egypt: appointments procedure	
4.3.1.3 Egypt: appointments process in practice	
4.3.2 Iraq	
4.3.2.1 Iraq: historical and political context	
4.3.2.2 Iraq: appointments procedure	
4.3.2.3 Iraq: appointments process in practice	
4.3.3 The judiciary-executive model: key constitutional considerations (low leve	
of political investment)	
4.4 The multi-constituency model: Turkey and Italy	
4.4.1 Turkey	
4.4.1.1 Turkey: historical and political context	
4.4.1.2 Turkey: appointments procedure	
4.4.1.3 Turkey: appointments process in practice	
4.4.2 Italy	
4.4.2.1 Italy: historical and political context	
4.4.2.2 Italy: appointments procedure	
4.4.2.3 Italy: appointments process in practice	
4.4.3 The multi-constituency model: key constitutional considerations (high le	
of political investment)	. 85

5 Judicial qualifications and removal procedures	89
5.1 Judicial qualifications	89
5.2 Removal procedures	92
5.2.1 Egypt	92
5.2.2 Germany	92
5.2.3 Iraq	92
5.2.4 Italy	
5.2.5 South Africa	93
5.2.6 Turkey	93
6 Analysis of the June 2013 draft Constitution of Tunisia	94
6.1 Appointments procedure and qualifications	
6.2 Removal procedure	
7 Conclusion	97
References	99
Appendix: selected constitutional and legal provisions on constitutional courts	. 110
Egypt	.110
Germany	. 112
Iraq	.116
Italy	.118
South Africa	. 120
Turkey	. 124
Endnotes	. 128

Executive summary

The Middle East-North Africa (MENA) region is experiencing an unprecedented moment of constitutional transition. Among other constitutional reforms, many countries in the region have adopted, are considering adopting or have strengthened systems of constitutional judicial review as a way of signalling the government's commitment to the rule of law. While constitutional judicial review is not new to the region, many countries have established a constitutional court—a specialist judicial body with exclusive jurisdiction over constitutional judicial review—in an attempt to strengthen the role of the courts in interpreting and enforcing the constitution. A constitutional court plays many important roles, including promoting the rule of law, protecting individual rights, providing a forum for resolving disputes, enforcing the separation of powers, holding different political players accountable to their constitutional commitments, serving as 'political insurance' for opposition parties and symbolizing the end of a period of authoritarian rule. The success of constitutional courts is closely tied to the success of constitutional democracy in the region.

Constitutional courts are often called upon to decide on a country's most pressing political issues, including questions about electoral laws and results, regulating the activities of political parties, enforcing the separation of powers among the branches of government, reforming the legal system after a period of authoritarian rule and overseeing constitutional amendment procedures. The litigants in these disputes are often political parties. Even if the cases do not frame the issues in this way, constitutional interpretation is a site of partisan political conflict among political parties, which constitutional courts are called upon to resolve.

The process of appointing judges is central to establishing or reforming a constitutional court. The judicial appointments process determines who will interpret the constitution. This report investigates how constitutional court appointment procedures can be designed to promote both judicial independence *and* judicial accountability to a democratically elected government. The fact that constitutional courts cannot avoid adjudicating disputes with partisan dimensions logically suggests that political actors should play a role in selecting constitutional court judges. Involving a wide range of political actors in the appointments process fosters a broad sense of political investment in the court, so that all actors have an incentive to continue supporting the court even when they are on the losing side of its decisions. Another reason to involve a broad variety of political actors in the appointment process is that although judges strive to interpret the law fairly and issue impartial decisions, they are only human, and a judge's political ideology will naturally play some role in how he or she views the law. The appointments process should be designed to strike an appropriate balance between the constitutional court's independence (insulating the court from political interference) and

its need to be responsive to the democratic society in which it operates. This report refers to this balance as *relative judicial independence*.

In the MENA region, the executive has historically dominated constitutional court appointments. Often, the executive controlled both appointments to the court and changes to the court's composition, adding judges to a court or removing them at will, which gave it tremendous influence over the court's judgements. Failing to involve other political actors in constitutional court appointments damaged the court's legitimacy in the eyes of political parties and the general public, and constitutional court decisions often protected the interests of the executive or the small group of elites that formed the court. Even in the midst of the Arab Spring, many of the newly formed constitutional courts in the region continue to give the executive significant, or even total, control over court appointments.

If carefully designed, the constitutional court appointments process can contribute to the formation of a relatively independent court that many different political constituencies play a role in shaping, which creates a sense of political investment in the court that encourages political actors to abide by the court's decisions, rather than challenge its legitimacy. The design of the constitutional court appointments process must be guided by three principles: (1) widespread participation from different political constituencies; (2) division of the powers to appoint and remove justices; (3) establishing qualifications to ensure the selection of judges of high legal expertise.

This report discusses and analyses four models for constitutional court appointments, with a particular emphasis on how effectively those models foster a sense of political investment on the part of multiple political actors. It examines how the four models have been applied in six countries: Germany, South Africa, Egypt, Iraq, Italy and Turkey. The report also compares the qualifications required for appointment to the constitutional court, and the rules for removing constitutional court judges, in these six countries. The four models, as well as the qualifications and removal sections, are briefly summarized below.

The legislative supermajority model

In a legislative supermajority appointments model, the legislature has primary control over the process of selecting judges. Depending on a country's political system, one or two chambers of the legislature are responsible for electing judges. A defining feature of the model is the required majority that a candidate needs for election: a supermajority. Whereas a simple majority would allow a governing party to dominate appointments procedures, a 'supermajority' of two thirds (or an even higher qualified majority) guarantees a role for opposition parties in the process. By requiring a supermajority vote to approve candidates, the judicial appointments process is intended to foster a process of negotiation and compromise between government and opposition leaders. The model is used in Germany to appoint members of the Federal Constitutional Court (FCC). It has successfully promoted a widespread sense of political investment in the FCC among political parties. However, the supermajority requirement can lead to legislative deadlock in countries with a high degree of political party fragmentation, or where the intensity of partisan conflict makes compromise on a nominee difficult. At present, no country in the MENA region uses this model for constitutional court appointments. However, Morocco's new Constitution calls for Parliament to select half of the Constitutional Court's appointees, and requires a supermajority vote to do so. Tunisia's proposed June 2013 draft Constitution also includes a provision requiring the Chamber of Deputies to select candidates, which are nominated by other political actors, by a supermajority vote.

The judicial council model

Judicial councils are created to insulate the appointments process from political actors by forming a council involving multiple political branches and, often, non-political groups such as bar associations, legal scholars and other civil society actors. This council oversees the appointments process, soliciting applications for court vacancies, interviewing candidates, and then either selecting a candidate or presenting a shortlist of candidates to the executive or legislature to make a final selection. A leading example of the judicial council model can be found in South Africa. Its Judicial Service Commission (JSC) plays a central role in appointments to the Constitutional Court of South Africa. The JSC includes executive appointees, members of both houses of Parliament, members of the judiciary, lawyers and law professors. The diversity of the JSC helps foster a sense of investment in the Court across the political spectrum, and it has largely succeeded in creating an independent Court whose decisions are widely respected. However, the continuing dominance of the African National Congress (ANC) in both the executive and legislative spheres allows the ANC to appoint the majority of the JSC's members, which may impact the Court's independence in the long term. Several countries in the MENA region have a judicial council, although not all such councils play a role in appointing constitutional court judges. The members of the Constitutional Court of Kuwait are selected by a judicial council composed of senior judges and political officials.

The judiciary-executive model

The judiciary-executive model divides the power to appoint judges to a constitutional court between the judicial and executive branches. In most iterations of the model, the judiciary (most often senior judges of the highest courts) nominates either one or a shortlist of candidates to the constitutional court. The executive must then select a candidate or approve the selection made by the judiciary, and formally appoint the judge

to the court. Other variations of this model provide that the executive nominates either one candidate or a list of candidates to the court, and the judiciary must approve the appointment. By relying on the joint consent of the judiciary and executive, the model intentionally excludes the legislature, in an effort to insulate the court from short-term political concerns. Egypt's Supreme Constitutional Court (SCC) and Iraq's Federal Supreme Court (FSC) are both appointed under variations of the judiciary-executive model. Because the model excludes many political actors from the appointments process, especially opposition political parties, the courts in both countries struggle with a low degree of political investment, leaving the courts vulnerable to accusations that their rulings are based on political loyalties. The experiences of Egypt and Iraq suggest that the constitutional court appointments process in a democracy requires the involvement of a broader range of actors than the judiciary-executive model permits.

The multi-constituency model

The multi-constituency model involves multiple institutions in the judicial appointments process, including the various branches of government and, in some countries, civil society organizations as well. In this model, the institutions involved in appointments may have direct or indirect power over them. Institutions with direct appointment power may select candidates and appoint them to the court without having to consult with or gain the approval of any other actor. Institutions with indirect power are generally given either the power to nominate one or a list of candidates for the court or to approve or veto a candidate nominated by another institution, but do not have the power to both nominate and confirm a particular candidate. Most commonly, the seats on the court are divided among the various institutions that have appointment power. In contrast to the judicial council model, the institutions that have a role in selecting the court's judges generally work independently of each other during the selection process. The Italian Constitutional Court has been appointed under a multi-constituency appointments model since 1953. The Turkish Constitutional Court's appointments model was changed from a judiciary-executive model to a multi-constituency model through constitutional amendments passed in 2010. In Italy, the multi-constituency approach has created a strong sense of political investment in the Constitutional Court, although its experience also indicates that legislative appointments may be delayed if the parties in the legislature cannot arrive at a compromise on a candidate. The model was introduced in Turkey too recently to assess its impact; however, the change was in part motivated by the desire to create a more representative and responsive Constitutional Court. Tunisia's proposed June 2013 draft Constitution uses a variation of the multiconstituency model, combined with elements of the legislative supermajority model, for Constitutional court appointments.

Qualifications

Specifying the required qualifications that constitutional court judges must hold is another way to ensure political investment from across the political spectrum. Setting out the level of education and professional achievement judges must have obtained, or specifying a minimum or maximum age at the time of appointment, ensures that the judges appointed to the constitutional court will have the expertise necessary to parse the complex and politically significant constitutional questions that will come before the court. It also creates an additional barrier to court packing: a political actor or party seeking to place its supporters on the constitutional court will have to ensure that the candidates it nominates possess the minimum qualifications specified in the constitution. Qualifications may also include a list of professions or offices that are incompatible with appointment to the constitutional court, usually political offices, which can help to insulate the constitutional court from political influence.

Removal

Another important issue for the design of constitutional courts is the procedure for removing sitting judges from the bench. Removal and appointment procedures are mutually reinforcing. If it is easy for a single political actor to remove judges, this can be used as a mechanism to circumvent even the best-designed appointments process, by enabling the manipulation of the constitutional court's membership. Indeed, the threat of removal can be used as a subtle tool of influence over the judges. To protect against this, some countries only permit the removal of a constitutional court justice if the constitutional court itself votes in favour of removal, sometimes requiring a supermajority vote. Other countries require a multi-step process to remove a judge, in which several different branches of government must approve the removal before it can be enforced.

Many countries in the MENA region are currently considering these important questions regarding the formation of a constitutional court and the appointment mechanism for its judges. The Arab Spring sparked a regional debate over constitutional reforms, providing a unique opportunity to create a strong judicial institution that can help promote the rule of law and hold all political actors accountable to the constitution.

There are two regional trends. Countries such as Tunisia have proposed a procedure for appointing constitutional court judges that will involve many different political actors, thus fostering a broad sense of political investment in the court and helping to protect the court's independence. In contrast, Jordan, Morocco and Syria have all granted the executive branch an enormous amount of power over constitutional court appointments.

If constitutional court judges fear that angering the executive may cost them their positions, their decisions may be influenced more by the need to please the executive than by the law's requirements. Without establishing procedures and rules that will allow a constitutional court to withstand political pressure, it will serve as mere window dressing for rulers who wish to give the appearance of respect for the rule of law without creating real checks on their power.

1 Introduction

The Middle East-North Africa (MENA) region is experiencing an unprecedented moment of constitutional transition. Among other constitutional reforms, many countries in the region have adopted, are considering adopting, or have strengthened systems of constitutional judicial review as a way of signalling the government's commitment to the rule of law and separation of powers. While constitutional judicial review is not new to the region, many countries have established a constitutional court, a specialist judicial body with exclusive jurisdiction over constitutional judicial review, in an attempt to strengthen the role of the courts in interpreting and enforcing the constitution.

This report aims to contribute to constitutional debates on the design of constitutional courts, with a particular focus on mechanisms for appointing constitutional court justices. The report investigates how constitutional court appointment procedures can be designed to promote both judicial independence and judicial accountability to a democratically elected government. The report begins with an overview of the history of constitutional judicial review in the MENA region, an assessment of the benefits of establishing a constitutional court and a discussion of the key role such courts play in the adjudication of politically controversial issues. The report then reviews the two core considerations that shape the design of a constitutional court appointment procedure: judicial independence and judicial accountability.

Part 2 discusses the challenges that the MENA region and Turkey have faced in determining a judicial appointments process that strikes the right balance between judicial independence and accountability. Part 3 identifies the principles of constitutional court design that should guide policymakers as they determine the court's appointments procedures.

Part 4 provides a detailed discussion and analysis of four different models for appointing constitutional court judges and how they have been implemented in six countries: (1) the legislative supermajority model in Germany; (2) the judicial council model in South Africa; (3) the judiciary-executive model in Egypt and Iraq and (4) the multi-constituency model in Turkey and Italy. For each country example, the report provides a brief overview of the country's political and historical context that is relevant to the constitutional court appointments process, followed by a description of the appointments have unfolded in practice. The discussion of each appointment model concludes with a discussion of key policy considerations that policymakers considering the model should keep in mind. Part 5 reviews the qualifications that appointees to the court must hold in each of the six countries discussed in Part 4, and the procedures used in each country to remove a constitutional court judge from the bench.

1.1 Constitutional review in the Middle East-North Africa region

The spread of constitutional courts in the MENA region follows a global wave of constitutional reforms that began in the late 1980s with the fall of the Soviet Union. The creation of a specialist court with exclusive jurisdiction over constitutional review, the task of which is to protect and enforce the supremacy of the constitution, has become an expected component of democratic transitions and democratic societies. Today, over 70 countries have a constitutional court or council.¹

Judicial review is far from a recent phenomenon in the MENA region. In 1925, Iraq became the first country in the region to adopt a constitution that explicitly mentioned judicial review (Article 81); Egypt followed suit two decades later when the Supreme Administrative Court asserted the power of judicial review in the absence of express constitutional authorization. Egypt was also one of the first countries to propose a specialized court for constitutional review, in 1979. Many MENA region countries have established constitutional review, either through a supreme court, a specialized constitutional court or a constitutional council.²

Three different scenarios are currently unfolding in the region regarding constitutional courts. Countries that have recently experienced a transition from authoritarian rule are establishing new courts, including Iraq, Libya and Tunisia; Egypt is contemplating the future of its Supreme Constitutional Court (SCC), which was established under authoritarian rule but now operates in a drastically different political environment; and several monarchies that have not experienced a regime change but are facing popular calls for political reform are considering what role a constitutional court may play in meeting those demands.

1.1.1 New courts in new democracies

Iraq['] Iraq's 2005 Constitution established a Federal Supreme Court (FSC) that has the power of constitutional review. The implementing legislation necessary to fully establish the FSC has not yet been passed by the Council of Representatives; as a result, the FSC is still operating under the Transitional Administrative Law (TAL) that came into force during Iraq's occupation. Under Article 44 of the TAL and Law No. 30/2005, the FSC consists of nine members, including the President of the Court.³ Although the FSC is not a specialized constitutional court, its experience is highly instructive for (and relevant to) the MENA region. The FSC is discussed in detail in Part 4 of this report.

Libya: During Muammar Qaddafi's rule, Libya largely lacked a functioning judicial system. While in power, Qaddafi sought both to weaken the judiciary's independence and bypass the ordinary courts entirely by creating special military and 'revolutionary' courts to try Qaddafi's opponents and handle most political and security issues. Qaddafi did preserve the Supreme Court that was established by the 1951 Constitution (Article

141), with some modifications. The Supreme Court is divided into five chambers for different areas of substantive law: civil and commercial, criminal, administrative, constitutional and sharia. In the later years of his regime, Qaddafi took steps toward judicial reform, such as removing the Justice Ministry from the jurisdiction of the security forces, as part of his rapprochement with the European Union and the United States. However, since the fall of Qaddafi, the Libyan judiciary has suffered from a lack of adequately trained staff, widespread corruption and susceptibility to political influence. During the transitional period, the Supreme Court's Constitutional Chamber has ruled on a small number of constitutional cases. The Libyan constitution-drafting process is ongoing, and it remains to be seen what reforms a new constitution will introduce for Libya's judiciary.⁴

Tunisia: Tunisia's draft Constitution of June 2013 would establish a new constitutional court. Prior to the 2011 revolution, Tunisia had a Constitutional Council, but because only the President could refer matters to the Council and he had effective control over appointments to the Council, that body lacked independence from the executive. The Council was dissolved by decree shortly after the ousting of President Zine El Abidine Ben Ali. The June 2013 draft Constitution would establish a 12-member Constitutional Court, and sets out an appointment mechanism that is a variation of the multiconstituency model (discussed in Part 4 of this report). The President, the Prime Minister, the Supreme Judicial Council and the Speaker of the Chamber of Deputies would each nominate six candidates. At least two thirds of the nominees must be legal specialists. The Chamber of Deputies would be appointed for a nine-year, non-renewable term. Appointments would be staggered so that one third of the Court's members are appointed every three years.⁵

1.1.2 Existing constitutional courts in transition

Egypt: The Egyptian Supreme Constitutional Court (SCC) was established by then-President Anwar Sadat in 1979, and was one of the few political institutions capable of standing up to the executive branch during President Hosni Mubarak's rule, at least for a time. After the 2011 revolution, the SCC has played a pivotal and often controversial role in Egypt's transition. The SCC is discussed in detail in Part 4. As Egypt's story continues to unfold, policymakers will be faced with important decisions regarding the SCC's composition and role under a new Egyptian Constitution.

1.1.3 New constitutional courts under existing governments

Jordan: Jordan's 2011 Constitution created a new Constitutional Court to determine the constitutionality of laws and regulations. According to the Constitution, the Court is comprised of at least nine members, including the Court's President. The King appoints all members of the Court (Article 58) and its decisions are final and binding on all authorities (Article 59). The Jordanian Parliament enacted the implementing legislation for the Court, the Constitutional Court Law, in June 2012. In October 2012, King Abdullah issued a royal decree appointing the Court's nine members, including its President, Taher Hekmat, who had previously served as a government minister and senator. The King cited the need for a constitutional court to review legislation and heralded it as a 'major step and milestone in the process of reform and democratic renewal'.⁶

Morocco: The 1996 Moroccan Constitution established a Constitutional Council to review all organic laws and any other laws referred to the Council prior to their enactment. In February 2011, thousands of Moroccans participated in protests calling for political reforms, a new Constitution and true constitutional monarchy. Four months later, King Mohammed VI announced a new draft Constitution, including a provision to replace the Constitutional Council with a new Constitutional Court. In July 2011, a national referendum on the proposed Constitution passed with 98 per cent of participating voters in favour, according to official results. Under the new Constitution, the Constitutional Court is comprised of 12 members who serve for a non-renewable term of nine years. The King appoints six members (including one member nominated by the Secretary General of the Ulema Higher Council) and each house of Parliament elects three members, by a two-thirds majority vote.⁷

Bahrain: Bahrain established a Constitutional Court in 2002, when it adopted a new Constitution. After anti-government protests broke out in early 2011, prompted in part by the events of the Arab Spring, the government of Bahrain launched a series of constitutional reforms intended to strengthen its Parliament and limit the power of the monarchy. On 15 August 2012 the King of Bahrain also issued a royal decree, Law 38-2012, amending a number of the provisions of Law 27-2002, which originally created the Bahraini Constitutional Court. The decree established that the Court would be composed of a Chief Justice, a Deputy Chief, and five other justices, all appointed by the King for five-year terms, renewable once.⁸

Kuwait: Kuwait's 1962 Constitution called for the creation of a Constitutional Court, but it took 11 years for Parliament to pass the necessary legislation to establish the Court. A judicial council, composed of senior judges and government officials, appoints the members of the Court. The perceived legitimacy of the Court has ebbed and flowed over time, and many view it as being under the control of the Emir. Its rulings in many politically sensitive cases have favoured the executive or have avoided a decision by dismissing the cases on procedural grounds.⁹

1.2 Functions of a constitutional court

The primary rationale behind a constitutional court rests on the notion of constitutional supremacy—the idea that the constitution rules over all branches of government and political actors, and that the constitutional court acts as the guardian of the constitution. By definition, a constitutional court is an 'organ of the state whose central purpose is to defend the normative superiority of the constitutional law within the juridical order'.¹⁰

In principle, constitutional review can take two forms: centralized or diffuse. In a centralized system, the model used by most European countries, including France, Germany and Italy, a dedicated body—a constitutional court or a constitutional council—is the only state organ with the power to make authoritative determinations on the constitutionality of a law or government action. When constitutional questions arise in cases before lower courts, they are referred to the constitutional court for adjudication. Diffuse or decentralized constitutional review, the model used in the United States, grants all courts in the judiciary the power of constitutional review. The Supreme Court is the highest court in the country, and it addresses questions of constitutionality when they arise in cases brought on appeal from lower courts.

A number of factors shape the choice between centralized and diffuse review. First, a specialised constitutional court is well suited for integration into a civil law system, which generally includes specialised courts in other areas (civil and criminal law, administrative law, etc.). Diffuse review is almost always found in common law systems, in which all levels of court, including the Supreme Court, have jurisdiction over all questions of law, including constitutional law. Given that most countries in the MENA region have civil law systems, centralized review with a constitutional court is the natural option.

A constitutional court also offers a relatively quick and definitive method of determining the constitutional validity of laws and decrees, because cases can be brought directly to it. In a diffuse system, by contrast, multiple courts may issue conflicting decisions regarding a law's validity. Only after cases have worked their way through the judicial system, i.e. when appellate courts or the supreme court make a determination, will there be a degree of certainty. Yet allowing an issue to percolate through the lower courts in a diffuse system allows a constitutional issue to be clarified and analysed by several courts, the judgements of which may assist the supreme court when it considers the issue.

Another argument in support of creating a specialised constitutional court centres on the nature of the cases such a court will hear. As discussed in Section 1.3, disputes over the constitution's provisions often involve the most sensitive political issues facing a country, including a review of the country's electoral laws and elections, the powers of the various branches of government and other questions. Decisions on these issues will have a major impact on the country's politics. Due to the political nature of constitutional cases, it is best to create a specialized body so that its members can develop expertise in the area of constitutional jurisprudence and insulate the rest of the judiciary from politicization.

Finally, many countries have established a new constitutional court during the transition from an authoritarian regime to a democratic system. Constitutional courts present several advantages in such a scenario. First, establishing a specialist court charged with interpreting the constitution and ensuring its primacy signals that the country is committed to the rule of law and is making a clear break with its authoritarian past. The court bears a special responsibility for ensuring that the constitution is applied fairly and equally to all members of society, no matter how powerful. Second, the ordinary judiciary might be considered suspect, given its function under the former regime. Policymakers may feel more comfortable entrusting the power of constitutional review to a new institution, the members of which are selected by democratic representatives. This rationale motivated, in part, the creation of the German Federal Constitutional Court (FCC) after World War II and the creation of the Spanish Constitutional Court after the fall of General Franco. Yet Kenya's recent experience shows that it is possible to address concerns regarding a politically compromised judiciary by creating a new supreme court whose judges have been carefully vetted to ensure that they are fully committed to the new constitutional order. However, this option is not easily implemented in civil law systems, which are pervasive in the MENA region.

In sum, constitutional courts are increasingly prevalent and considered a core component of new democracies, particularly in the post-Soviet era and continuing into the Arab Spring. During democratic transition processes, political actors negotiate the terms of the new democracy, which are formalized in a written constitution. The new democracy then faces the pressing question of how to enforce that constitution. Since World War II, it has become standard practice to entrust the judiciary with the responsibility of enforcing the constitutional bargain. As a result, careful thought must be given to designing the mechanism of judicial enforcement. There is a clear trend toward establishing a new constitutional court to enforce the constitution.¹¹

A constitutional court plays many important roles, including promoting the rule of law, protecting individual rights, providing a forum for resolving disputes, enforcing the separation of powers, holding different political players accountable to their constitutional commitments, serving as 'political insurance' for opposition parties and symbolizing the end of a period of authoritarian rule. Each of these functions is discussed below.

1.2.1 Upholding the rule of law

As guardians of the constitution, constitutional courts play a critical role in the broader mission of promoting and institutionalizing the rule of law. Under a system of rule of law, 'authority operates on the basis of fixed, identifiable, and predictable legal rights rather than unlimited personal discretion'. An independent and effective constitutional court can contribute to predictability, stability and accountability in the administration of justice alongside the ordinary courts, ensuring that the law is applied fairly and equally to all. It does so by enforcing the constitution against public institutions and officials, and ensuring that ordinary courts uphold legal norms against the state as well.¹²

A country's commitment to the rule of law also has other benefits. International investors often regard the existence of a properly functioning judiciary as a sign of a country's stability and investment potential: a judiciary that offers fair judgements, timely procedures and a comprehensive body of law that clearly protects private property rights is a signal that a country is ready for investment. For example, in Egypt, President Anwar Sadat established the Supreme Constitutional Court in part to demonstrate to investors that the country was committed to the enforcement of property rights.¹³

1.2.2 Protecting individual rights

Constitutional democracies are intended to ensure majority rule while protecting the rights of individuals and minorities. Courts, and in particular constitutional courts, play an essential role in upholding these rights in a constitutional democracy. Constitutional courts can provide a forum for individual citizens or minority groups to bring complaints regarding government violations of their constitutionally protected rights.

1.2.3 Providing a forum for the arbitration of disputes among political parties, branches of government and government officials

As the body charged with determining the constitutionality of all laws and government actions and upholding the rule of law, a constitutional court provides a forum for resolving disputes among different political parties, branches of government or government officials, which routinely arise in constitutional democracies (for further examples, see Section 1.3 below). Furthermore, a constitutional court's decisions are the final determination of whether a particular law or government act is permissible under the constitution. This can provide a valuable sense of finality in political disputes, thus preventing a dispute from becoming a protracted source of contention between political actors that undermines the ability of political institutions to function.

1.2.4 Separation of powers

Constitutional courts can issue an authoritative determination on the competences assigned to the different branches of government, and can determine whether one branch is usurping powers that are constitutionally granted to a different branch.

1.2.5 Enforcing the constitutional pact

Constitutional courts are often established during a country's transition from authoritarian rule to constitutional democracy. During the transitional period, various political actors negotiate the country's new constitution, which represents a statement of the country's values and system of government as well as a pact among political forces.

As part of this agreement, the negotiating parties may establish a constitutional court as a commitment to one another, and to the people, that they will uphold the terms of the constitution. If any single actor tries to violate the constitution's terms, the others can take that actor to court to enforce the constitutional pact.¹⁴

1.2.6 Political insurance

During political transitions, in which multiple political parties are vying for control of the government, each party has an incentive to create a constitutional court as a form of political insurance to hedge against the risk of electoral loss. If a party finds itself in the opposition after a future election, it can bring a case before the court to protect its own interests from abuses of authority by the new governing party, which may attempt to entrench its authority, and to constrain the policy decisions of the governing party if it cannot defeat them in the legislature.¹⁵

1.2.7 Symbolic value

A constitutional court can also have tremendous value as a symbol of a country's break with the past. In countries with a history of authoritarian rule and human rights violations, establishing a constitutional court is a concrete message that the rule of law has been established and that impunity will no longer be tolerated.

1.3 Constitutional courts and politically contested cases

Once established, a constitutional court will necessarily become involved in some of the most fraught political issues of the day. It is the court's duty to uphold the constitution, which may involve resolving questions about electoral laws and results, regulating the activities of political parties, enforcing the separation of powers among the branches of government, reforming the legal system after a period of authoritarian rule and overseeing constitutional amendment procedures. The litigants in these disputes are often political parties. Even if the issues are not framed in this way, constitutional

interpretation is a site of partisan political conflict among political parties that constitutional courts are called upon to resolve. These issues are always sensitive, and never more so than in the context of a democratic transition. Constitutional courts are frequently called upon to adjudicate on constitutional disputes that have a partisan political character. As in Section 1.4 below, the principal challenge in designing a constitutional court's appointments process is to ensure that political parties on the losing side of constitutional cases accept the court's ruling and do not respond by attacking it.

1.3.1 Review of electoral laws and elections

Constitutional courts are often called upon to determine the constitutionality of electoral laws, to play a role in the oversight of elections and to certify electoral results. Their decisions can have a dramatic impact on the electoral fortunes of political parties. Germany's Federal Constitutional Court (FCC) has issued many rulings on the country's electoral laws, which include a requirement that a political party win at least 5 per cent of the vote nationwide in order to qualify for representation in the *Bundestag*, Germany's Parliament. The FCC has upheld the 5 per cent requirement in numerous cases, often brought by minority parties that won a large percentage of the vote in a particular region or among a particular demographic but failed to meet the 5 per cent threshold, on the grounds that the threshold is intended to protect against excessive party fragmentation in the *Bundestag* in order to ensure that it can function effectively.

However, the FCC made one exception to this requirement, in advance of the first election held after the reunification of West and East Germany in 1990. In this instance, the FCC suggested that the *Bundestag* create a one-time exception to the requirement, because many of the newly formed, small East German political parties were unlikely to secure 5 per cent of the vote nationwide. The *Bundestag* complied, and for the 1990 election only, permitted small East German parties to form joint tickets in order to give them a better chance of clearing the 5 per cent threshold and securing representation in the *Bundestag*.¹⁶

In the MENA region, Egypt's Supreme Constitutional Court (SCC) has played a pivotal, and controversial, role in reviewing the country's electoral laws on a number of occasions both under authoritarian rule (where it was alleged that the governing authorities had manipulated the rules in order to shape the outcome of electoral contests) and during the transition after the overthrow of President Hosni Mubarak.

Both during Mubarak's rule and during the country's constitutional transition, the SCC invalidated multiple electoral laws as unconstitutional. Under Mubarak, the SCC issued several rulings that struck down provisions of electoral laws that it found to be biased against independent candidates. In 2012, the SCC found the electoral law that

governed Egypt's first post-revolution parliamentary elections unconstitutional, also due to its rules regarding independent candidates, and ordered the Supreme Council of the Armed Forces to dissolve Parliament (see Section 4.3.1).¹⁷

1.3.2 Political party regulation

Constitutional courts may also issue rulings that affect the formation and activities of political parties. The rules governing political parties determine who is legally eligible to contest elections, and therefore have a direct impact on the structure of democratic politics. The rules are prone to partisan abuse, because governing parties can use them to undermine their political opponents. This is a particularly fraught issue in countries emerging from authoritarian rule, where existing political parties may be institutionally weak or closely tied to the former authoritarian regime, and many new parties may rapidly form after the transition to democracy.

The actions of constitutional courts affect political parties in various ways. For example, the constitutions of some countries, including Germany, South Korea and Turkey, empower their constitutional courts to regulate (and even ban) certain political parties based on their substantive policy programmes and internal party organization. The Turkish Constitutional Court (TCC) has repeatedly, and controversially, ordered the closure of political parties—most notably those associated with political Islam and with Turkey's Kurdish minority—for violating the constitutional prohibition on political parties whose programmes and activities conflict with the principles of the democratic and secular republic or Turkey's indivisible integrity with its territory and nation (Constitution of Turkey, 1982, Article 68). As will be discussed in detail in Part 4, the TCC's recurring confrontations with Islamist-oriented parties eventually led to the adoption of extensive constitutional amendments that modified the procedure for appointments to the Court.

Constitutional courts also weigh in on party matters in the MENA region. In these cases, the bans on political party formation or political candidacies are statutory, and courts have been asked to rule on the constitutionality of those bans. The leading example is the Egyptian SCC, which has a long history of involvement in the regulation of political parties and candidates. In 1986, the SCC overturned a law that banned many opposition activists from participating in politics, paving the way for hundreds of political opponents of the ruling regime to re-engage in political activities. In June 2012, during the transitional period, the SCC overturned a law banning members of the Mubarak regime from running for election, holding that it was an unconstitutional deprivation of political rights. The ruling thus permitted Mubarak's former Prime Minister, Ahmed Shafik, to run for president. Shafik lost to Mohamed Morsi in a run-off election.¹⁸

Finally, South Africa's Constitutional Court upheld constitutional provisions that barred members of political parties from changing party affiliation after an election, a practice sometimes known as floor crossing. These provisions were originally adopted due to fears that the powerful African National Congress (ANC) would recruit members of Parliament (MPs) from smaller, minority parties to further consolidate its hold on legislative power. These protections for minority parties were later repealed through a constitutional amendment process led by the ANC, which was arguably motivated by the ANC's desire to take control of provincial legislatures controlled by opposition parties. The Court rejected constitutional challenges to this amendment, notwithstanding its partisan implications.¹⁹

1.3.3 Enforcing the separation of powers

Constitutional courts may be called upon to define or clarify the competences of the different branches of government—including what each branch is responsible for doing, and what each branch may not do, according to the constitution. Although framed as disputes about the scope of authority of different branches of government, these debates may take on a partisan character if different political parties control the different branches, or if the manner in which these powers are exercised enhances the power of the governing party relative to the opposition.

In 2011, South African President Jacob Zuma relied on the Judges' Remuneration and Conditions of Employment Act to unilaterally extend the term of the Constitutional Court's then-Chief Justice Sandile Ngcobo beyond the constitutionally mandated maximum. Zuma's actions met with significant protest from opposition political parties, which culminated in a challenge brought before the Constitutional Court. The Court unanimously held that both the relevant section of the Act and Zuma's actions were unconstitutional, because the Constitution only authorizes the South African Parliament to extend the term of a Constitutional Court justice. Although framed in terms of the separation of powers, the effect of the judgement was to shift the decision regarding the extension of the Chief Justice's term from the ANC-dominated executive to Parliament, where opposition parties could participate in an open debate on the issue.²⁰

Iraq's Federal Supreme Court (FSC) issued a controversial decision in 2010 regarding the power to introduce proposed legislation in the Council of Representatives. The 2005 Constitution's provisions on this power were somewhat ambiguous. Article 60(1) states that 'Draft laws shall be presented by the President of the Republic and the Council of Ministers.' Article 60(2) states that 'recommendations of law' may be presented by a Council of Ministers' subcommittee, or by at least ten members of the Council of Representatives. Article 61 further provides that the Council of Representatives shall enact federal laws. The FSC interpreted these articles to mean that only the executive branch has the power to introduce bills in the Council of Representatives; members of the Council of Representatives may only offer recommendations for potential legislation to the executive branch, which the executive may then choose to take up in a bill. The decision deprives the Council of Representatives of an independent power to make laws, and critics of the FSC point to the decision as evidence of the executive's influence over the Court.²¹ The decision's effect is to diminish the role of opposition parties in setting the legislative agenda.

1.3.4 Post-authoritarian legal reform

After the end of an authoritarian regime, whether brought about by the popular overthrow of a leader or a negotiated transition to democracy, an important decision must be made regarding what to do with the body of law promulgated under the previous regime. A constitutional court can play an important practical and symbolic role in reforming a country's legal system. The court is specially tasked with evaluating the constitutionality of laws, making it the institution best situated to undertake a broad evaluation of authoritarian-era laws and determine which can pass muster under the new constitution. Furthermore, the creation of a constitutional court during the transitional process sends an important message: the new democratic regime has created an institution that can ensure not only that the rule of law will be respected in the future, but also that former injustices and repressive laws will be relegated to the past.

The constitutional courts created in Germany and Italy after the end of World War II played a critical role in marking a legal break with the horrors of the past. Both courts set about reviewing and dismantling the prior authoritarian legal regimes shortly after their establishment. For example, the Italian Constitutional Court's first ruling was a firm statement that not only did it have the authority to review laws passed before its creation, but also that it intended to strike down many of those laws, including the Fascist-era security laws that enabled so much repression of political activity and free expression. In its early years, Germany's FCC strengthened its moral authority by working to eliminate the remnants of the old regime. For example, in the 'Beamtenverhältnisse' case, the FCC ruled that the public officials who held office during the Third Reich were not entitled to stay in office in the new Federal Republic of Germany.²²

1.3.5 Overseeing constitutional amendment procedures

Constitutional courts also play an important role in the process of amending a constitution. They may be called upon to ensure that a constitutional text sufficiently protects the interests of minority political parties, particularly if a single political party holds a clear majority in the legislature.

The Constitutional Court of South Africa was given a central role in overseeing the creation of the new South African Constitution. During the multiparty negotiation process that ended apartheid, an interim Constitution was created in 1993 to govern the country until a final Constitution was adopted. This interim Constitution included a set of constitutional principles with which the final Constitution, to be drafted by a Constitutional Assembly elected in 1994, would have to comply. The interim Constitution assigned the newly established Constitutional Court the task of determining whether the draft Constitution faithfully adhered to the negotiated general principles set out in the interim Constitution. In July 1996, the Constitutional Court received the draft Constitution for review and issued a ruling two months later in what is known as the Certification Decision. The Court refused to certify the draft Constitution, despite the fact that it was adopted by 86 per cent of the democratically elected Constitutional Assembly. According to the Court, the Constitution did not fully comply with the interim Constitution's principles. This sent a strong message that the post-apartheid judiciary's only duty was to uphold the rule of law, even when doing so was politically inconvenient or unpopular. However, the Court was also aware of the political implications of its decision, and made sure to issue a narrow ruling that endorsed the overwhelming majority of the Constitution and clearly identified the few issues that needed to be addressed to satisfy the interim Constitution's principles. These issues included better safeguards for the independence of the auditor-general and public protector (ombudsman), a more stringent process for amending the Constitution's Bill of Rights, and more clarity on the structure and functions of the provincial governments. The deficiencies identified by the Constitutional Court were united by a concern that the ANC would command legislative majorities for the foreseeable future, and therefore could abuse its power against opposition parties without sufficient constitutional safeguards. Ultimately, the Constitution was amended by the Constitutional Assembly and approved following a favourable second review by the Constitutional Court.23

1.4 Relative judicial independence and political investment

In order to maintain its authority and public legitimacy, a constitutional court must be perceived as independent. Judicial independence, however, is a highly contested concept. In essence, an independent court is one that is sufficiently insulated from political interference and control for its decisions to demonstrate the court's mandate to uphold the rule of law, rather than suggest its allegiance to a particular political official or party. Judges who fear retribution for unfavourable decisions, or whose decisions unjustifiably favour a particular political actor, cannot carry out their duty to administer justice. The constitutional court's decisions must be based on legally defensible arguments. Finally, various political actors and the public must perceive the court as independent in order to retain its legitimacy and support.²⁴

However, a constitutional court operating in a democracy is also expected to be accountable to the public, as are the other branches of government, all of which operate with the consent of the governed. If the court has the power to strike down laws promulgated by representatives of the public, then the public must have some role in deciding who will sit on that court. A court whose decisions do not in any way reflect society's values and concerns is unlikely to retain the public's support, and will either come to be disregarded as an unjust body or dismantled by the elected branches. The same holds true for political parties. If the court has the power to issue constitutional decisions that affect the power, status and rights of political parties, then parties should have some role in appointing members of that court. Moreover, the court will become politically irrelevant if its judgements consistently stray far beyond the mainstream of the political spectrum.

Constitutional drafters creating a constitutional court must therefore strive for relative judicial independence: a court that operates independently of the influence of political concerns, while remaining responsive to a democratic society. This is a difficult balance to strike. A core component of relative judicial independence is a sense on the part of all political actors that they are politically invested in the court and its decisions, so that when the court issues decisions that are unfavourable to some groups, they will accept the decision rather than attempt to undermine (or even destroy) the court's legitimacy.

1.5 The importance of constitutional court appointments

The question of how to appoint constitutional court judges is central to the establishment of a well-functioning, independent court. Judges strive to interpret the law fairly and issue impartial decisions. Since constitutional courts cannot avoid adjudicating disputes with partisan dimensions, political actors should play a role in selecting constitutional court judges. Involving a wide range of political actors in the appointments process fosters a broad sense of political investment in the court, so that all actors have an incentive to continue supporting the court even when they are on the losing side of its decisions. Another reason for an appointment process that involves a broad variety of political actors is that although judges strive to interpret the law fairly and issue impartial decisions, they are only human, and a judge's political ideology will naturally play some role in how he or she views the law.²⁵

This report analyses four of the most commonly used models for constitutional court appointments, with a particular emphasis on how effectively these models foster a sense of political investment on the part of multiple political actors. Part 2 discusses the challenges that the MENA region and Turkey have faced with respect to the appointments process and striking the right balance between judicial independence and accountability. Part 3 identifies the principles of constitutional court design that should guide policymakers as they determine the court's appointments procedures.

2 Constitutional court appointments: MENA region challenges

This report presents options for creating an appointments process that can promote an independent and accountable constitutional court in the MENA region. To do so, it is necessary to identify some of the region's recurring challenges related to judicial independence and accountability.

2.1 Threats to the court's independence

2.1.1 Judicial appointments dominated by the executive

A constitutional court whose judges are selected solely by the executive, without the participation of any other political or civil society actors, stands little chance of being able to act independently. First, the executive will attempt to capture the court by selecting judges believed to be sympathetic to the executive's policies, in order to insulate itself from constitutional accountability. Second, the judges on the court, knowing that they owe their positions solely to the executive, will likely be unwilling to issue a ruling that the executive would oppose.

In the MENA region, the executive has historically exercised strong control over appointments to constitutional courts. In Egypt (discussed in detail in Part 4) President Mubarak was able to rein in the formerly assertive SCC by exerting his legal authority over court appointments, in particular the Chief Justice. In Kuwait, the Higher Judicial Council—which includes a representative from the executive branch, the Minister of Justice—technically controls the appointment of justices to the Constitutional Court. Although the Council includes members of the judicial posts. As a result, the Emir exercises significant influence over the Court in practice. This is reflected in the Court's jurisprudence, which generally either avoids ruling on controversial subjects or does so in a way that supports the Emir's interests.²⁶

Morocco's 2011 Constitution, written by a group of experts selected by King Mohammed IV and approved in a popular referendum, establishes a Constitutional Court of 12 members. The King appoints six members, and each house of Parliament elects three of the remaining six members by a two-thirds majority vote (Article 130). While this procedure divides the appointment power between the King and Parliament, it still gives the King the power to appoint half of the Court's members—which in effect allows the judges appointed by the King to block any decision with which he disagrees.²⁷

In Syria, the President appoints all justices on the newly formed Supreme Constitutional Court for a renewable period of four years (Constitution of Syria, 2012, Articles 141, 143). The appointments process mirrors that of the Syrian Supreme Constitutional Court created under the 1973 Constitution (Articles 139, 141). Under this process, the executive appoints the Court's members, who rely on continued executive support for their tenure; thus the judges have little incentive to exercise any measure of judicial independence.²⁸

Similarly, in Jordan, the King appoints all nine members to the recently established Constitutional Court, including its President. The current President, Taher Hikmat, is a former senator who previously served in the government and as President of the Court of Cassation and the Higher Judicial Council. The King's complete control over Court appointments will likely influence its decisions, although it is too early to make a fair assessment of the Court's jurisprudence.²⁹

Allowing any single political group or branch of government to dominate a constitutional court's appointments process is generally not recommended, with the possible exception of appointments controlled by a democratically elected legislature (i.e. the legislative supermajority model). The power to remove constitutional court justices is equally important. Constitutions that are silent on removal provisions—or constitutional provisions that permit judges to be removed relatively easily or unilaterally by the executive or another government actor—leave judges vulnerable to the influence of political actors.

2.1.2 Executive manipulation of court composition

If there are no firm standards set out in a country's constitution or ordinary laws defining the number of judges on the constitutional court, or if the constitution or other legislation that sets out the number of justices is easy to amend, then the executive may be able to change the number of judges on the court essentially at will if the appointments process is largely in its hands. The executive can add judges to the bench to ensure that a majority will always rule in its favour or force an unpopular judge off the bench. Egypt's SCC has experienced both forms of manipulation of court composition. During President Mubarak's rule, after the SCC declared a regime-crafted election law unconstitutional, Mubarak appointed a new Chief Justice known as a regime loyalist, who promptly appointed five new judges to the SCC, increasing the total number of judges on the bench by 50 per cent. This change in the Court's composition transformed the SCC from an institution that to some extent offered a real check on executive power into a body that was effectively under executive control. After the fall of Mubarak, the Egyptian Constitution passed in December 2012 reduced the number of judges on the SCC to 11 from the previous 19 (including the Chief Justice). This necessitated the removal of the eight most junior judges. One of those removed was a judge who had been an outspoken critic of the Muslim Brotherhood. While it is unclear whether the provisions in the 2012 Constitution were written with the purpose of removing this judge from the Court, this episode illustrates the ways in which

expanding or contracting the number of judges on the constitutional court can be used to further political ends. $^{\rm 30}$

2.2 Threats to the court's legitimacy: locating relative judicial independence

Achieving relative judicial independence—i.e. an appropriate balance between judicial independence and public accountability—is extremely difficult. It is perhaps easier to discuss cases in which a constitutional court has demonstrably failed to achieve this balance than to provide examples of a court that has done so successfully. When the process of selecting a constitutional court's judges fails to reflect the need to secure political investment from a broad array of the political spectrum, the resulting court may be made up of judges who are supported by a particular political party; therefore the rulings may reflect that party's values and interests while ignoring those of other parties.

A court that consistently rules against majority interests in favour of minority groups, such as the military or political elite, is likely to provoke public frustration and anger. The judiciary is essentially defenceless: it depends on the goodwill of the government to comply with its decisions and relies on a measure of public support for (or at least tolerance of) the court's rulings. While widespread public support does not seem to be necessary to the survival of a court, widespread disapproval of a court can be very damaging, and sometimes fatal, to the court's existence. The history of the Turkish Constitutional Court (TCC), discussed in detail in Part 4, demonstrates this point. The formation of the TCC (including its appointments process) was dominated by the military, which sought to create an institution that would protect its interests and uphold its values. As a result, the TCC consistently issued decisions on controversial subjects, such as the closure of political parties with a religious affiliation, that upheld the preferences of the military elite even where popular opinion diverged (in one case considering the closure of an Islam-affiliated party, the AKP, after that party had won a majority in parliamentary elections). Eventually, the AKP won power through elections and was able to secure the passage of a series of constitutional amendments that changed the appointments process for the TCC and other important aspects of its structure.

Iraq's Federal Supreme Court, also discussed in Part 4, was created during a fraught period of foreign occupation by a small group of constitution drafters that was not representative of the broader Iraqi society. Because it was formed under a nondemocratic process, and since the executive still largely dominates the appointments process, critics of the Court allege that it lacks independence and credibility and that its decisions are unfairly favourable to the executive.

3 Principles of constitutional design

The MENA countries, like most countries that have adopted a constitutional court, have faced challenges in determining a procedure for appointing constitutional court judges that secures the court's relative judicial independence as well as political investment in the court from a broad array of political parties. In part, this entails protecting the court from executive capture and promoting its accountability to a democratic government. Below, a series of principles for designing constitutional court appointment processes is discussed, including the need for widespread participation from various sectors of politics and society in the judicial selection process, the advisability of entrusting the power of appointment and the power of removal to different political actors, and the importance of specifying the qualifications that constitutional court judges must hold.

3.1 The appointments process should encourage widespread participation from different political constituencies

Courts that are appointed by a single political actor, particularly the executive, are at high risk of being unduly influenced by that actor. Appointments processes that are dominated by only a few actors may stand a better chance of creating a court that operates independently of external influence, but since they still exclude many segments of the political spectrum from the selection of judges they may fail to create a broad sense of political investment in the court. An appointments process that instead aims to be inclusive of all interests by engaging different branches of government, political parties, civil society organizations, legal academia, bar associations and similar groups in some element of judicial selection is best able to create a court that represents society and that is supported by many different political interests. This engagement can take many different forms, including public consultation processes, inviting nominations from various sectors, making the appointments process as transparent as possible at all levels of the process, allowing a particular group to appoint a certain number of judges on the court, allowing a group to veto appointments made by others, or tasking one group with nominating a set number of candidates and another with selecting a certain number of judges from the nominations provided. The unique political and social context of a country will determine the best ways to promote inclusion in appointments processes and foster political investment in the constitutional court.

3.2 Different actors should oversee appointment and removal processes

If the same political actor or institution is responsible for both selecting and removing constitutional court judges, that actor will most likely be able to unduly influence the court. Judges will be aware that they owe their jobs to this entity. Therefore the powers

to appoint and remove judges from the constitutional court should be granted to separate political actors or institutions. As discussed in Part 5, some countries only permit the removal of a constitutional court justice if the court itself votes in favour of removal, which sometimes requires a supermajority vote. Other countries require a multi-step process to remove a judge, in which several different branches of government must approve the removal before it can be enforced.

3.3 Qualifications should be set to select judges with a high level of legal expertise

In addition to carefully constructing an appointments procedure that involves a wide range of political constituencies, policy makers can help ensure that the constitutional court will act independently and issue decisions on firm legal grounds by delineating the specific qualifications that constitutional court judges must hold. As will be reviewed in Part 5, many countries require that judges appointed to the constitutional court have served for a number of years as a judge prior to their appointment and have attained a certain level of legal education. Some countries also specify a minimum age that judges must be in order to be appointed or a maximum age beyond which judges are not eligible for appointment. Finally, some countries do not allow judges to hold particular offices or professions (e.g. political positions) while they are on the bench because those offices are seen as incompatible with judicial independence.

4 Judicial appointments: design options

The process of appointing judges is central to establishing or reforming a constitutional court. The judicial appointments process determines who will interpret the constitution, so the institution(s) or actor(s) with the power to appoint judges to a constitutional court wield considerable influence over it. An appointments process that promotes a sense of political investment in the constitutional court by involving a wide range of political actors stands a better chance of achieving relative judicial independence: the balance between independence and accountability.

Generally speaking, appointments processes are divided into two categories: unitary and mixed. Under a unitary model, one branch of government controls the judicial appointments process; the mixed model distributes the appointment power among multiple branches of government and in some cases involves civil society organizations. This section introduces four models for constitutional court appointments and describes the experiences of six countries in implementing those models: 1) the **legislative supermajority model** in Germany; 2) the **judicial council model** in South Africa; 3) the **judiciary-executive model** in Egypt and Iraq and 4) the **multi-constituency model** in Turkey and Italy. The legislative supermajority model is the only unitary model considered in this report. The remaining three are mixed models.

Each country example includes a brief overview of the country's political and historical circumstances that led to the decision to implement a certain type of appointments model, followed by an analysis of how constitutional court appointments have been made in practice. The discussion of each appointment model concludes with a discussion of key policy considerations that constitutional drafters examining that model and attempting to draw lessons from it should bear in mind, including an assessment of the level of political investment promoted by the model.

4.1 The legislative supermajority model: Germany

In a legislative supermajority appointments model, the legislature has primary control over the process of selecting judges. Depending on a country's political system, one or two chambers of the legislature are responsible for the election of judges. A defining feature of the model is that a candidate needs a supermajority for election. Whereas a simple majority would allow a governing party to dominate appointments procedures, a 'supermajority' of two thirds (or an even higher qualified majority) makes it more likely that opposition parties will play a meaningful role in the process. By requiring a supermajority vote to approve candidates, the judicial appointments process is intended to foster a process of negotiation and compromise between government and opposition leaders. Countries using the legislative appointments model may choose to delegate the process of nominating and negotiating over candidates to a smaller group of legislators, sometimes constituted as an official, permanent parliamentary subcommittee or an informal ad hoc working group. 31

The legislative supermajority model is used to appoint one of the most admired and well-known constitutional courts in the world, the German Federal Constitutional Court (FCC). Germany's transition to democracy following World War II emphasized the supremacy of the Constitution and the rule of law. As in other post-authoritarian transitions, the government's institutions were designed to establish a new constitutional order and prevent the return of dictatorship.

4.1.1 Germany: historical and political context

One of the key elements of Germany's Basic Law, its post-World War II Constitution adopted in 1949, was the creation of the FCC. In 1948, a meeting of constitutional experts was convened to draft a Constitution for the new state of West Germany. The draft produced by these experts was then reviewed and amended by a constituent assembly, the Parliamentary Council (composed of elected representatives from West Germany's regional state legislatures) and eventually became the Basic Law. The Allied powers overseeing Germany's post-Nazi era transition and the German Constitution drafters all agreed on the general desirability of judicial review. This principle was not new to Germany; the Weimar Constitution of 1919 had established a State High Court, which exercised a limited form of judicial review, and the post-war drafters sought to build on some of the old Constitution's democratic framework. The drafters wanted 'a clean and unequivocal break with the Nazi experience', and saw a constitutional court as a way to guarantee the enforceability of fundamental rights. The drafters created the FCC (*Bundesverfassungsgericht*) to serve this purpose, and as the institutional guardian of the Basic Law.³²

Both the drafters' conference and the Parliamentary Council debated aspects of the new FCC, in particular the Court's composition and areas of competence. However, the details of the procedures for appointing judges to the FCC were only partially addressed in the Basic Law, which provides that half of the judges of the Court shall be appointed by each chamber of the German legislature: the directly elected *Bundestag*, the lower chamber, and the *Bundesrat*, the upper chamber elected indirectly by the German *Länder* (states) (Article 94). The Basic Law does not indicate the required majority needed to make an appointment or the total number of judges on the FCC. Furthermore, although the Basic Law stipulates that the FCC should consist of federal judges and other members, it does not indicate the proportions of each. These questions were left to the new German legislature to decide when drafting the implementing legislation for the FCC.

The German legislature ultimately passed this implementing legislation, the Federal Constitutional Court Act (*Bundesverfassungsgerichtsgesetz*, hereinafter FCCA), in 1951. Both of the major parties represented in the legislature, the then-governing Christian Democratic Party (CDU) and their opposition, the Social Democratic Party (SPD), were convinced that a broad agreement on the Court's design was necessary so that the FCC would have a firm foundation on which to begin working, and legislators strove to find consensus while debating the FCCA. However, they took divergent positions during the FCCA drafting process on a number of specific issues. The CDU proposed a large Court of 24 judges, of whom half should be selected from federal judges and given lifetime tenure on the FCC, while the others would serve six-year terms. The CDU also proposed that the entire *Bundestag* should be involved in electing its half of the FCC.³³

In contrast, the SPD proposed that the *Bundestag* form an Electoral Committee composed of eight members, with parties represented in proportion to their standing in the *Bundestag*, which would choose judges on behalf of the entire chamber. The SPD believed that delegating appointment powers to a separate committee would better facilitate agreement among political parties and prevent the ruling party from dominating the process. The SPD also proposed requiring a supermajority vote for appointments made by the Electoral Committee and by the entire *Bundesrat*, arguing that this would also help preclude the possibility that any one party could dominate the appointments process: the Electoral Committee would elect judges by a three-fourths supermajority and the *Bundesrat* would use a two-thirds supermajority. In addition, the SPD feared that the existing German judiciary—which was generally thought to be affiliated with the CDU, and which included officials who had served under the Nazi regime—would dominate the new Court. Therefore it also advocated a smaller Court of ten judges, four of whom would be selected from the high federal courts to serve seven-year terms, while the six other judges would serve five-year terms.³⁴

After extensive negotiation, the final FCCA struck a balance between the demands of the CDU and SPD. In its original form, the FCCA created an FCC with 24 judges, with four seats on each of the two senates reserved for high federal court judges. The appointment procedures prescribed for each legislative chamber were different: the *Bundestag* would select 12 judges via an Electoral Committee, with a three-fourths vote required to appoint a judge, whereas the *Bundestat* would appoint 12 judges by a two-thirds vote (some FCCA provisions were later amended; see Section 4.1.2). The FCCA passed by an overwhelming majority in February 1951 and the Court became operational later that year.³⁵

4.1.2 Germany: appointments procedure

The FCC consists of two independent senates, or panels, with mutually exclusive jurisdiction and personnel. The twin-senate structure was originally intended to assign

specific sets of constitutional issues to each senate. The first senate focused on matters of constitutional interpretation, including constitutional complaints brought by citizens and cases referred from other courts. The second senate addressed questions involving constitutional organs (e.g. disputes between branches of government), contested elections, impeachment procedures and the regulation of political parties. However, this structure was substantially reformed in 1956, after it became clear that the two panels faced unequal workloads.³⁶

Initially, each senate included eight members with renewable eight-year terms, plus four career judges who held lifetime tenure, for 24 judges in total. However, both the number of judges and their term lengths have changed over time. The number of justices was first reduced in 1956 from 24 to 20, and then again in 1963 from 20 to 16 (discussed in Section 4.1.3). In 1970, the FCCA was amended to introduce nonrenewable terms, which was seen as beneficial for judicial independence. Today, eight justices sit on each senate for non-renewable terms of 12 years. At least three of the justices on each FCC senate must be selected from one of Germany's five federal supreme courts, and must have served there for at least three years prior to being appointed to the FCC. For each of the FCC's two panels, half of the judges on the panel (four) are elected by the Bundestag and half by the Bundestat. Once either chamber selects a candidate, the President formally appoints him or her to the Court (FCCA Sections 2-10). Germany's Ministry of Justice also plays a small role in appointing judges to the FCC by providing two lists of potential candidates to the Bundestag and Bundestat: one list containing the names of all sitting federal judges who meet the criteria for appointment to the FCC and another containing the names of all candidates nominated for election by political parties (FCCA Section 8). However, the legislature is not required to select FCC candidates from these lists, and in practice these lists have not been decisive.37

Appointment of FCC judges: Bundestag

The *Bundestag* appoints judges indirectly through a 12-person Electoral Committee, which holds closed sessions. The members of the Electoral Committee are elected by a party list system, so that opposition parties are proportionally represented on the Committee (FCCA Section 6). Each party can present its own list of potential committee members or compile a list along with its fellow coalition parties. In practice, most elected members are legal experts, party leaders or other highly regarded politicians. Once elected, the Committee's members hold their positions until the end of the parliamentary term, which usually lasts four years. The Committee originally elected judges by a three-fourths majority, but now does so by a two-thirds majority vote, which means that at least eight Committee members must vote in favour of a candidate for that candidate to be appointed to the FCC (FCCA Section 6).³⁸

Appointment of FCC judges: Bundesrat

The *Bundesrat* selects judges as a plenary body, with a two-thirds majority vote required to select a judge. In practice, a special *ad hoc* advisory committee agrees upon candidates before the full plenary vote. This committee, mainly composed of the justice ministers of the federal states and other selected party officials, conducts negotiations on judicial candidates ahead of the vote, so that the success of the candidate is generally assured before the official *Bundesrat* vote. This advisory committee also coordinates with the *Bundestag*'s Electoral Committee to avoid duplicate nominations.³⁹

Informal appointment procedures

Because German politics has been dominated by two major political parties, the CDU and the SPD, appointments to the FCC tend to follow a *de facto* partisan quota system. There is an informal agreement that each major party will appoint half of the judges to each of the FCC's two senates. Moreover, the two parties subdivide 'their' seats on each senate between three judges who are also party members and one judge who is considered politically neutral (i.e. not a political party member; generally, these judges are law professors). Over the years, two smaller political parties, the Free Democratic Party (FDP) and *Bündnis 90/Die Grünen* (Greens), have managed to secure representation on the FCC through alliances with one of the major parties. Usually, this means that the CDU occasionally defers its right to nominate a candidate to the FDP, and the SPD defers its right to the Greens. However, other minority parties, in particular the socialist party *Die Linke*, are still excluded from the appointments process due to their lack of affiliation with the major parties.⁴⁰

Despite this *de facto* quota system, the requirement that each nominee be approved by a two-thirds supermajority vote in the relevant legislative chamber serves as an important moderating influence on the parties that select candidates for nomination. The supermajority requirement helps to protect minority interests and continues to ensure that no candidate holding extremely left- or right-wing political views is appointed to the FCC.⁴¹

4.1.3 Germany: appointments process in practice

The FCC is widely considered to be a model for constitutional courts in new democracies, as well as an important factor in the success of Germany's post-war democratization. After 60 years in operation, the FCC's appointments process functions smoothly, thanks to both formal and informal rules for how the political parties represented in the German legislature select the Court's judges. However, in its early years (particularly the 1950s) the FCC faced several challenges to its independence, and the appointments process required amendment. A review of the FCC's early history

supplies useful insights about the use of the legislative supermajority model in practice, and lessons for policymakers who are considering adopting this method.

While the first set of appointments to the FCC was relatively uncontroversial, once created, the status of the FCC was hotly debated. The first post-war government, headed by Chancellor Konrad Adenauer of the CDU, repeatedly criticized the Court for its alleged interference in politics. Adenauer's frustration with the FCC stemmed from its early decisions, which held that some of the most important policies of his CDU government were unconstitutional, including Germany's post-war rearmament and the introduction of a government-controlled TV station. Even while criticizing the Court, however, Adenauer's government recognized the important role it would play in political disputes, and sought to use the Court to its advantage.⁴²

In the early 1950s, the CDU and the opposition SPD were engaged in a dispute over the European Defence Community Treaty. The CDU advocated joining the treaty, which would have required a defence contribution by the German military, because German rearmament was a core component of Adenauer's foreign policy agenda. The SPD opposed the treaty both because they thought it would make the reunification of West and East Germany nearly impossible, and because they argued that its provisions demanded too many concessions from West Germany.⁴³

Both the CDU and the SPD looked to the newly formed FCC for support, launching several petitions in short succession. First, in February 1952 the SPD requested an abstract review proceeding in the First Senate of the FCC to assess whether the Basic Law permitted the military defence contribution that the treaty would require. At the time, it was widely believed that 'the Court's First Senate was dominated by judges loyal to the SPD, while the Second Senate was believed to be controlled by judges favourable to the CDU'. The CDU, fearing that the First Senate would rule in the SPD's favour, also sought the FCC's help: at the request of the CDU, in June 1952 West Germany's President asked the plenary FCC for an advisory opinion regarding the constitutionality of the treaty. The CDU hoped that because the Second Senate was dominated by judges loyal to the CDU, a plenary session of the FCC would yield a majority in favour of the CDU, thus ending the SPD's suit.⁴⁴

The First Senate rejected the SPD's petition on the grounds that it could not review a treaty that had not yet been ratified. However, the SPD intended to re-submit its challenge to the First Senate as soon as the treaty was ratified. The CDU attempted to block the SPD from doing so: in December 1952, immediately after the treaty's ratification by the *Bundestag*, the CDU filed another FCC petition, this time before the Second Senate (seen as sympathetic to the CDU), to prevent the SPD from challenging the constitutionality of the treaty in the First Senate.⁴⁵

Although it was well known that the FCC justices held differing opinions on the constitutionality of the treaty, the justices recognized that both the CDU and the SPD were trying to 'forum shop' in a way that would damage the FCC's reputation, especially if the First and Second Senates issued conflicting opinions on the treaty. In response to this concern, the FCC issued a plenary resolution on 8 December 1952 stating that an advisory opinion by the plenary Court would bind both of its Senates. The FCC's President, Höpker Aschoff, went a step further the following day, when he opened the FCC's proceedings by reading a statement that offered a stinging rebuke to both the CDU and SPD, affirming that all future advisory opinions offered by the plenary FCC would be binding on both Senates and explicitly stating that the FCC had created this rule to prevent 'attempts to target a suit to a particular senate's jurisdiction for exogenous and irrelevant reasons'. The FCC also made it publicly known that 20 of its 22 sitting justices had supported the creation of the rule, in a show of FCC unity.⁴⁶

Realizing that the strategy of seeking a plenary opinion from the FCC on the constitutionality of the treaty would fail, the CDU withdrew its request, but continued to pursue the case it had filed with the Second Senate. Adenauer and Justice Minister Thomas Dehler also went on the attack against the FCC, publicly claiming that the resolution regarding the binding nature of plenary advisory opinions had no legal basis, and stating that the government would not accept the decision. Both Adenauer and Dehler suggested that the FCC's structure might be changed by revising the FCCA.⁴⁷

However, the CDU's attack on the FCC was short-lived: after Adenauer and Dehler's remarks were widely criticized in the German media, including by outlets sympathetic to the CDU, the government realized that its attack of the FCC was generating a massive public backlash and quickly took steps to reaffirm its support of the FCC. It dropped all references to altering the FCCA, though the CDU would renew its efforts to change the FCC a few years later. Meanwhile, the treaty became a moot point after France failed to ratify it.⁴⁸

At the same time the dispute over the European Defence Community Treaty was playing out before the FCC, the SPD and the CDU were also in conflict over the appointment of new judges to the Court. When two vacancies arose on the Court in the early 1950s, the inability of the opposition SPD and the governing CDU to nominate candidates of whom the other party also approved—which was necessary to obtain a supermajority vote of approval for those candidates—led to lengthy delays in filling those vacancies. During this period of tension between the CDU and the FCC, as well as between the CDU and the SPD, the CDU government began to question the FCCA's appointments procedures. The CDU sought to reduce the total number of judges on the FCC, in part because several of the judges whose terms were up for renewal in 1956 were believed to be opposed to the government's proposals for German rearmament. It also wanted to reduce the three-quarters supermajority vote required for the *Bundestag*'s Electoral Committee to elect a judge to the FCC to a simple majority, hoping that this would allow a CDU-led coalition in the *Bundestag* to overcome SPD opposition to its nominees. The SPD opposed these proposals, accusing the CDU of trying to pack the Court. The CDU also advocated reducing the total number of FCC judges to 12; eliminating the two-senate system; and abolishing the legislative appointments procedure in favour of a judicial council composed of the presidents of federal high courts, law professors and presidents of state constitutional courts, which would compile lists of nominees when vacancies arose, and from which the *Bundestag* and *Bundesrat* would make appointments. Because both the SPD and the *Bundestat* opposed the CDU's proposals, the CDU backed down, but raised the issue again a year later.⁴⁹

Ultimately, the German legislature passed a compromise plan in 1956 in which the CDU and SPD agreed to lower the number of judges in stages: from 24 to 20 immediately, and then from 20 to 16 in 1963. The plan also reduced the supermajority requirement for the selection of judges by the Bundestag's Electoral Commission from three quarters to two thirds, which made arriving at a compromise agreement easier while still protecting against court packing by one party. The agreement to reduce the number of judges on the FCC appeared to be driven less by the government's desire to eliminate judges who did not support its programmes than by a general concern for the Court's administrative efficiency. Following this compromise, both the CPU and SPD began to nominate more centrist candidates to the FCC, which resulted in more judges being elected unanimously. While the compromise agreement did not eliminate tensions between the two parties or between the government and the FCC, controversies over FCC appointments became less common, and the informal agreement between the CDU and the SPD to divide FCC appointments between them solidified. A final amendment to the Court's appointments procedure was carried out in 1970, which limited all FCC judges to one non-renewable term of 12 years. This was done primarily to eliminate any suspicion that the German legislature could exert undue pressure on FCC judges, since prior to this amendment the German legislature was responsible for deciding whether or not to re-elect FCC judges after their eight-year term ended.50

After its turbulent beginnings in the 1950s and early 1960s, the FCC increasingly managed to isolate itself from party politics and strengthened its authority as an independent judicial body. In the following decades, the FCC issued many decisions that have been unpopular with both the government and the opposition, particularly regarding challenges brought by individual citizens alleging violations of their constitutional rights. Public support for the FCC, echoed and aided by the independent German media's support for the Court, has played a critical role in discouraging the executive's attempts to interfere with the FCC. As the Court's popularity with the public increased, Germany's political elites became far more reluctant to openly question its legitimacy, limiting themselves to criticizing the merits or outcomes of particular cases. Indeed, the public's respect for the FCC occasionally prompts political parties to try to bolster their own popularity by claiming that its decisions validate their own policies; the government and opposition parties sometimes both claim that the same FCC decision supports their competing positions.⁵¹

Similarly, the process of appointing judges to the FCC moves relatively smoothly in most cases, thanks to the informal agreement between the CDU and the SPD on dividing seats on the Court between them. Even when the parties disagree over specific nominations, they usually keep the identities of nominees private until a compromise is reached. For the few nominations that are controversial enough to make headlines, the problems centred on the potential candidate's political views and ideological differences between the parties. For example, in 1993, the CDU opposed the SPD-nominated candidate Herta Däubler-Gmelin because she was an SPD party member and had a reputation for holding strong ideological opinions, for example opposing the NATO Double-Track Decision (which deployed missiles in Western Europe) even though a majority of her party favoured it. Däubler-Gmelin's nomination ultimately collapsed after the SPD failed to win the CDU's support during negotiations in the Bundestag. Another exceptional example was the 2008 SPD nomination of law professor Horst Dreier. Dreier was criticized from all sides: by conservatives for his liberal views on stem cell research and by liberals for his alleged support for limited exceptions to the absolute prohibition of torture, as in the 'ticking-time-bomb scenario'. Eventually, Dreier withdrew his candidacy.52

4.1.4 The legislative supermajority model: key constitutional considerations (medium level of political investment)

Legislative control of judicial appointments can help bolster the constitutional court's accountability to the people, as represented by political parties. Democratically elected legislatures are intended to represent and reflect the will of their constituents. A court composed of justices chosen by the parties in the legislature improves the public's perception of the court as a political and moral authority and neutral arbiter. Furthermore, if this model is used in a federal state, where the second chamber of the legislature consists of state representatives (or where state-level officials are otherwise involved in selecting judges), it can promote a sense of political investment in the court on a regional level through the involvement of the second chamber.

Requiring a supermajority vote to approve a nominee for the constitutional court offers two benefits. First, it ensures that both the current legislative majority party and opposition parties are politically invested in the court's composition. The supermajority requirement guarantees that the governing party cannot elect a candidate without having consulted the opposition, and *vice versa*, except in rare cases in which one party holds a supermajority in the legislature. Assuming that all major political forces are represented in a country's legislature, the legislative appointment procedure can be said to include the interests of the broader public and major societal groups. Studies of Germany's FCC show that the consensus-oriented appointments procedure generally prevents the Court from systematically discriminating against any given political actor, because most political interests are represented on the Court.⁵³

Second, a supermajority requirement tends to favour moderate judicial candidates whose ideological views are acceptable to most or all major political parties. This can help provide a certain degree of stability in a court's rulings, and adherence to precedent. Of course, this could also be seen as a disadvantage, because it prevents candidates with especially progressive or conservative political views from being appointed to the bench, however impressive their legal qualifications may be.

The exact supermajority required should be carefully considered when implementing this model. If it is too high, it may result in deadlock or long delays in filling vacancies on the Court if the legislature is unable to identify nominees who can gain that level of support. For example, the supermajority required to confirm a nominee for the FCC in the Electoral Committee of the *Bundestag* was initially set at three-quarters. Yet after observing the lengthy delays in the appointments process resulting from the need to find a candidate who would gain such wide approval, Germany's major parties soon agreed to lower the supermajority to two thirds.

One potentially significant disadvantage of the legislative supermajority model is its inevitable ties to the political party system. This model may tend to privilege constitutional court candidates who are members of (or otherwise supported by) the country's dominant political parties, and disadvantage candidates who have no political affiliation or who are affiliated with minority parties. In turn, the court's membership may be less politically inclusive than desired. This is a particular concern for countries undergoing a democratic transition after authoritarianism, as are many MENA region countries. In these cases, one party may dominate the legislature while new political parties are struggling to win seats in the legislature and gain greater stature in the political system. Excluding smaller political parties from the appointments process will reduce the sense of political investment in the constitutional court felt by those parties and by the citizens who support them. Similarly, if the public perceives the appointments process as determined simply by political horse-trading rather than by seeking out the best candidates for the job, the court's legitimacy as a whole may be damaged.

The supermajority requirement also makes the success of this model highly contingent upon the ability of the political parties represented in the legislature to cooperate and compromise. If a vacancy arises on the constitutional court at a time when the major political parties are bitterly divided, obtaining the supermajority required to elect a judge to the court may be difficult, resulting in delays and deadlock in the nominations process. Furthermore, as the German experience shows, this model operates most smoothly when (1) the same political parties dominate politics over a long period of time and (2) those parties are able to reach an informal agreement on how to allocate seats on the court bench among themselves. In a more volatile legislative environment, it may be more difficult to muster sufficient support among different parties to reach the supermajority vote required to appoint a candidate. For this reason, the legislative supermajority model may be less suitable for MENA region countries emerging from authoritarian rule, because new democracies often experience a degree of volatility in political parties: new parties form while others dissolve, and political parties may gain or lose a large number of seats in parliament with every election cycle.

Unitary legislative control over judicial appointments has understandable advantages, particularly for countries emerging from authoritarian rule, where corrupt presidents and elites have previously co-opted the judiciary and appointments process. However, this model does not necessarily insulate judicial appointments from any executive branch influence. In a parliamentary system, this model also provides indirect leverage for the executive, since the executive is generally controlled by the majority party (or coalition) in the legislature.

If the legislative appointments model is used in a country in which the legislature is consistently dominated by the same party, the model may collapse into executive control of the appointments process because judges may be selected based on their loyalty to the ruling party. Just as the executive branch can capture a court, so too can a political party or legislative majority, if the political party structure does not create real competition between parties for control of the legislature.

The model as applied in Germany

The experience of the FCC offers a few additional points for constitutional drafters to consider. One frequent critique of Germany's legislative supermajority model stems from the way in which the *Bundestag* selects judges. These judges are not elected by the full German legislature, but rather by an Electoral Committee formed of *Bundestag* members. Opponents of this system (most often smaller political parties) have repeatedly argued that this approach is undemocratic because the small Electoral Committee does not adequately represent all political parties in the *Bundestag*. In addition, the negotiations over nominees are led by a small group of powerful party representatives in strict confidence, rendering the whole process detached from public scrutiny and insufficiently transparent. However, attempts to amend the appointments procedure have thus far been unsuccessful; a 2012 FCC decision affirming the

constitutionality of the Electoral Committee makes any change in the near future highly unlikely. $^{\rm 54}$

Seen as a whole, the German appointments procedure, combined with non-renewable limited terms for judges, has created a strong and independent Constitutional Court that is widely respected domestically and internationally. However, constitutional drafters considering the legislative supermajority model must take into account the many factors operating in Germany's favour when implementing this model, in particular its robust political party system. An appointments procedure run by the legislature requires the existence of strong, stable and competitive political parties that are capable of negotiating and compromising with each other. Other factors contributing to the FCC's success include the commitment to the rule of law that has evolved over time within the German judiciary, as well as the strong public support the Court enjoys, which is strengthened by an independent media. Finally, the creation of the FCC and its success may also be attributed to factors unique to Germany's democratic transition, such as the influence of the post-war occupying forces and the legacy of the Holocaust and the Nuremberg trials.⁵⁵

4.2 The judicial council model: South Africa

The judicial council model is a mixed appointments model. Judicial councils are intended to insulate judicial appointments from partisan politics, and to foster judicial independence by acting as an intermediary between the executive and legislative branches and the judiciary. In so doing, they generate political investment in the court by opposition political parties by reducing the risk that the governing party will capture the court through appointments. Judicial councils erect a buffer around appointments to the constitutional court by diversifying the appointments process through involving multiple political branches and, often, non-political groups such as bar associations, legal scholars and other civil society actors. There is no set formula for the composition of a judicial council; countries will select a council's membership according to their own political context.⁵⁶

The judicial council's precise role in the constitutional court appointments process varies by country but, generally speaking, the council is responsible for some or all of the following:

- soliciting applications or nominations for vacancies on the constitutional court;
- reviewing the applicants or nominees and creating a shorter list of candidates to investigate more thoroughly;
- conducting in-person interviews of the candidates on this shorter list and/or other investigations (for example, requesting financial or other documentation from the

candidate and speaking to each candidate's colleagues and peers about his or her qualifications and suitability); and

• selecting either one candidate to be appointed to the court by the executive or legislature or a shortlist of candidates from which the executive or legislature will select a person to appoint to the court.

Some judicial councils also oversee the discipline, promotion and/or removal of judges. One of the leading examples of the judicial council model is South Africa's Judicial Service Commission (JSC), which manages all judicial appointments. This Commission was created during South Africa's historic transition from apartheid to multiparty democracy in the early 1990s. In stark contrast to the apartheid-era judiciary, in which judges were unilaterally appointed by the executive branch and were generally barred from engaging in judicial review of Parliament's actions, the new South African Constitution sought to establish an independent and impartial judiciary. The Constitution also established a new Constitutional Court with full powers of judicial review. The JSC, in turn, was created to manage many judicial functions, including the selection of judges for the Constitutional Court.

4.2.1 South Africa: historical and political context

Apartheid-era South Africa was characterized by legalized political and social discrimination against the majority black population (as well as other non-white ethnic groups) perpetrated by the Afrikaner-dominated National Party. Under apartheid, the President officially made all judicial appointments; in practice, the Minister of Justice played the key role in selecting appointees, sometimes on the advice of the chief justice or judge president of the relevant judicial division. There was no transparency in the selection process for judicial appointees, but there is general agreement that many judicial candidates were chosen for their pro-apartheid political views and personal connections. Until the beginning of South Africa's democratic transition in 1990, the judiciary was entirely white and almost entirely male. In the early years of apartheid, courts lacked the express power to review the constitutionality of legislation and did so exceedingly rarely; in 1983 a new Constitution explicitly stripped the courts of judicial review powers over parliamentary acts. Judges had no power to strike down laws that violated human rights, and very few judges even spoke out against the racist apartheid system. Most judges limited themselves to determining Parliament's intent in promulgating the law and enforcing it.57

As a result, when South Africa began the process of democratic transition, judicial reform was a key focus area during the negotiations between the National Party, the African National Congress (ANC) and other opposition parties. The ANC, which was virtually certain to take power after the first democratic elections were held, sought to transform South Africa's judiciary, making it more diverse and transparent, granting it

the power of judicial review, and creating a new, powerful Constitutional Court to be the final arbiter on the constitutionality of government actions and to bolster public confidence in the judiciary.⁵⁸

The importance of South Africa's Constitutional Court cannot be overstated. Viewed as the 'institutional embodiment of South Africa's new democracy', its structure and functions were widely debated during the negotiations that shaped South Africa's democratic transition. These negotiations took place in two stages: an initial phase in which an interim Constitution was agreed upon by the ANC and the National Party to govern the country until democratic elections were held, and a second phase during which the newly elected Parliament produced a final Constitution for a post-apartheid South Africa.⁵⁹

During negotiations for the interim Constitution, it was generally agreed that 'an overwhelmingly white and male judiciary [could] not adequately and fairly deliver justice to a majority black and female population'. But there was disagreement over the appointment process. The National Party, anticipating a future role in government, wanted to retain as much power for the executive over judicial appointments as possible, as did many ANC representatives. However, some members of the ANC and a host of other political parties and interest groups that participated in the negotiation process sought to create a more participatory appointments process that included a broader variety of actors that would insulate judicial appointments from the governing party (the ANC).⁶⁰

The Democratic Party, which had served as the main opposition party during apartheid, also advocated a participatory process that would include non-governmental actors, such as members of the legal profession. Many academics and civil society groups also supported this position. Another model proposed during negotiations was that appointments be made by the President in consultation with a joint standing committee of Parliament. Under this model, each nominee would need to gain the approval of a three-fourths majority of a joint sitting of both houses of Parliament before being appointed by the President. This proposal proved short-lived, however: the Democratic Party opposed it because they wanted non-political actors to play a role in appointments as well, to prevent the process from becoming a mere exercise in political horse-trading between parties. The National Party opposed the proposal because they favoured executive-controlled appointments, on the assumption that South Africa's transitional government would include members of minority parties who would have veto powers in certain circumstances. As a result, they strongly favoured judicial appointments by the President, acting on the advice of the cabinet, because they believed that they would have veto power as a minority cabinet member. The ANC, which everyone expected to control a majority in the new government, also supported this model.⁶¹

However, when it became clear that no minority party would have the option of a veto over Constitutional Court nominees, the Democratic Party opposed the executive-based appointments system and insisted that the parties renegotiate the appointments process. The Democratic Party warned that giving the executive control over Constitutional Court appointments would allow it to manipulate the process and jeopardize the Court's independence from its inception. A wide variety of media outlets, along with prominent representatives of the legal and judicial professions (including the deans of various law schools around South Africa), joined the Democratic Party to oppose an executive-based appointments process as a way to erect a buffer between the governing party (the ANC) and the Court.⁶²

Building on these proposals, the Democratic Party argued that the JSC, which was established during the negotiations as the body that would carry out appointments for all other courts in South Africa, should also appoint the Constitutional Court's judges. Negotiations over the membership of the JSC had been hotly contested, and the ultimate compromise created a large commission that included members of the legislature, the judiciary, the executive, the legal profession, civil society and academia. This satisfied proponents of an independent commission by ensuring that the Commission would include a large contingent of legal professionals while also preserving a role for the executive and the legislature, thereby promoting the idea of judicial accountability to the public. The Commission's appeal lay in its diverse composition. Ultimately, this proposal won the day in the negotiation process. The interim 1993 Constitution set out the rules for appointing the first Constitutional Court judges (Articles 97–99). With a few minor changes, these rules were implemented in South Africa's 1996 Constitution, which uses a judicial council model for Constitutional Court appointments (Article 178).⁶³

4.2.2 South Africa: appointments procedure

The South African Constitutional Court is comprised of 11 justices: the Chief Justice, the Deputy Chief Justice and nine other justices (Article 167). Justices are usually appointed for a non-renewable term of 12 years, or until he or she attains the age of 70, whichever occurs first (Article 176). However, in certain cases a justice may remain on the bench for 15 years, or until he or she reaches the age of 75, if necessary to fulfil the legal requirement of 15 years of active service. Appointments to the Constitutional Court are managed by the JSC, which also facilitates disciplinary matters and the administration of justice generally (Article 178).⁶⁴

Composition of the Judicial Service Commission

The JSC is comprised of 23 members. Compared to judicial councils in other countries, the JSC has a relatively large number of members, because it is intended to represent 'a wide section of the South African legal and political establishment' and to include as many different interest groups as possible. Eleven of the JSC's members are appointed by the President: the presiding Chief Justice of the Constitutional Court, the President of the Supreme Court of Appeal, the Minister of Justice, two practicing advocates and two practicing attorneys (who are appointed by the President after being nominated by their respective professions) and four laypersons selected after consultation with the leaders of all parties represented in the National Assembly (Article 178(1)). Six of the remaining 12 members of the JSC are chosen from among members of the National Assembly, the lower house of Parliament. At least half of those six must be members of opposition parties (Article 178(1)(h)). Another four members are chosen from among the permanent delegates to the National Council of Provinces, the upper house of Parliament (Article 178(1)(i)). The remaining two members must be a law professor designated by his or her peers at South African universities, and one judge president, also designated by his or her judicial peers (Articles 178(1)(g) and (c), respectively).⁶⁵

The appointments process

The first stage of the appointments process is identical for all judicial appointments, including the Constitutional Court. When a judicial vacancy arises, the head of the relevant court notifies the JSC, which publishes a call for nominations. Candidates are required to submit a completed application, which includes the candidate's resume, a statement confirming that he or she is a member in good standing in his/her professional organization, and a questionnaire that 'solicits information about the applicant's personal and professional life, including the applicant's contribution in the struggle against apartheid, commitment to the principles underlying the Constitution, financial interests, practice, and other relevant experience'. This application package is then circulated among JSC members, and a subcommittee is appointed to examine the applications and select a shortlist of candidates to interview.⁶⁶

In determining which candidates to interview, the Constitution requires that consideration be given to South Africa's gender and ethnic composition (Article 174(2)) in the interest of creating a diverse and representative judiciary. The JSC has indicated that they consider this to mean more than simply increasing the numbers of black justices and women on the bench: 'the values and visions of the appointed individuals must also comport with the explicit social justice commitments embodied in the Constitution'. Once the subcommittee has selected the candidates to be interviewed, the list is distributed to the plenary JSC. Once approved, the list of interviewees is published and interviews are conducted, which are open to the public.⁶⁷

At this stage in the selection process, the JSC's role differs depending on whether the vacancy pertains to the Constitutional Court or another court within the South African judiciary. In all cases, the President makes the formal appointment to the Court. However, for all appointments except those to the Constitutional Court, the President is bound by the advice of the JSC (Article 174(6)). In practice, this means that the JSC recommends only one candidate after the interview process, who is then appointed by the President.⁶⁸

The President appoints the Chief Justice and Deputy Chief Justice of the Constitutional Court after consulting with the JSC and the leaders of the political parties represented in the National Assembly. For the remaining nine members of the Court, the President's appointment power is subject to important limitations due to the Court's crucial political role. The JSC is required to prepare a list of candidates for the President must make the appointments from this prepared list of nominees, but may reject the first list of candidates or particular candidates on the list, giving reasons for the rejection. In this case, the JSC must supplement the list with additional candidates. The President must choose candidates from this supplementary list; an additional list of candidates cannot be requested. The President must also consult with the National Assembly before making a final decision on whom to appoint. At any given time, at least four of the justices on the Court.⁶⁹

4.2.3 South Africa: appointments process in practice

In a short span of time, South Africa's Constitutional Court has become one of the most respected in the world. Thanks both to the prominent jurists who were the Court's first appointees after its creation, and to the ruling ANC's early commitment to respecting the rule of law, the Court has been able to operate as an effective check on executive and legislative power. It has also played a key role in consolidating South Africa's democratic transition.⁷⁰

The Constitutional Court has not shied away from confrontations with the ANC-led government. Among many decisions striking down or demanding modification of ANC-supported laws and policies, the *Treatment Action Campaign* (TAC) and *Glenister* decisions stand out for the international attention they received and their impact on the government.

In 2001, South African civil society organizations, led by the TAC, lodged a legal challenge against the government's HIV/AIDS prevention and treatment strategy. They argued that the government had an obligation to implement an effective strategy to

prevent the transmission of HIV from a mother to her child, and that the government programme in place at the time violated South Africans' constitutional rights by restricting access to nevirapine, a drug shown to prevent mother-to-child transmission. The government had ample supplies of nevirapine, but argued that further research on the safety of the drug and a comprehensive implementation programme were needed before it was made widely available.⁷¹

In a landmark judgement, the Court held that the government's restrictions on the distribution of nevirapine did not fulfil its constitutional obligation to provide reasonable measures within available resources for the progressive realization of the right to health. It ordered the government to revise its HIV/AIDS strategy, including the removal of restrictions on access to nevirapine. The judgement was an embarrassment to then-President Thabo Mbeki and his Ministry of Health, which had already been widely criticized by public health specialists for its insufficient response to the AIDS epidemic in South Africa.⁷²

The *Glenister* case concerned South Africa's anti-corruption unit. In 1999, the government created the Directorate of Special Operations (known as the Scorpions), a special investigative unit focused on corruption and organized crime. The Scorpions' work drew both praise and criticism. After numerous investigations into high-ranking ANC members, including then-Deputy President Jacob Zuma, the government accused the Scorpions of acting outside their jurisdiction and closed the unit. Opponents criticized the ANC for shuttering what many people saw as a highly effective investigative team, and alleged that the Scorpions unit was closed because of the embarrassment it had caused the ANC. It was replaced by a new Directorate for Priority Crime Investigation, nicknamed the Hawks.⁷³

Hugh Glenister, a businessman, challenged the constitutionality of the law that closed the Scorpions and replaced the unit with the Hawks. In 2011, the Constitutional Court ruled that this law was unconstitutional because the Constitution obligates the government to establish an effective anti-corruption mechanism, and that the structure of the Hawks did not fulfill this obligation because the unit was insufficiently independent: the law's requirement that a Ministerial Committee coordinate the Hawks' activities, and the insufficient job security protections for members of the Hawks, rendered the unit vulnerable to government pressure. The Court's ruling required the government to remedy these flaws in the Hawks' structure.⁷⁴

A number of ANC officials have publicly criticized the Court and called for reforming various aspects of the judiciary, presumably in response to Court decisions ruling against the ANC's position on various issues. These statements are generally swiftly taken up for debate in the media and by various civil society actors, and often result in criticism of the ANC. Only a few years into the JSC's operations, then-ANC Secretary-General

Kgalema Motlanthe raised the possibility in 1998 of reviewing the JSC's role, remarks which were likely prompted by ANC displeasure with the JSC's appointments. In 2005, the National Executive Committee of the ANC said that 'many within our judiciary do not see themselves as being part of these masses, accountable to them, and inspired by their hopes, dreams and value systems', warning that this could result in 'popular antagonism' toward the judiciary. Again in 2008, ANC Secretary-General Gwede Mantashe made remarks about the Constitutional Court that suggested he viewed it as 'counter-revolutionary'; the ANC distanced itself from the comments and Mantashe later claimed he was misquoted. Finally, in 2012, President Zuma called for a review of the Constitutional Court's decisions, possibly including its powers. The results of this review and its practical impact on the Court have yet to be seen.⁷⁵

The ANC has not yet taken serious steps toward changing the Constitutional Court or the JSC's central role in the appointments process. However, the Court enjoys only limited public support in South Africa, and like any other court, it has no means of resisting changes to its structure, powers or appointments process. The ANC's persistent control of the executive and legislative branches highlights how vulnerable the Court could be if the ANC decides to alter the appointments procedure or exert more influence over selecting members of the JSC. The Court's appointments procedure has already drawn criticism, chiefly due to the ANC's dominance of the JSC and a perceived lack of transparency in the selection process.⁷⁶

One-party dominance

The ANC's continuing popularity has an impact on Constitutional Court appointments that may have serious long-term implications for the Court's independence. The JSC was intended to encompass a wide range of political interests and to reflect the diversity of South African society. For this reason, the JSC's members include both political and non-governmental actors, and the government's power to select Commission members is divided between the executive branch and the two chambers of Parliament. In practice, because of its electoral dominance, the ANC selects the majority of the JSC's members, raising questions as to whether the JSC can be considered truly independent.

As described above, the President appoints 11 of the JSC's 23 members. Four of these appointees must be nominated by the advocates' and attorneys' professions, however, and should not be viewed as purely presidential appointments. In addition, the Chief Justice of the Constitutional Court and the President of the Supreme Court of Appeal are nominated by the President after consultation with the JSC and, in the case of the Chief Justice, with the party leaders in the National Assembly (Article 174), thus somewhat tempering the President's ability to make these appointments unilaterally. However, both the National Assembly and the National Council of Provinces (NCOP), which together select another ten of the Commission's members, are dominated by the

ANC, which holds 66 and 65 per cent of the seats in the National Assembly and the NCOP, respectively. Thus even with the Constitution's requirement that three of the six National Assembly appointees to the JSC must be members of the opposition, the ANC effectively has the power to appoint well over half of the Commission's 23 members.⁷⁷

In the early years of the JSC's operations, the large size of the Commission and the wide range of interest groups represented appeared to prevent any one group from dominating its decisions. The fact that the vast majority of Commission members were lawyers, judges or law professors also appeared to temper political influence over the Commission. These factors help to insulate the JSC against ANC dominance, although the ANC has recently taken several steps that may suggest an increased effort to exert influence over the Court.⁷⁸

In 2009, it replaced a member of the opposition party Democratic Alliance (the only member of the opposition from the NCOP on the JSC) with an ANC member. The replacement was completely legal; the Constitution does not require that any set number of opposition members from the NCOP be appointed to the JSC. However, in earlier years the parties in the NCOP had followed an informal practice of designating Commission members in proportion to the relative strength of party representation in the NCOP, thereby reserving at least one seat on the Commission for an opposition party member. As of 2011, that practice appears to have been abandoned, and all Commission members from the NCOP are now ANC designates.⁷⁹

Concerns about the ANC's influence over the Court have been heightened by controversies over its Chief Justice, who according to the Constitution is selected by the President. In 2011, President Jacob Zuma attempted to extend then-Chief Justice Sandile Ngcobo's term beyond the constitutionally mandated maximum length, relying on the Judges' Remuneration and Conditions of Employment Act as his legal basis for doing so. This decision was widely criticized and resulted in a challenge brought before the Constitutional Court, which unanimously held that both the relevant section of the Act and the President's extension of Chief Justice Ngcobo's term were unconstitutional.⁸⁰

Chief Justice Ngcobo ultimately stepped down, but President Zuma again drew criticism when it came time to select his replacement. Deputy Chief Justice Dikgang Moseneke was viewed as a frontrunner for the position, having served as acting Chief Justice after Ngcobo's departure. However, Moseneke was reportedly disfavoured by President Zuma for personal reasons, and the President appeared to snub Moseneke by appointing Mogoeng Mogoeng (who many perceived to be less qualified) instead. However, other Constitutional Court appointments made by Presidents Zuma and Kgalema Motlanthe (a caretaker president who took office after the resignation of

President Thabo Mbeki) have been widely applauded, and provide reason to contest the accusation that the ANC is trying to pack the Court. Former Chief Justice Sandile Ngcobo, initially appointed to the Court by Nelson Mandela and promoted to Chief Justice by Zuma, has been praised for his independence and support for democratic participation. Justice Edwin Cameron, appointed by Motlanthe, is an openly gay and HIV-positive judge who publicly criticized former President Mbeki's HIV/AIDS policies.⁸¹

Transparency

Apartheid South Africa's completely opaque process of selecting judges made transparency in judicial selection processes a key goal for the new South African democracy. The JSC is intended to help fulfil this goal. It publishes all judicial vacancies with an open call for nominations, publishes the shortlist of candidates selected for an interview, solicits comments on those candidates prior to the interviews from several different legal associations (including the South African Law Society and the Bar Council), opens the candidate interviews to the public, and often refers to third-party information provided to the Commission about a particular candidate during interviews, rather than keeping such opinions confidential. The print media is also permitted to attend the JSC's interviews and report on them, further contributing to the transparency of the process.⁸²

Overall, the JSC's procedures for selecting judicial appointees are markedly more transparent than the apartheid-era appointments process. However, the first stages of the JSC's process, when it reviews applications for a posted judicial vacancy and decides which candidates to shortlist for interviews, are less transparent. It does not publish a list of all individuals who apply for a judicial vacancy, which makes it difficult to discern how they select the shortlisted group to interview. Furthermore, the JSC's post-interview deliberations on which candidates to recommend for appointment are also not public. The JSC defends this practice by arguing that it protects the privacy of individuals who are not shortlisted, that making deliberations public would make Commission members more reluctant to express their unvarnished opinions of candidates' qualifications, and that little would be gained by disclosing its private deliberations. The JSC has also made public the criteria it looks for in judicial candidates, which sheds some light on the focus of their private deliberations; these criteria include characteristics such as intellectual ability, fairness, independence, perceptiveness, courage and integrity.⁸³

The developments summarized here have fuelled fears that the ANC is attempting to exert undue influence over the Constitutional Court, though it is too early to assess the validity of those fears. Some scholars note that fewer applicants are coming forward when the JSC announces a vacant judicial post, and have suggested that this may be due to the perception that the JSC is controlled by the ANC, and that therefore the successful candidate has been chosen before the appointment process even begins. It is not yet clear whether these issues will affect the Court's independence, and at present the Constitutional Court continues to be one of the most admired in the world.⁸⁴

4.2.4 The judicial council model: key constitutional considerations (high level of political investment)

The judicial council model's greatest strength is the opportunity it presents to involve a wide spectrum of society in the judicial appointments process. Diversity of membership on the council both reduces the risk that any one political group will be able to dominate the appointments process and promotes a broad sense of political investment in the court, since many different actors play a role in forming the court. For example, the South African JSC includes members of the legislature (including members of opposition parties) and the judiciary, legal professionals and law professors. Its inclusion of non-governmental members, and the constitutional requirement that a set number of seats (three) be reserved for opposition political parties, are both good ways to promote the representation of many different interests on the judicial council. Particularly if a country's history suggests that one political party may remain dominant for long periods of time, creating constitutional requirements that aim to guarantee opposition party representation on the judicial council at all times is advisable.

Although the ISC was designed to include a wide range of political constituencies, South Africa's experience also shows that a judicial council is not immune from capture by a political party. Although South Africa's political context is unusual-few democracies in the world feature one political party that is as consistently dominant in both the executive and legislative branches as the ANC-it nonetheless provides a lesson for policymakers considering adopting a judicial council. South Africa's President is responsible for appointing almost half of the ISC. When these seats are added to the seats appointed by Parliament, which is dominated by the ANC, the President's party appoints a clear majority of the members of the JSC. Given the concern in South Africa that this may lead to ANC control of the JSC, policymakers composing a Judicial Council should take care to ensure that no single actor or political group can control a large proportion of seats on the Judicial Council. Ultimately, the Constitutional Court will only be as independent as the judicial council that selects its judges. This risk has not materialized yet in South Africa, but there is the danger that it could. Therefore careful attention must be given to the composition of a judicial council to reduce the risk of capture by a dominant political party. One option for dealing with this problem is to ensure that institutions controlled by the governing political party do not appoint a majority of seats on a judicial council.

The importance of transparency in a judicial council's proceedings

The South African experience also highlights the importance of transparency in the judicial council's proceedings. On the whole, the JSC is highly transparent. However, the lack of transparency in the candidate shortlisting process has raised concerns, and opened the JSC to accusations of bias towards ANC-friendly candidates. Over time, this may result in diminished political and popular support for the JSC as an effective barrier against executive capture of the judiciary.

Insulation of judiciary from partisan politics promotes judicial independence

A judicial council also creates a barrier between the process of selecting judges and the often combative atmosphere of partisan politics. This can be particularly helpful for encouraging public trust in the judiciary in countries emerging from authoritarian rule, where the judiciary may previously have been under the control of the executive. While the executive and legislative branches often select members of the council, they are not directly involved in the appointments process (with certain exceptions, as in South Africa, where the executive appoints the Constitutional Court's Chief Justice). This can help counter accusations that judges' decisions are politically motivated, because the judges are not selected directly by political parties. Over time, this can promote public trust and confidence in the judiciary as an independent arbiter of political disputes.

4.3 The judiciary-executive model: Egypt and Iraq

The judiciary-executive model is a mixed appointment model that divides the power to appoint judges to a constitutional court between the judicial and executive branches. In most iterations of the model, the judiciary (most often representatives of the highest courts) nominates either one or a shortlist of candidates to the constitutional court. The executive must then select a candidate (or approve the selection made by the judiciary) and formally appoint the judge to the court. Other variations of the judiciary-executive model provide that the executive nominates either one candidate or a list of candidates to the court, and the judiciary must appoint the single candidate or a name from the list. By relying on the joint consent of the judiciary and executive, the model intentionally excludes the legislature.

The stated rationale behind the judiciary-executive model is to create an independent court that is insulated from the short-term political concerns of the legislature, while maintaining accountability to the public through the executive. Excluding the legislature is designed to protect the appointments process from being held hostage by political parties that may try to place their supporters on the court or that are unable to agree with other parties regarding the best candidates for the court. However, because it is also desirable for the members of a court to have a degree of accountability to the public through the elected branches of government, the judiciary-executive model includes a key role for the executive (president or prime minister), who is typically elected. Judicial involvement in appointments is intended to ensure that they are based on a candidate's qualifications and legal training rather than political affiliation, thereby safeguarding judicial independence. Including both the judiciary and the executive in appointments helps prevent the executive from exercising too much control over appointments and simply selecting judges who will rule in its favour, while also preventing the appointments process from becoming too insular and detached from the broader political context, as might be the case if the judiciary were solely responsible for selecting the constitutional court's membership.

However, the model carries significant risks. Completely removing the legislature from the constitutional court appointments process is likely to generate distrust of the court on the part of political parties. Because parties will have little ability to influence appointments to the court, they will not invest in it politically, and they may attack the court as illegitimate or biased if its rulings displease various political parties. Furthermore, because the only elected official involved in appointments is the president or prime minister—who may not be directly elected by citizens—the court is unlikely to be seen as democratically accountable to the public, further diminishing its public support. The constitutional court will thus be highly susceptible to attack from politicians who are unhappy with its rulings.

The experiences of Egypt and Iraq, which both use the judiciary-executive model for appointments, underscore the problems that can arise. The Egyptian case illustrates how the model can leave the judiciary vulnerable to domination by the executive branch, and shows how the judiciary has struggled to balance independence with political accountability during the post-Mubarak transition. The Iraqi experience demonstrates how a flawed constitution-drafting process, coupled with a difficult and volatile political context and the use of the judiciary-executive model, can result in a court whose legitimacy is seriously damaged by perceptions that it is dominated by the executive branch.

4.3.1 Egypt

4.3.1.1 Egypt: historical and political context

The 1971 Egyptian Constitution created the Supreme Constitutional Court (SCC), which began operating after the passage of implementing legislation in 1979, Law No. 48 of 1979 (SCC Law). Then-President Anwar Sadat and the drafters of the SCC Law faced both external and internal pressure to create an independent constitutional court to replace the Nasser-controlled Supreme Court, which was created in 1969 by presidential decree and composed solely of Nasser-appointed judges known for their

loyalty to the regime. During the decade that the Supreme Court functioned, from 1969 to 1979, the Court issued over 300 rulings, 'not one of which significantly constrained the regime'. In response, the legal profession and civil society within Egypt demanded reforms guaranteeing greater judicial independence. It has been suggested that an independent constitutional court was also desirable to the government because it would help bolster the state's credibility with respect to the protection of private property interests, attracting badly needed foreign investment.⁸⁵

The SCC was created by an autocratic government, which left little room for debate among political actors over the appointments process. Instead of various political forces coming together to establish an appointments process that all found reasonably satisfying, the decision on how to appoint SCC judges was left to then-President Sadat and the drafters of the SCC Law, who were party loyalists. The result was an appointments process that was legally dominated by the executive, but informally left largely under the control of the Egyptian judiciary.

While Sadat advanced judicial independence in Egypt with the creation of the SCC, he still sought to retain ultimate authority over the Court by controlling the appointments process. Under the 1971 Constitution and the SCC Law, the President had the formal authority to appoint the Chief Justice, assuming that his chosen candidate satisfied the minimum qualifications (SCC Law article 5). Despite this formal authority over SCC appointments, for the first two decades of its existence, the SCC followed an informal procedure whereby the President promoted the most senior serving justice on the SCC to be Chief Justice. For the other justices on the SCC, the Chief Justice and the General Assembly of the Court (which consisted of all members of the Court (SCC Law article 7)) each nominated one candidate, and the President appointed one of these two candidates (SCC Law Article 5). In practice, the Chief Justice and the General Assembly of the Court always nominated the same person, whom the President would then appoint. In effect, the SCC became a self-perpetuating body, relatively free from executive control. Throughout the 1980s and 1990s, the SCC distinguished itself in the MENA region as a relatively independent court, issuing decisions that had a real impact on limiting autocratic rule and pressuring successive authoritarian regimes for greater rights and freedoms. As will be discussed below, however, the SCC did not always rule against the regime, and many of its rulings upheld key elements of the autocratic state and politically repressive practices.⁸⁶

At the end of the 1990s, the informal judiciary-controlled appointments process broke down under President Mubarak's rule. The SCC continued to issue rulings that curtailed executive power to an extent, particularly in the areas of press freedoms, operations of non-governmental organizations (NGOs), electoral law and tax law. These rulings against Mubarak's regime ultimately proved to be more than he was willing to tolerate, and in the early 2000s he reasserted his formal authority over the appointments process and appointed a Chief Justice who was loyal to his government. Because the Chief Justice controls the SCC's docket and oversees the process of writing the Court's decisions, he is a very influential member of the Court. Furthermore, because the SCC Law did not specify the number of judges on the Court, the Chief Justice immediately increased the SCC's size by 50 per cent, appointing new judges who were aligned with the executive.

Because executive influence over the SCC had never been legally curtailed, the continuation of its informal judiciary-controlled appointments was completely dependent on the goodwill of the executive. Mubarak was able to bring the SCC to heel as soon as he wished. He was able to capture the Court through the formal appointments procedure (which remained the same as in the 1970s) and its subsequent decisions reflected the influence of the executive.⁸⁷

Following the fall of Mubarak in 2011, the SCC has continued to function, first under the rule of the Supreme Council of the Armed Forces (SCAF), then under President Mohamed Morsi's rule and now (at the time of publication) under an interim authority appointed by the SCAF. Egypt has operated under a series of constitutional frameworks during this period that have preserved the authority of the SCC to adjudicate on constitutional disputes: the Constitutional Declaration of 30 March 2011 (Article 49), the 2012 Constitution of Egypt (Article 175) and the Constitutional Declaration of 8 July 2013 (Article 18). An important issue throughout this process has been the procedure for appointing justices to the SCC. In response to Mubarak's previous domination of the Court and the changing political landscape, the SCC pressured the SCAF to pass a decree that amended the SCC's appointments procedure—instituting a procedure similar to the informal one that was followed until Mubarak began exercising more authority over the selection of judges—in June 2011. The SCC is currently functioning under the appointments model set out under this decree, although the constitutional situation is far from stable.⁸⁸

4.3.1.2 Egypt: appointments procedure

While the June 2011 SCAF decree still governs the SCC's appointments procedure, Egypt's now-suspended 2012 Constitution changed other aspects of the SCC. Under the 2012 Constitution, the SCC was composed of 11 judges, including one Judge President. Other than stating that the President appointed SCC judges by decree, the 2012 Constitution did not specify the procedure for appointments to the SCC, deferring that question and many others to future implementing legislation (Article 176). While the 2012 Constitution was in force, no new legislation regarding the SCC's appointments procedure was passed; therefore the appointments procedure outlined in the June 2011 SCAF decree is still in force. For similar reasons, the term length of SCC judges is still governed by pre-transition legislation, which allows SCC judges to remain

on the Court until they reach the mandatory retirement age, which is currently set at 70.⁸⁹ The Constitutional Proclamation of 8 July 2013 likewise leaves these matters to legislation (Article 18) that has yet to be enacted.

According to the June 2011 SCAF decree, the General Assembly of the Court selects the SCC's Chief Justice from among the Court's three most senior members, and the President formally appoints the Chief Justice. For all other appointments to the SCC, the decree requires the President to give precedence to the Court's Commissioner's Body, which is a group of judges that helps the SCC judges prepare cases and opinions. It is responsible for managing the cases that come before the SCC and issuing advisory reports detailing the issues raised in the cases.⁹⁰

4.3.1.3 Egypt: appointments process in practice

As mentioned above, during the 1980s and 1990s the SCC was able to exercise a degree of independence, ruling against the Mubarak regime in numerous decisions. The Court was one of the only institutions in authoritarian Egypt that presented any real challenge to the regime. The SCC issued important rulings regarding electoral laws and press liberties. In 1993, the SCC struck down a provision in the criminal code that required defendants in libel cases to present proof for published statements within five days. The SCC found that the five-day limit was too restrictive and interfered with the ability of the press to monitor the government. The SCC also found a provision of the Press Law that required newspapers to receive prior approval from the Council of Ministers before publishing to be unconstitutional. In addition, the SCC struck down Article 15 of Law 40/1977, which imposed vicarious criminal liability on the heads of political parties for material published in party newspapers in cases claiming libel of public officials, as well as a similar law imposing vicarious criminal liability on newspaper editors for libellous statements published in their papers. The SCC also issued several decisions that strengthened protections for NGOs in Egypt, including a 2000 ruling that struck down Law 153/1999, which severely restricted the activities and rights of NGOs. The SCC's relative independence enabled constitutional litigation to become a promising avenue for NGOs and human rights groups to challenge illiberal legislation.⁹¹

However, the SCC's ability to check executive power was far from unlimited. While it produced liberal judgements in the area of human rights, it was aware of its limited institutional capital and careful not to rule on matters that struck at the heart of the regime. The SCC upheld the constitutionality of Egypt's emergency state security courts, which were responsible for handling all cases prosecuted under the emergency law. The President had ultimate authority over state security courts, and could 'overturn the court's ruling or demand a retrial'. The SCC often delayed ruling on politically sensitive topics (including electoral laws and the use of military courts to try civilians), sometimes for years, in an effort to avoid a direct confrontation with Mubarak's regime. Through a cautious approach that expanded rights and freedoms in peripheral areas while preserving the core mechanisms of state repression, the SCC was able to maintain its institutional security under an authoritarian regime.⁹²

In July 2000, the SCC issued a decision on Egypt's Political Participation Law (Law 73/1956). The case had been pending before the Court for ten years, but the ruling was issued in the midst of heated debate between the regime and opposition parties over regime-sponsored amendments to the Political Participation Law, the impact of which would have facilitated the rigging of elections by the regime. The SCC's ruling found part of the law unconstitutional because it did not provide for full and direct monitoring of elections by the judiciary, as the Constitution required. The SCC further found that since the parliamentary elections of 1990 and 1995 were carried out under the law, the assemblies formed as a result of the elections were unconstitutional. The decision 'struck at the heart of [Mubarak's] regime-managed liberalization'. In the election following the ruling in 2000, many leading members of Mubarak's National Democratic Party were defeated.⁹³

This SCC decision proved to be a greater exercise of judicial power than Mubarak was willing to tolerate. In 2001, the retirement of SCC Chief Justice Galal provided Mubarak with an opening to exert greater influence over the Court. Mubarak broke the long-standing informal rules for appointments, choosing a loyal member of his regime and a high-ranking official in the Ministry of Justice, M. Fathi Naguib, as the new Chief Justice. Naguib had been the author of 'the vast majority of the government's illiberal legislation during the previous decade'. Mubarak stunned the SCC, legal scholars, opposition parties and human rights groups by ignoring decades of SCC appointment norms. While the President always had the legal authority to appoint whomever he wanted to the role of Chief Justice, he had never asserted that authority until that point. The SCC judges, as well as legal scholars, expected Mubarak to continue the tradition of promoting the most senior sitting SCC judge to Chief Justice.⁹⁴

However, Mubarak had lost patience with the SCC's rulings against the regime, and Naguib's appointment signalled a shift in the balance of power between the executive and the SCC. Naguib's loyalty to the regime was well established. Naguib justified his appointment by arguing that since the Constitution required the Chief Justice to step in as President in the event that the President is incapacitated and the Speaker of the People's Assembly is unavailable, Mubarak needed a Chief Justice he could fully trust. Naguib also argued that his new role did not threaten the independence of the SCC because a majority of justices was required to determine a ruling. In practice, however, former SCC judges have affirmed that there are strong informal norms controlling the Court's operation that discourage junior justices from going against the Chief Justice's will, and that 'even if a majority dares to vote against the will of the Chief Justice, he can simply refuse to sign the ruling'. Naguib also offered a more frank explanation for his appointment, stating in a 2002 interview that the SCC's past rulings had favoured the opposition parties, and 'were not in the interests of the country. This needed to be corrected'.⁹⁵

After his appointment, Naguib moved swiftly to change the ideological composition of the SCC by appointing more justices who were loyal to the regime. Because neither the Constitution nor the SCC Law specified the total number of justices on the Court, Naguib immediately appointed five additional justices to add to the SCC's existing nine. Moreover, while Naguib's method for appointing these new justices to the SCC was technically legal, it violated another informal SCC procedural norm: he selected the new justices from ordinary courts (mainly the Court of Cassation and the Cairo Court of Appeals) rather than recruiting from the State Council, from which junior SCC justices had most often been recruited. The State Council was Egypt's administrative judicial organ, and was viewed by some as more willing to challenge the executive than the ordinary courts. It is unclear whether the sitting SCC judges protested these new appointments, because deliberations over new appointments were not made public.⁹⁶

With Chief Justice Naguib in office and the addition of his handpicked judges to the Court, the SCC changed almost overnight from a powerful, relatively independent institution to one controlled almost entirely by the executive. Executive control over the SCC continued after Naguib's death with Mubarak's appointment of Mamduh Mara'i, who had 'spent much of his career in the inspection department of the Ministry of Justice', as Chief Justice in 2003.⁹⁷

The impact that Mubarak's assertion of control over the SCC had on its independence can be clearly seen in its subsequent rulings. After the appointment of Chief Justice Naguib and the new judges, the Court modified its earlier ruling on the question of authority over election monitoring. After the 2000 elections, the SCC was asked to interpret the meaning of 'judicial authorities' in the electoral law that called for the judicial monitoring of elections, at Mubarak's request. The SCC ruled that 'judicial authorities' included the State Cases Authority and the Administrative Prosecution Authority, which were both under the control of the Ministry of Justice. This decision was a defeat for opposition parties, as it enabled the government to once again tamper with election results through biased election monitoring bodies.⁹⁸

The SCC issued another ruling on elections that similarly favoured the regime. In 2005, Egypt held its first multi-candidate presidential election. Prior to these elections, the ruling party-dominated People's Assembly announced a constitutional amendment that introduced sweeping changes to the election laws that created extremely favourable conditions for the ruling party in the upcoming elections. The amendment required the SCC to review the constitutionality of the new electoral laws within 15 days. Once the

SCC had approved the laws (or requested changes to the laws from the People's Assembly), the laws could not be subject to future challenges. Under the leadership of Chief Justice Mara'i, the SCC approved the constitutional amendment, despite the fact that many opposition political leaders and judges viewed it as 'deeply flawed, contradicting both the spirit and the letter of multiple articles of the constitution'. Notably, the 2005 constitutional amendment regarding Egypt's electoral laws also made the SCC's Chief Justice the head of the Presidential Elections Commission. After Mara'i completed his term of office, Mubarak continued to select Chief Justices who had not previously served on the SCC, appointing Maher Abd al-Wahid in 2006 and Farouk Sultan in 2009. Sultan not only had not served on the SCC, but had 'no background in constitutional issues...[and] his career has brought him through some of the more sordid parts of the Egyptian judicial apparatus—military courts [and] state security courts'.⁹⁹

After the overthrow of Mubarak in 2011, the SCC attempted to reclaim its control over judicial appointments. While the details of its interactions with the Supreme Council of the Armed Forces are unknown, it won an important concession from SCAF with the decree law on SCC appointments, which gives SCC judges an important role in appointments decisions and limits the President's choices regarding candidates (see Section 4.3.1.2). During the constitution-drafting process, the SCC indicated its displeasure with proposed articles on the judiciary by calling a press conference. It is unclear what influence this had on the drafters, but the final version of the Constitution says little about the SCC, deferring most important questions about the Court to future legislation.¹⁰⁰

However, the SCC also issued several highly controversial decisions regarding parliamentary elections during the transition and under the 2012 Constitution, which drew criticism from many quarters and demonstrated the disadvantages of mechanisms that create barriers between the Court, political actors and the general public. In June 2012, the SCC found the electoral law governing the first elections to the House of Representatives (the lower chamber of Egypt's Parliament) after the political transition invalid, on the grounds that the rules governing independent candidates had not been correctly applied in the elections. The SCC ordered SCAF to dissolve the body, which SCAF subsequently did, effectively leaving post-Mubarak Egypt without a fully functioning Parliament (the upper chamber, the Shura Council, was not affected by this ruling). The SCC's decision provoked strong public criticism from certain groups, particularly the Muslim Brotherhood. The Freedom and Justice Party, the political wing of the Muslim Brotherhood, won 235 seats (47.2 per cent) in the election in question and perceived the ruling as politically motivated in favour of secular political groups. Defenders of the Court argued that its decision simply fulfilled its mandate to uphold the Constitution.¹⁰¹

After the dissolution of Parliament and the passage of the 2012 Constitution, the SCC was again called upon to review an electoral law, because the 2012 Constitution required the Court to exercise mandatory prior review over election laws (Article 177). The SCC rejected a new electoral law drafted by the Shura Council for parliamentary elections, stating that several provisions of the law did not satisfy the 2012 Constitution, and returned the law to the Shura Council for revision. The SCC's rejection of the draft electoral laws meant that elections for a new House of Representatives under the 2012 Constitution were never held. The SCC issued another controversial decision in June 2013, ruling that the electoral law under which the Shura Council was elected was also unconstitutional. However, the Court did not order the dissolution of the Shura Council, finding that the 2012 Constitution granted the Council the power to legislate until new parliamentary elections were held. While the SCC's decisions on parliamentary elections were generally considered sound in terms of their legal reasoning, the Court drew criticism because its decisions appeared to consistently favour anti-Islamist interests over those of Islamist parties. Regardless of its motives, many perceived the Court as biased. In a transitional political environment that is radically different from the authoritarian regime under which the SCC was created, the vulnerabilities that result when a court's members have been selected without any participation by a broad cross-section of political parties from across the spectrum or the public became clear.102

The 2012 Constitution did not go into great detail regarding the structure and processes of the SCC, and has now been suspended. While it was in force, the SCC's appointments processes remained intact, but the Court was made smaller. Whereas the 1971 Constitution left the number of justices unspecified, the 2012 Constitution set the number at 11 (the Chief Justice and ten judges), forcing the removal of several sitting justices—which some perceived as a political move on the part of then-President Morsi and the Freedom and Justice Party to remove justices who were viewed as opponents and potential impediments to the legislation they were drafting.¹⁰³ As Egypt's transition continues to unfold, the SCC will need to continue to adapt to the changing political context.

4.3.2 Iraq

4.3.2.1 Iraq: historical and political context

While Iraq's modern judicial system dates back to the British mandate period following the fall of the Ottoman Empire, for the purposes of this report, this section will address the judiciary during the period of Saddam Hussein's rule and following the invasion of Iraq in 2003.¹⁰⁴

Before the Ba'ath Party came to power in 1968, the Iraqi judiciary was a relatively independent institution governed by the Council of Judges, which was headed by the President of the Court of Cassation, the highest court in Iraq. The Judicature Act, also known as the Judicial Authority Act, governed the Iraqi judiciary. It was intended to establish the judiciary as an independent government institution equal in stature to the executive and legislative branches. The Council of Judges oversaw all judicial appointments in this era. During the 1950s and 1960s, the Iraqi courts displayed a degree of judicial independence, ruling against the executive in a limited number of cases.¹⁰⁵

The Ba'athist Constitution and Saddam Hussein's rise to power in the late 1960s and 1970s marginalized the judiciary and ended any separation of judicial and executive power. In 1979, through legislation, Hussein abolished the Council of Judges and brought the Iraqi judiciary under the legal control of the Ministry of Justice, thereby effectively introducing executive control over judicial appointments (Ministry of Justice Act 101/1977). Hussein also established special revolutionary and military courts that were directly answerable to the executive rather than the Court of Cassation. These courts, staffed with Ba'ath Party members with no legal training, were established to deal with matters of state security and corruption. The revolutionary courts had the power to deny habeas corpus, and their decisions were final, with no right of appeal.¹⁰⁶

The judiciary's record under Hussein's regime is ambiguous. He generally avoided using the civil courts and relied on the revolutionary and security courts to support his regime, leaving the general judiciary essentially intact and led by experienced and independent judges. Yet some judges and lawyers supported the Ba'ath police state, which undermined Iraqi confidence in the judiciary as a whole. In addition, Hussein sometimes called on the regular courts to carry out his executive orders. Judges who resisted these efforts or refused to carry out executive orders were subject to intimidation, torture or forced retirement.¹⁰⁷

Following the 2003 invasion of Iraq and the fall of Hussein's regime, the United States and United Kingdom assumed the role of occupying powers and created the Americanled Coalition Provisional Authority (CPA). Among other measures, the CPA established a Judicial Review Committee, which vetted judges based on merit, Ba'athist affiliation and allegations of past corruption to determine their eligibility for continued judicial service. The Committee reviewed approximately 870 judges and dismissed approximately 20 per cent for incompetence, corruption or high-level Ba'athist membership. In September 2003 the CPA also formally re-established the Council of Judges, composed of the most senior judges on Iraq's highest courts and the directors of the legal and prosecutorial supervisory authorities, and gave it authority over Iraq's judicial and prosecutorial system (CPA Order No. 35). This action removed the judiciary from the Ministry of Justice's control, and was intended to help insulate it from executive pressure. $^{108}\,$

In March 2004, the Iraqi Governing Council adopted an interim Constitution, the Law of Administration for the State of Iraq for the Transitional Period (TAL), which established a system of government for Iraq during the transitional period and provided a framework for drafting the permanent Constitution. When the TAL came into effect on 28 June 2004, the CPA dissolved and transferred governing authority to the Iraqi Interim Government, headed by interim Prime Minister Ayad Allawi. The TAL continued efforts to reconstruct the Iraqi judiciary, including the creation of a Federal Supreme Court (FSC) with the power of judicial review (TAL Article 44). The drafters of the TAL included key leaders of Hussein-era opposition groups. The TAL, however, was criticized for being drafted by a body that did not fully represent Iraqi society, in a non-transparent process.¹⁰⁹

The drafters of the TAL were acutely aware of the role that courts can play in deciding high-stakes political matters. They also feared the creation of a supreme court that could, over time, usurp executive and legislative power and become an ideological force like Iran's Guardian Council. As a result, the drafters of the TAL deliberately crafted the provisions establishing the FSC in order to ensure the Court's moderation and preclude manipulation by political factions, focusing specifically on the composition of the Court. The appointments procedure set out in the TAL gave the Higher Juridical Council (HJC)-which replaced the Council of Judges and is composed of the presidents of Iraq's highest courts-the power to nominate a list of candidates to the FSC (TAL Articles 44, 45). The three-member Presidency Council was given the power to appoint judges from the list of nominees (TAL Article 44). The goal of this mechanism was to ensure that judicial appointments were based on candidates' professional abilities rather than political allegiances, while at the same time seeking to imbue the FSC with a sense of legitimacy and accountability. As discussed below in Section 4.3.2.2, this appointment mechanism remains in place, with Iraq's President replacing the transitional Presidency Council.¹¹⁰

The TAL was an interim constitutional document that was replaced by the 2005 Constitution, which preserved the new FSC (Article 92) and the Higher Juridical Council. However, while there was relatively little controversy over drafting the TAL's provisions on the FSC, the Court generated heated debates during the process of drafting the 2005 Constitution. Initially, religious groups wanted a constitutional council that would review laws prior to enactment to ensure they were in accordance with sharia law. Secular and Kurdish parties favoured an FSC with the power of constitutional review. The drafters were able to persuade the Shi'a Alliance to agree to the FSC rather than a constitutional council; however, the Shi'a parties also sought to secure a set number of seats on the FSC for Islamic jurists. The Kurdish and secular parties opposed this, fearing the idea of an FSC packed with sharia scholars.¹¹¹

Iraq's constitution-drafting process was complex and difficult, and its democratic legitimacy has been challenged by many Iraqis. An initial draft of the Constitution was formulated by a Constitutional Committee made up of 55 members, whose membership reflected the political groups represented in Iraq's Transnational National Assembly, the interim body governing the country until the passage of a new Constitution. However, the Committee was effectively dissolved when an ad hoc body referred to as the Leadership Council—a very small group that had the active participation of the US Embassy—took over drafting duties and produced a final draft Constitution without meaningful participation from the Constitutional Committee. This draft text was adopted in a referendum in October 2005.¹¹²

The final text of the 2005 Constitution does not specify the number of judges on the FSC. It only requires that the Court include experts in Islamic jurisprudence and legal scholars in addition to judges (Article 92). It defers the questions of the size of the Court and the precise mechanism for appointing its members to implementing legislation passed by a two-thirds supermajority of the Council of Representatives (Article 92). This framework gave Shi'a Islamists a symbolic victory by explicitly providing for Islamic experts on the Court, but also gave secular parties significant leverage over the implementing legislation through the supermajority requirement and the stipulation that at least one member of the Court must be a judge.¹¹³

The drafters' decision to defer so many of the issues necessary to the FSC's operation to future implementing legislation, and the requirement that the legislation be passed by a supermajority, have severely impeded the FSC's ability to establish itself as a credible and independent body. The Council of Representatives has thus far failed to pass the implementing legislation, creating uncertainty around the FSC's authority and leaving the question of its appointments procedure unanswered. Further complicating matters, the Council of Representatives in December 2012 passed the Federal Court Act, which would have separated the Higher Juridical Council (HJC) from the FSC by removing the FSC's Chief Justice as head of the HJC. However, the FSC struck down the Federal Court Act in September 2013, reinstating Chief Justice Medhat Al-Mahmoud as head of the HJC, and thus restoring his role in selecting candidates for nomination to the FSC.¹¹⁴

4.3.2.2 Iraq: appointments procedure

Given the continuing lack of implementing legislation following the passage of the 2005 Constitution, the FSC still operates under the relevant provisions of the TAL and Law No. 30/2005, which were passed during the transitional period of occupation, as well as the applicable provisions of the 2005 Constitution. The TAL provides that the FSC consists of nine members (Article 44). Under the TAL and Law No. 30/2005, the HJC, in consultation with the regional judicial councils, nominates three judges for every FSC vacancy (TAL Article 44; Law No. 30/2005 Article 3). The HJC includes the presidents of Iraq's highest courts and is headed by the President of the Court of Cassation (TAL Article 45). The President of the FSC also serves as the head of the HJC.¹¹⁵

Under the TAL, Iraq had a Presidency Council made up of three members, one to represent each of the major sectarian groups in Iraq: Shi'a, Sunni and Kurd. The Presidency Council considered the three nominees submitted to it by the HJC, and either had to reach a unanimous agreement on a nominee for the FSC or reject all three nominees and request a new list of nominees (TAL Article 44). The unanimous vote requirement was intended to ensure that the major sectarian groups all had a role in selecting the FSC's members. This appointments procedure purposely bypassed the National Assembly. To initially constitute the nine-member FSC, the HJC proposed 25 nominees, from which the Presidency Council selected nine members and named one of them, Medhat al-Mahmoud, Chief Justice. Under the 2005 Constitution, the Presidency Council continued to operate for the first legislative term after the Constitution's enactment, from 2006–10. At the end of that term, a President elected by the Council of Representatives replaced the Presidency Council (Articles 70, 138). The President has since taken on the role formerly played by the Presidency Council in appointing FSC judges.¹¹⁶

Iraq's Parliament, the Council of Representatives, has not yet passed the implementing legislation that will determine the permanent appointments procedure for the FSC under the 2005 Constitution. Scholars of the Iraqi legal system suggest that the delay is due in part to the ongoing debate over the inclusion of Islamic jurists on the FSC and to the supermajority required to pass the legislation, which in the current political environment is exceptionally difficult to reach. On 5 February 2011, the Council of Representatives held a first reading of proposed draft legislation for the FSC. The 2011 draft law would establish a 13-member FSC with a Chief Justice and eight other judges (for a total of nine judicial positions), and a four-member advisory board made up of two ordinary legal specialists and two Islamic law specialists. Under the draft law, for the nine judicial positions, the HJC would propose three candidates for each FSC position, from which the President would select one. This appointments procedure is very similar to that defined in the TAL. For the four members of the advisory board, the Ministry of Higher Education would nominate four civil law judges, and the two Islamic endowment authorities (one Sunni and one Shi'a) would nominate four Islamic judges. The Council of Ministers would then select two candidates from each group, and the Council of Representatives would approve the candidates selected by a majority vote. The 2011 draft law also stipulates that FSC judges must have at least 20 years of experience and would serve a six-year term. This appointments process, if enacted, would ensure the involvement of a wide range of actors in the selection of FSC appointees: the judiciary, the President, the Council of Ministers, the Council of Representatives, and both secular and Islamic legal scholars.¹¹⁷

4.3.2.3 Iraq: appointments process in practice

The FSC is in a precarious legal position. Its authority rests in the TAL, which has been largely superseded by the 2005 Constitution. Due to the Council of Representatives' failure to pass implementing legislation, the FSC's jurisdiction, composition and appointments procedure all remain uncertain, even while the FSC tries to operate and exercise necessary judicial review. The appointments procedure established under the TAL was intended to put experienced civil law judges on the Court who would be less influenced by, or better able to withstand the pressures of, Iraq's complicated and tense transitional politics. Indeed, the FSC has been called upon to decide many controversial political issues. Their rulings have led critics to suggest that the FSC, in particular Chief Justice Medhat al-Mahmoud, is loyal to Prime Minister Nouri al-Maliki and that its decisions are unjustifiably favourable to the current executive. Supporters of the Court argue that its rulings are based on legally defensible positions and reflect judicial restraint in entering into highly charged political disputes.¹¹⁸

One example of a controversial FSC ruling resulted from the 2010 parliamentary election, in which the Iraqiyya coalition, led by Ayad Allawi, narrowly won the highest number of seats in the Council of Representatives. Iraqiyya won two more seats than the State of Law Coalition, led by Nouri al-Maliki. Article 76(1) of the 2005 Constitution states: 'The President of the Republic shall charge the nominee of the largest Council of Representatives bloc with the formation of the Council of Ministers within fifteen days from the date of the election of the President of the Republic.' The legal question presented by the elections was who, according to Article 76, was empowered to form a government. Article 76 does not define the term 'bloc'. Iraqiyya argued that Article 76 should grant the right to form a government to the party that won the largest number of seats; the State of Law coalition argued that the provision should be interpreted to mean that the right to form a government fell to whichever parties could form the largest *coalition* after the election, even if the coalition did not include the *party* that won the largest number of seats. The FSC adopted State of Law's interpretation. The decision left open the possibility that either Maliki or Allawi could form a government, depending on who could win support amounting to the greatest number of seats in the Council of Representatives.¹¹⁹

Both in Iraq and abroad, the FSC's decision was perceived as politically motivated. Allawi publicly dismissed the FSC opinion and challenged the FSC's jurisdiction, questioning the Court's legitimacy because it was implemented under the TAL rather than the Constitution. Furthermore, some saw the decision as anti-democratic since Iraqiyya had won more seats in the election. There are, however, numerous legal and practical arguments in favour of the FSC's decision. Iraq is a parliamentary system, which means that the government derives its right to govern from parliamentary approval, not simply from gaining the most votes in an election. The FSC's opinion notes that the Prime Minister must survive a vote of confidence from the Council of Representatives, and therefore, a post-electoral majority alliance will be necessary to form a government from a practical standpoint. Furthermore, allegations that the FSC decision was biased toward Maliki should also be considered in light of the fact that the FSC certified Iraqiyya's electoral victory despite Maliki's challenge to the electoral results.¹²⁰

The FSC again drew accusations of bias toward Maliki in a subsequent ruling related to its decision on the 2010 elections. After the FSC's decision, the various political parties entered into negotiations in order to ultimately form coalitions, the largest of which would win the right to form the new government. These negotiations failed to produce widespread agreement, leading to a stalled Council of Representatives and no government. The 2005 Constitution requires the Council to elect a Speaker during its first session and the President of the Republic within 30 days of the first session (Articles 55, 72(2)). In October 2010, the FSC ruled the Council of Representatives' refusal to elect a Speaker and President unconstitutional, because this deadline had passed without an election for either position. At the time of the FSC's ruling, Maliki's party claimed to be close to forming a majority coalition. The FSC's ruling may have further favoured Maliki and hindered Allawi because it forced an end to negotiations among the various parties that, if allowed to continue, may have eventually included the Iraqiyya party in the final arrangement. Maliki was able to gain sufficient support to form the largest coalition, and proceeded to form a government in the wake of the FSC's ruling.121

Another FSC decision regarding executive powers exhibits troubling signs of bias toward the executive—and, by implication, the governing party. The case focused on a number of independent commissions established in Iraq's Constitution, including the Independent High Electoral Commission (IHEC). The 2005 Constitution provides that some of these commissions fall under the oversight of the Council of Representatives, and others under the executive; most significantly, the electoral commission is placed under the 'monitoring' of the Council of Representatives (Article 102). In December 2010, the Prime Minister's office requested that the FSC review the constitutional provisions on the independent commissions and their oversight. The Court's ruling, issued in January 2011, declares that the work of the independent commissions is more executive in nature, and thus the commissions should *all* fall under

the executive's authority (not that of the legislature) in order to avoid a violation of the principle of the separation of powers. 122

This ruling is difficult to defend from a legal perspective, given the clarity of the 2005 Constitution regarding which branch of government oversees the commissions. It is also politically significant, given the importance of the IHEC. This decision could give Prime Minister Maliki direct influence over the body that will oversee future elections, including his own, although to date the IHEC remains relatively independent. The FSC's decision was widely criticized, and according to some scholars it prompted the Council of Representatives to fast-track legislation on the FSC's appointments procedures, leading to the first reading of the draft law on the FSC on 5 February 2011 (see Section 4.3.2.2).¹²³

4.3.3 The judiciary-executive model: key constitutional considerations (low level of political investment)

The judiciary-executive model excludes many political actors from the judicial appointments process. In most variations of the model, only the judiciary—and often, only the highest levels of the judiciary—and the executive are involved in appointments decisions. The legislature, legal academia, civil society and often a large portion of the judiciary, do not participate at all in deciding who will sit on the constitutional court. Using this model makes it more difficult to create a sense of political investment in the court among the many actors that are excluded, especially opposition political parties, given the partisan character of the disputes resolved by constitutional courts, as Egypt and Iraq illustrate.

The judiciary-executive model may be suited to certain political contexts, particularly non-democratic regimes, in which insulating the constitutional court from politics may serve to protect the court's independence. The judiciary-executive model was used in Egypt under a long period of autocratic rule. The model, combined with an informal practice that allowed the judiciary to essentially control the entire appointments process for many years, allowed Egypt's SCC to maintain a degree of independence, including issuing rulings that constrained executive power, under an authoritarian regime.

This model may also be attractive in countries with weak legislatures that are prone to extreme polarization and fragmentation. Legislative politics tend to be dominated by party rivalries and short-term political concerns. The judiciary-executive model specifically excludes the legislature from appointing judges to the constitutional court, which may help to ensure that judicial appointments are not delayed by legislative deadlock and that political parties will not use appointments to pack the constitutional court with sympathetic judges.

However, the experiences of Egypt and Iraq illustrate the difficulties of employing the judiciary-executive model. Both countries are currently experiencing difficult political transitions. In Egypt, the judiciary remained largely intact after the end of authoritarian rule. As a result, many political parties, as well as many members of the public, view the judiciary as a remnant of the former regime, and rightly or wrongly will disregard its decisions as biased. In Iraq, the situation is somewhat different. Iraq carried out a controversial vetting process to remove judges who were considered too tainted by Hussein's regime to continue in their posts and created a new FSC. However, the FSC's provisional appointments mechanism follows the executive-judiciary model and therefore excludes any involvement by opposition political parties, even those that have significant numbers of seats in the Council of Representatives. But although the Iraqi context is different, the result is the same: a lack of political investment in the FSC from across the political spectrum.

In this context, the use of the judiciary-executive model has produced constitutional courts that are vulnerable to accusations of elite capture. Iraq's FSC has issued several controversial decisions in favour of Prime Minister Maliki, leading many to conclude that it is not impartial and must be reformed. While the Egyptian SCC was viewed as somewhat successful under a non-democratic regime, it has lost credibility and accountability during the transition; the changing political landscape has created new demands and public expectations of democratic accountability and transparency in the judiciary. In both situations, it may be advisable to adopt a different appointments process that allows the full range of post-authoritarian political constituencies to be involved in selecting constitutional court judges.

Vulnerability to executive capture

In authoritarian political contexts, the judiciary-executive model offers limited protection against elite capture of the judiciary. The Egyptian SCC relied for many years on an informal agreement that allowed it to control its own appointments. It was this informal practice, rather than the legal procedures set out for appointments, that allowed the SCC a measure of independence. However, Mubarak was able to discard this informal agreement as soon as he chose, and when he did, it effectively ended the SCC's ability to operate independently. The Iraqi example also indicates that under democratic rule, the judiciary-executive model may allow a president or prime minister to exercise (or be perceived as exercising) undue influence over the court.

Dangers of postponing important questions to implementing legislation

Iraq's experience also offers a lesson to policymakers in the process of drafting a new constitution. Iraq's Constitution drafters, caught in a dispute over the composition of the FSC and the selection of its members, ultimately deferred these questions to future

implementing legislation. However, in Iraq's polarized political environment, it has so far proven impossible for the Council of Representatives to reach a satisfactory compromise on the implementing legislation. The FSC is thus in a sort of legal limbo, operating under a largely defunct law from the occupation period.

While negotiations during the constitution-drafting process can be arduous, the process may provide the best opportunity to agree on the critical questions surrounding a constitutional court, especially regarding who will sit on the court and how they will be appointed. During the drafting process, all parties are aware of the need to compromise, and the deadline for producing a draft constitution may help push the parties toward an agreement. In contrast, the environment in a legislature is quite different: party representatives are more concerned with short-term needs and protecting the interests of their party and constituents than they are with the broader interests of the country. No single party in a legislature has a strong incentive to reach an agreement on the court, and there is no firm deadline for doing so, which may lead to long delays in the court's implementation, as has been the case in Iraq (and in Italy, as discussed below).

4.4 The multi-constituency model: Turkey and Italy

The multi-constituency model involves multiple institutions in the judicial appointments process, including the various branches of government and, in some countries, civil society organizations. In this model, the institutions involved may have direct or indirect power over appointments. Institutions with direct appointment power may select candidates and appoint them to the court without having to consult with, or gain the approval of, any other actor. Institutions with indirect power are generally given either the power to nominate one or a list of candidates for the court, or to approve or veto a candidate nominated by another institution; they do not have the power to both nominate and confirm a particular candidate. Most commonly, the seats on the court are divided between the various institutions with appointment power.

Under this model, the various institutions and branches of government that have a role in selecting the court's judges generally work independently of each other during the selection process; in other words, the different actors charged with appointing judges do not consult with one another when making their decisions. This distinguishes the multi-constituency model from the judicial council model, discussed earlier, in which a council composed of representatives from a range of governmental and nongovernmental bodies works together to select candidates for the court. Much like the judicial council model, however, the multi-constituency model's goal is to create an inclusive appointments process that involves many different constituencies in determining the best candidates for the court. The resulting court, it is hoped, will reflect the varying political and social forces in the country and promote a sense of political investment in the court on the part of all the actors involved in appointments.

Turkey and Italy use this model. Turkey adopted it in 2010, when it enacted constitutional amendments to significantly reform the Turkish Constitutional Court (TCC). The amendments changed the process of appointing justices to the TCC from a model dominated by the judiciary (which was indirectly influenced by the military) to a multi-constituency model that includes a more diverse pool of actors, including the judiciary, the Turkish Grand National Assembly, the President, civil society organizations and bar associations. The amendments also expanded the size of the TCC from 11 permanent and four alternate justices to 17 permanent justices. Proponents of these constitutional amendments argued that they would open up the appointments procedure in Turkey and allow more political parties, as well as non-political entities, greater influence over the composition of the court. By comparison, the Italian Constitutional Court, often seen as a leading example of the multi-constituency model, adopted this model at the time of the Court's formation after the end of World War II. The Italian Constitutional Court has 15 members; the President, Parliament and the highest courts in the judiciary each appoint five justices, following their own internal nomination and selection procedures.

4.4.1 Turkey

4.4.1.1 Turkey: historical and political context

To understand the history of the Turkish Constitutional Court (TCC), it is necessary to begin with a brief explanation of the formation of the modern Turkish state. Following the dissolution of the Ottoman Empire after World War I, Turkey began a transition to a republican political system led by the Turkish National Movement, and Mustafa Kemal (also known as Atatürk), the Turkish Republic's first President. He spearheaded the efforts to build a new Turkish state and institutionalized an official ideology for the new republic, which is generally referred to as Kemalism.

Under Kemalism, outward manifestations of religion, ethnicity or other non-Kemalist ideologies were considered subservient to the national goal of a unified state, and subject to state regulation or repression as appropriate. Kemalism remains the dominant ideology among many ruling and secular Turkish elites, particularly the military. However, supporters of political Islam have repeatedly attempted to form political parties to contest power and challenge Kemalism. In the last decade, Turkey has seen the rise of the Justice and Development Party (AKP), which has Islamic roots. Founded in 2002, it quickly came to power, winning a majority of seats in Turkey's parliament, the Grand National Assembly (GNA), in the 2002 elections. For many in Turkey, particularly the secular elites, the AKP and its supporters pose a direct challenge to the Kemalist legacy; understanding this dynamic is critical to understanding the current political landscape in Turkey and the TCC's role in this conflict.

The TCC was originally established in Turkey's 1961 Constitution, which was drafted after a military coup by a constituent assembly composed primarily of representatives from the military and the Republican People's Party (CHP), a Kemalist party aligned with the military. The 1961 Constitution employed a multi-constituency appointments model for the TCC, wherein appointment power over the 15 regular and five alternate justices was divided among the highest courts, the National Assembly, the Senate, the President, and (to a lesser degree) Turkey's universities and the Military Court of Cassation (Constitution of Turkey, 1961, Article 145). The TCC under the 1961 Constitution has been characterized as a powerful activist court, and its decisions on key political issues often defied the interests of the military, including rulings declaring military security courts unconstitutional and refusing to sanction a new state-sponsored body to curb the autonomy of Turkey's universities.¹²⁴

In 1980, Turkey's military staged another coup, citing frustration with the political status quo and civilian institutions deemed anathema to Kemalism. Seeking to solidify the military's dominance and enshrine secular, nationalist principles, the military annulled the 1961 Constitution and drafted a new one. This drafting process was heavily dominated by the military and did not include any participants with connections to political parties. The resulting 1982 Constitution promoted the development of Turkish civilization (according to the values of Kemalism) and entrenched the military's dominance over political institutions. The 1982 Constitution also brought the TCC firmly to heel, correcting what the military viewed as the Court's partisan role in politics. In order to ensure that the TCC would be sympathetic to its political aims, the military reformed the Court's appointments process. In a drastic departure from the model used under the 1961 Constitution, the military eliminated the legislature's role in TCC appointments, instead granting the President (who, from 1982-89, was the leader of the military coup, Kenan Evren) the power to appoint all TCC justices, selecting them from nominations made by the high courts, including the military courts, and the Council of Higher Education which was also influenced by the military (Constitution of Turkey, 1982 (prior to 2010 amendment), article 146). In short, the Turkish military replaced the 1961 Constitution's multi-constituency model with a variation of the judiciary-executive model that was shaped by military personnel and ideology.¹²⁵

Notably, the Supreme Board of Judges and Prosecutors (HSYK), made up of Ministry of Justice personnel and high court judges, controlled promotions to the high courts from which TCC justices were largely drawn under the 1982 Constitution. Because the HSYK was comprised of members sharing a pro-secular, pro-military ideology, 'the narrow composition of the HSYK facilitated the imposition of an ideological litmus test on judicial promotions, ensuring that the high judiciary [comprising most of the pool of

judges from which TCC judges could be selected] was a relatively politically homogeneous group'.¹²⁶

After this restructuring of the TCC, it fulfilled the role the military had intended for it as a protector of Kemalist values. The 1982 Constitution gave the TCC the explicit power to regulate political parties, including reviewing their budgets, and the power to order the closure of any party that violated the indivisibility of the state and its territorial integrity or the principles of a democratic and secular republic. Relying on these constitutional provisions, the TCC frequently, and controversially, shut down Kurdish and Islamist parties. In fact, the TCC's dissolution of certain political parties was so normalized that 'spare parties' would be formed for members of one party to join in case the original party was dissolved.¹²⁷

Kurdish parties were generally closed on the grounds that their platforms advocating greater Kurdish autonomy and better protection of cultural rights violated the Kemalist principles of Turkish unity and nationalism. Islamist parties enjoyed similar fates. The TCC closed five political parties affiliated with Islamism on the grounds that they violated the Constitution's secularity requirements. These closures occurred even as electoral results indicated that many Turkish citizens supported these parties and had chosen them as their democratic representatives in the GNA; two of these five parties had won notably high proportions of seats in the GNA in the election prior to their closure by the TCC.¹²⁸

One of the successors to the parties closed by the TCC was the AKP, which has governed Turkey since 2002. Since its formation in 2002, the AKP won significant victories in parliamentary elections in 2002 and 2007, and in 2011 won a near-absolute majority of all votes cast, earning it 327 of the 550 seats in the GNA. The TCC became a key institutional platform for Kemalist elites to challenge the AKP's growing dominance. Despite the AKP's repeated victories in democratic elections, several legal challenges were mounted in an effort to thwart the party's aims and, ultimately, to close it.

One of these challenges involved the selection of the Turkish President, who at that time was elected by the GNA according to the 1982 Constitution (article 102). When it came time to elect a new President in 2007, the AKP controlled enough seats in the GNA to elect its chosen candidate, Abdullah Gül, an observant Muslim who is considered a political moderate. In an attempt to prevent the AKP from electing Gül, the Kemalist-aligned Republican People's Party (CHP) brought a challenge before the TCC arguing that a vote to select a President required a 'super quorum' of at least two thirds of the members of the GNA. There was no such rule in the 1982 Constitution, and 'the CHP's interpretation would mean that opposition groups unable to seat their own preferred candidate would be able to indefinitely postpone or derail the appointment of another candidate so long as they could persuade one third of MPs to boycott the vote'. Nevertheless, the TCC ruled in the CHP's favour, reading a two thirds voting quorum into the Constitution for the first time.¹²⁹

In response, the AKP called early elections in 2007 and won a landslide victory. However, other efforts were made to prevent the AKP from governing. A substantial number of Turkish citizens saw the AKP as a threat to Turkey's constitutional commitment to secularism. In 2008, Turkey's chief prosecutor sought the closure of the AKP before the TCC, arguing that the party had become a focal point for anti-secular activities. Most of the evidence presented in support of the closure related to the AKP's efforts to end Turkey's ban on headscarves. The case 'was the first time a *sitting* elected government was threatened with ouster by judicial action', and many characterized the actions of the TCC and the prosecutor as a 'judicial coup'. The court nearly unanimously found the AKP guilty of anti-secular activities. But the TCC declined to order the AKP's closure because only six of the 11 judges voted for closure, just short of the two-thirds majority of seven judges required to dissolve a political party. Instead, the TCC penalized the AKP by reducing its state funding.¹³⁰

In response to what was seen as the TCC's assertion of power over the political sphere, the AKP introduced a package of constitutional amendments in 2010 that, among other changes, sought to change the process of appointing judges to the Court. By 2010, a popular movement had emerged in support of constitutional reform. A broad coalition of academics, journalists, politicians, NGO representatives and members of the public criticized the 1982 Constitution for its non-democratic origins and for hindering Turkey's effort to become part of the European Union.¹³¹

The 2010 amendments altered the constitutional framework of the TCC in several important ways:

- The number of TCC judges was increased from 11 permanent and four alternate judges to 17 permanent justices (Article 146, amended 2010).
- The GNA was given the power to appoint three of the 17 TCC judges (Ibid.).
- The pool of candidates from which TCC judges are selected was broadened (Ibid.). Prior to the amendments, seven of the 11 TCC judges were appointed from the five next-highest courts: the High Court of Appeals (Court of Cassation), the Council of State, the Military High Court of Appeals (Military Court of Cassation), the Military High Court of Administration and the Court of Accounts. Senior lawyers and administrators and members of the Council of Higher Education provided the pool for the remaining four. The 2010 amendments reduced the proportion of TCC judges drawn from the highest courts and made a wider range of judges, lawyers and administrators eligible to be TCC candidates.¹³²

• The composition of the Supreme Board of Judges and Prosecutors (HSYK), which controls judicial promotions to Turkey's appellate courts, was changed (Article 159, amended 2010). Prior to the amendments, the HSYK consisted of senior judges, the Minister of Justice and the Undersecretary to the Ministry. The amendments significantly expanded the size of the HSYK and supporters of the amendments argued that they made the body 'more representative' of the [judicial] profession at all levels'.¹³³

The constitutional amendments as a whole extended beyond changes to the appointment and composition of the TCC, and were the subject of heated political debate. While the amendments sought to liberalize the military-dominated 1982 Constitution, because they dismantled some of the privileges preserved for the military and secular elites, they were also treated as a clash between secular and Islamist forces. The increase in the number of TCC justices prompted accusations from opponents that the AKP was attempting to pack the Court. Opponents also criticized the AKP for putting the constitutional amendments to a referendum as a single package, so that voters were forced to approve the whole slate of amendments even if they only supported some. Others argued that the modifications to the TCC appointments, in part by involving the GNA in the process. The constitutional amendments ultimately passed in the referendum by a wide margin.¹³⁴

4.4.1.2 Turkey: appointments procedure

Prior to 2010, the 1982 Constitution provided for appointments using a variation of the judiciary-executive model, in which the executive appointed candidates nominated by the judiciary. The President appointed all 11 permanent judges and four alternate judges to the TCC (Constitution of Turkey, 1982 (prior to 2010 amendment), Article 146). Seven of the TCC's 11 permanent judges were drawn from Turkey's high courts, including the military high courts. The other four were selected from the Council of Higher Education, senior administrative officers and lawyers (Ibid.). This approach to appointments ensured that the judges appointed to the TCC would share a 'relative ideological conformity', and ultimately created a Court that many viewed as disconnected from popular opinion and democratic will.¹³⁵

Turkey adopted a multi-constituency model for appointments to the TCC in 2010. In particular, the 2010 constitutional amendments significantly changed the appointments process set up in the 1982 Constitution, including altering the number of judges on the court, their term length and the procedure for appointing them. The 1982 Constitution called for 11 permanent justices and four alternates. There was no set term limit for the judges, but retirement was mandated at 65 years (Articles 146, 147). Because the minimum age at which a judge could be appointed to the TCC was 40, a single judge

could theoretically serve for 25 years (Article 146). The 2010 amendments increased the number of judges to 17, all of whom are permanent. Judges now serve one non-renewable term of 12 years; the mandatory retirement age remains 65 (Article 147, amended 2010).

The multi-constituency model adopted by the 2010 amendments also created a broader pool from which TCC judges could be selected and introduced greater legislative influence over the appointments process. The President selects 14 of the 17 TCC judges from specific institutions and professional categories. Seven of these 14 come from Turkey's high courts: three from the High Court of Appeals (Court of Cassation), two from the Council of State, one from the Military High Court of Appeals (Military Court of Cassation) and one from the Military High Court of Administration. For each of these seven positions, the President appoints a candidate from a list of three judges nominated by the courts' plenary assemblies. The President appoints three of the 14 from candidates nominated by the Council of Higher Education; the Council nominates three candidates for each seat from a pool of legal academics, economists and political scientists. Finally, the President appoints four of the 14 from among lawyers, prosecutors and judges from the lower courts, and senior administrative officers (Article 146, amended 2010).

The GNA appoints the remaining three judges according to the following guidelines. The Court of Accounts submits a list of three candidates, selected from among its President and members, for each of two seats on the TCC. The heads of Turkey's bar associations submit a list of three candidates (who are self-employed lawyers) for the third seat. The GNA then votes to elect a judge to each of the three seats. In the first round of voting, a candidate must win a two-thirds majority to be appointed. If a candidate does not prevail in the first round, there is a second round of voting, in which a candidate must win an absolute majority. If no candidate succeeds in winning an absolute majority in the second round, there is a run-off between the two candidates who received the most votes; the winner of that run-off vote is appointed (Ibid.).

4.4.1.3 Turkey: appointments process in practice

Because the multi-constituency model has only been in place for a short time, it is not yet possible to accurately assess its impact on the TCC. Turkey is also considering drafting a new Constitution, which may introduce additional changes to the Court. Supporters of the changes to the TCC argue that it is now a more inclusive body and that it is formed through a more participatory judicial selection process. Opponents claim that the changes amount to court packing, and point to a 2012 TCC ruling that current President Abdullah Gül of the AKP may remain in office until 2014, and may then run for a new term, as proof that the TCC is now under the influence of the AKP. Yet assertions that the ruling, which resolved an ambiguity stemming from a 2007 constitutional amendment providing for the popular election of the President, was unfairly favourable to the AKP do not take into account the fact that the AKP opposed allowing Gül to run for a second term. As one major Turkish newspaper commented, the ruling 'failed to satisfy either the opposition parties or the [AKP] government as it runs counter to both sides' political plans'. In any case, the history of the TCC, and the adoption of the multi-constituency appointments model, provide a useful example for policymakers.¹³⁶ However, more recent AKP-led proposals for further judicial reforms have been widely criticized as an attack on judicial independence.

4.4.2 Italy

4.4.2.1 Italy: historical and political context

The Italian Constitutional Court was established during Italy's transition from fascism under Mussolini to a republic at the close of World War II. In 1946, a Constitutional Assembly was elected and tasked with drafting a new Constitution. The Assembly was divided among the major political parties of the era, including Communists and Socialists (the left wing of the political spectrum), the Christian Democrats, Liberals and Republicans (considered moderate), and right-wing parties, including the monarchists and a small group of neo-fascists. The left and moderate parties controlled the vast majority of seats in the Assembly, with the Communists and Socialists holding a combined 219 seats and the Christian Democrats, Liberals and Republicans holding 271 out of a total of 556.¹³⁷

Debates during the Constitutional Assembly

During the Constitutional Assembly's proceedings, the Christian Democrats supported the creation of a constitutional court with the power of judicial review to ensure that Parliament's actions complied with the proposed Constitution. Other arguments made in favour of a constitutional court were the need for a body capable of protecting and enforcing the long list of rights contained in the proposed Constitution. The left-wing alliance of Communists and Socialists, joined by the Liberals, opposed the creation of a constitutional court on the grounds that Parliament, as the elected representative of the Italian people, should be sovereign. The left also opposed the establishment of a constitutional court because it expected to win a majority in the upcoming parliamentary elections, and was against creating institutions that could limit its power once elected. For similar reasons, the Christian Democrats' support for a court was in part due to their anticipation of a possible left-wing victory at the polls; they saw a court as a form of 'political insurance' that could compel the ruling party to respect the Constitution. Ultimately, the argument that a constitutional court was needed to limit Parliament's powers proved persuasive.¹³⁸ When the Italian Constitutional Court was debated in the Constituent Assembly, several proposals were made with respect to how to appoint the Court's judges. The subcommittee responsible for producing the first draft of the Constitution initially suggested that the lower house of Parliament should elect the judges. A later version of the draft modified this proposal, providing for parliamentary elections of judges from candidate lists selected by the judiciary and law professors. Negotiations continued to expand the number and types of political actors with a role in the appointments process. Left-wing parties in the Constitutional Assembly wanted to ensure that appointments to the Court were sufficiently democratically accountable by involving three different political constituencies: the judiciary, the Parliament and the President. Italy's President is elected by Parliament, rather than directly by the people, and is explicitly designated in the Constitution as a representative of national unity, thus bringing a theoretically non-partisan element to the appointments process (Article 87).¹³⁹

The final version of the Italian Constitution, which took effect in 1948, provided for a Constitutional Court of 15 judges, of whom five are appointed by the President, five by the Parliament in a joint sitting of both houses and five by the highest courts of the judiciary (Article 135). Several important questions regarding the details of the appointments procedure—including how the judiciary would make its appointments to the Court and what majority would be required in Parliament to make its appointments—were left to Parliament to decide in future implementing legislation.

Debates in Parliament

The results of Italy's first parliamentary elections in 1948 'shifted the political landscape and disproved the fears and ambitions of the two sides during the Constituent Assembly', causing a subsequent shift in the positions of the major political parties regarding the Italian Constitutional Court and significantly delaying the process of adopting the implementing legislation necessary to enable appointments to the Court to be made. The Christian Democrats won an absolute majority of seats in Parliament and reversed the position they took in the Constituent Assembly; they became less eager to pass the legislation necessary for the Court to begin functioning once they took control of the government. The Communists and Socialists, finding themselves in the political opposition, eventually came to support the Constitutional Court. Subsequent battles in Parliament over the implementing legislation, including disputes over the details of the appointment procedures, delayed the passage of this legislation until 1953. The first set of judges for the Constitutional Court was not sworn into office until December 1955.¹⁴⁰

4.4.2.2 Italy: appointments procedure

According to the Italian Constitution, the Constitutional Court consists of 15 judges, each of whom serves a non-renewable nine-year term. Appointment of these 15 judges is divided equally among the President, Parliament and the highest courts: each appoints five judges (Article 135).¹⁴¹

The implementing legislation that was finally passed in 1953 further specified the appointments procedure set out in the Constitution with respect to the judges appointed by the judiciary (five) and Parliament (five). The five appointments to the Constitutional Court made by the judiciary are subdivided among the three highest ordinary and administrative courts: the Court of Cassation (which elects three judges), the Council of State and Court of Accounts (which each elect one judge). Each of these courts forms a panel, whose members are specified by law, to appoint the judges, and the panels approve appointments by an absolute majority vote. For the five parliamentary appointments, the implementing legislation adopted a supermajority requirement. A judge must win election by a three-fifths majority of the total number of members of Parliament. If this majority is not obtained on the first ballot, the majority requirement is lowered to three-fifths of those voting for subsequent ballots. This supermajority requirement was motivated in part by explicitly political concerns. The ruling Christian Democrats wanted to prevent the Communists from being able to appoint a candidate to the Court. A three-fifths majority requirement would mean, in theory, that political parties would have to work together to find a candidate acceptable to multiple parties. The calculation was that in a worst-case scenario, two-fifths of Parliament would be opposed to a Communist candidate for the Constitutional Court. However, the Communists were able to negotiate for one seat on the Constitutional Court after subsequent elections reduced the Christian Democrats' majority in Parliament.142

The supermajority requirement was raised to two-thirds in 1967, thus further encouraging the nomination of candidates who could command the support of multiple political parties. If no one has been elected after three voting rounds, the majority needed is lowered to three-fifths.¹⁴³

4.4.2.3 Italy: appointments process in practice

In practice, the Italian appointments process has functioned relatively smoothly. Appointments to the Constitutional Court made by the high courts are quick, and an informal norm has developed by which the judges generally appoint presidents of sections of the high courts to the Court. The supermajority requirement for Parliament's appointees to the Constitutional Court at first caused considerable delays in selecting judges. After the 1953 parliamentary elections, no one party held a three-

fifths majority in Parliament. It took the political parties three years to establish an informal compromise, which would hold until 1994: the five seats selected by Parliament were divided among parties based on the number of parliamentary seats each held. Under this informal arrangement, the Christian Democrats appointed two of the five judges, the Socialists and Communists each appointed one, and the final appointment rotated among the smaller parties.¹⁴⁴

The three-year delay was also due in part to a dispute over presidential appointments, which were not addressed in the 1953 implementing legislation. In the Constitutional Court's first years, the President and the government argued over whether the President's appointments required a government countersignature. After President Giovanni Gronchi made the Court's first five presidential appointments without seeking the approval of the government, 'the precedent was set that presidents named individuals to the Court without formal consultation'. Although the President is seen more as a statesman than as a political actor, and thus his judicial appointments are perceived as less politically motivated than those of Parliament, presidential appointments have 'paralleled the partisan affiliations of the parliamentary appointees'. The President generally appoints pre-eminent legal scholars who are perceived as political centrists, rather than judges or lawyers. The President has also made appointments with symbolic significance, such as appointing Fernanda Contri as the first female Constitutional Court justice in 1996.¹⁴⁵

Parliament's informal agreement to allocate seats according to party representation helped the appointments process work effectively for several decades. As in Germany, the supermajority requirement and the tacit understanding among parties in Parliament has also meant that candidates who hold 'extreme' views are not typically chosen. However, the importance of this informal agreement, and its inherent vulnerability, were highlighted in the mid-1990s. In 1992, prosecutorial investigations into corruption resulted in almost all the officials in power losing their seats in the election following the investigation. Italy's political parties underwent a drastic reorganization, with new parties emerging and some older parties changing their names in the process. Since the new political parties elected to Parliament had no informal agreement regarding Constitutional Court appointments, three seats on the Court remained vacant until 1996 because Parliament could not agree on a candidate who could obtain the necessary two-thirds majority vote. The seats were finally filled when the new parties represented in Parliament agreed to allocate the appointments among them in a similar manner to the prior agreement: the right (Forza Italia) controls two appointments, the left (the Democratic Party of the Left and its allies) controls two, and the final seat rotates among the remaining minority parties.¹⁴⁶

Generally, the Italian Constitutional Court has a low political profile. Its decisions are rarely at the centre of mainstream political discussions, and there is relatively little animosity between the Court and the legislative and executive branches. For the first 30 years of its rule, the Constitutional Court was reluctant to get involved in disputes among the branches of the central government or in matters of an overtly partisan character. Instead of risking the appearance of partiality to any one institution or political party, the Constitutional Court has abstained from hearing certain cases, and intervenes in high-level political disputes only when it perceives a serious threat to the constitutional order. However, as explained below, this changed after Silvio Berlusconi came to power.¹⁴⁷

Moreover, the Constitutional Court has, on many occasions, issued bold decisions with far-reaching consequences for the government in power and which have protected the political opposition. In its very first case, the Constitutional Court issued a decision that affirmed its own authority and declared its independence. At issue were some of the Fascist-era security laws requiring government authorization to distribute flyers or newspapers, or to hold a public demonstration using sound amplifying equipment. In Parliament and the press, those hostile to the idea of judicial review argued that the Constitutional Court lacked the authority to determine the constitutionality of laws passed prior to the 1948 Constitution. This argument 'was an undisguised attempt to preserve the entire corpus of Fascist law'. The Christian Democrats, in fact, were using the security laws in the early 1950s to repress workers' demonstrations from the left of the political spectrum. The Constitutional Court's ruling firmly asserted its authority to review all legislation in force, regardless of its date of origin, and struck down the public security laws for violating the Italian Constitution's guarantee of freedom of expression, thereby protecting the right to political expression of the government's political opponents. The Constitutional Court proceeded to systematically dismantle the fascistera legal corpus. These rulings were popular with the public, and shored up the Constitutional Court's support base.¹⁴⁸

The Court's overall policy of avoiding involvement in high-stakes political issues has helped it avoid serious criticism from the other political branches and the public. In fact, during the corruption investigations in 1992 that brought many politicians under criminal investigation, the Constitutional Court was 'the only national institution that remain[ed] untarnished'. However, the rise of Silvio Berlusconi, a controversial politician and businessman who served as Prime Minister several times, brought about a confrontation that the Constitutional Court could not avoid.¹⁴⁹

Berlusconi repeatedly clashed with the Constitutional Court. In 2003, he proposed constitutional amendments related to the Court's composition and appointments process. The amendments would have decreased the number of judges appointed by the President and the judiciary from five to four each and increased the number of parliamentary appointees from five to seven. The amendments would also have changed the method by which Parliament selected its nominees: instead of being selected by

both houses of Parliament in joint sitting, the reform proposed that a Federal Senate (a new body that was proposed in the reform package that would have represented Italy's regional governments) would make the appointments. The result of these constitutional amendments would have been to increase the number of appointees selected by political parties rather than by actors perceived as less politically charged (the judiciary and the President) and to give the federal regions more influence over appointments, which would have had implications for the Court's consideration of cases involving conflicts between different regions or between a region and the central government.¹⁵⁰

Critics called the proposed constitutional amendments an attack on the Constitutional Court's independence. After the package of reforms failed to pass in Parliament by the required two-thirds majority, Berlusconi unsuccessfully attempted to secure their adoption through a constitutional referendum held in 2006. His attempts to alter the composition of, and appointments to, the Constitutional Court occurred around the same time that he was engaged in an effort to delay or prevent his prosecution on various criminal charges, including fraud and bribery. While serving as Prime Minister from 2001-06, and again from 2008-11, Berlusconi secured passage in Parliament of laws that would have effectively protected him from prosecution, either by granting high-ranking political officials immunity from prosecution while in office or by delaying the start of proceedings against officials while they held office. On several occasions, the Constitutional Court struck down all or part of these laws. In 2009, after the second Constitutional Court judgement (which found one of these laws unconstitutional on the grounds that it violated the principle of equality before the law), Berlusconi lashed out against the Court and called for judicial reform. In 2011, when the Court overturned parts of another law that would have suspended criminal proceedings against him while he remained in office, Berlusconi stated that he would seek reforms of the Court, including requiring a two-thirds majority vote of the 15-member Court in order to find a law unconstitutional. These hostile proposals to change the Constitutional Court stalled after Berlusconi lost power in November 2011.151

4.4.3 The multi-constituency model: key constitutional considerations (high level of political investment)

The multi-constituency model provides perhaps the greatest opportunity of any of the models reviewed in this report to promote a constitutional court with judges who represent a diverse cross-section of the political spectrum and, more generally, a broad variety of stakeholders. A multi-constituency model can include as many different constituencies as constitutional drafters choose—including (but not limited to) elected and unelected branches of government, educational councils, bar associations, legal academia and NGOs. Furthermore, the model can be designed to guarantee that each constituency successfully places candidates on the constitutional court, either indirectly or directly. In Turkey, various constituencies produce lists of nominees, from which the

President or the GNA must choose the judges to appoint to the TCC. In Italy, each branch of government is constitutionally guaranteed the right to appoint one third of the Italian Constitutional Court's judges. This is in stark contrast to the judicial council model, in which various constituencies are represented on the council that selects judicial candidates, but no constituency is guaranteed that a nominee of their choice will be appointed to the court.

The involvement of various branches of government and civil society organizations can have several positive consequences for the stability of the constitutional order. Smaller political parties or minority interests have more ways in which to influence the court's composition in a multi-constituency model. Different actors may be more willing to abide by the court's rulings, even when the rulings hinder their current interests, because they are politically invested in the court through their role in selecting the justices. A diverse constitutional court is likely to foster moderate viewpoints and decisions, since compromise among many different opinions will be necessary to issue a final decision in a case. Furthermore, the public is likely to perceive a diverse constitutional court as more independent and less influenced by any one political actor or ideology.¹⁵²

The multi-constituency model provides an opportunity to directly engage many different political and non-governmental actors in the constitutional court appointments process. Italy and Turkey have designed their appointments processes somewhat differently, but both approaches reflect an effort to involve many different political forces in the selection of constitutional court judges.

Italy's decision to divide appointments equally among the three branches of government—the presidency, Parliament and judiciary—appears to have fostered strong support for its Constitutional Court from the elected branches of government. All three branches of government have an equal role in selecting the Court's judges, which creates a sense of fairness and balance. The Italian President's position as a representative of national unity, coupled with political judgement, has meant that presidential appointments to the Court also contribute to its reputation as non-partisan and impartial. Finally, the inclusion of the judiciary in the process increases the likelihood that the most highly qualified members of the judiciary are selected for the Court.

Turkey adopted the multi-constituency model too recently to determine its impact on the TCC, or whether it has contributed to a sense of political investment in the TCC among a broader range of political actors and the public. However, the move to a multiconstituency appointments model in 2010 has expanded the pool of candidates eligible for appointment to the TCC and allows more government institutions and civil society organizations to play a role in determining who will sit on the Court. Turkey's new multi-constituency appointments process is sometimes criticized for not including enough constituencies. While it does allow a broader range of actors to participate in the appointments process, that range is still somewhat narrow. The President selects 14 of the TCC's 17 members from a pool of candidates largely drawn from the judiciary. At first glance, this does not appear to be much of a departure from the pre-2010 judiciary-executive model used for TCC appointments. However, the pool of candidates now eligible for appointment is much broader than it was prior to the 2010 reforms. Yet the three judges selected by Turkey's legislature are also drawn from a relatively narrow pool of nominations from the Court of Accounts and Turkey's bar associations. Thus, some scholars suggest that the new appointments process may not seriously alter the TCC's composition, and will preserve executive control over TCC appointments. The inclusion of two military judges on the TCC has also been controversial; it raises questions regarding whether the military should play any role in the judiciary in a democracy and whether such judges can be impartial, considering that they may choose to return to the military justice system after their term on the Court. For similar reasons, countries in the MENA region should consider insulating the constitutional court appointments process from military influence, given the controversial role that some military forces in the region play in political issues.¹⁵³

Potential danger in allocating appointments to the executive

In Italy and Turkey, the executive unilaterally appoints a certain number of judges to the constitutional court: five in Italy and four in Turkey. Allowing the executive to select constitutional court judges without any consultation or oversight from the other branches of government carries an obvious risk, as the executive has a strong interest in selecting judges who are inclined to uphold its policies. In Italy, this danger has been mitigated by the fact that the President's role is to act as a figure of national unity, and because he or she is not popularly elected. In other nations, this may not be the case. Given the MENA region's history of executives who attempt to use their appointment powers to influence constitutional courts, policymakers should consider constraining the executive's role in a multi-constituency model, either by allowing a different political actor to review and ultimately approve its nominations to the court or by requiring the executive to make its allotted appointments from a shortlist of candidates selected by a different political actor.

Potential for deadlock in legislative appointments

Italy, like Germany, has developed an informal agreement among the parties represented in the legislature to allow parties to 'control' a certain number of seats on the court, in proportion to their representation in Parliament. The Italian case shows the risk inherent in such informal agreements. In 1994, when the balance of power in Parliament shifted dramatically, the informal arrangement collapsed, resulting in years of deadlock as the new parties represented in Parliament fought over appointments to the Constitutional Court. In countries transitioning away from authoritarianism, like some MENA region countries, political parties may be weak and fragmented. In this situation, relying on parliament to select some or all of the constitutional court's members carries the risk that court appointments may be delayed by parties' inability to agree on nominees.

5 Judicial qualifications and removal procedures

Specifying the qualifications that constitutional court judges must hold is another way to ensure political investment from across the political spectrum. Setting out the level of education and professional achievement judges must have obtained, or specifying a minimum or maximum age at the time of appointment, ensures that the judges appointed to the court will have the necessary expertise to parse the complex and politically significant constitutional questions that will come before the court. But it also creates an additional barrier to court packing: a political actor or party seeking to place its supporters on the court will have to ensure that its candidates possess the minimum qualifications specified in the constitution. Qualifications may also include a list of professions or offices that are incompatible with appointment to the constitutional court, usually political positions, which can help to insulate the court from political influence. Some countries also require the court to include, at any given time, a set number of members with experience in the judicial sector (for example, a set number of career judges). This section briefly compares the age restrictions and professional or legal qualifications stipulated by the six countries discussed in this report.

Another important issue for the design of constitutional courts is the procedures for removing sitting judges from the bench. Removal and appointment procedures are mutually reinforcing. If it is easy for one political actor to remove judges, the removal mechanism can be used to undermine even the best-designed appointments process by enabling the manipulation of the court's membership. Indeed, the threat of removal can be used to influence the judges. It should be noted, however, that it is relatively rare for a country to remove a sitting constitutional court judge. It is more common for political actors to seek to influence the court through the appointment process than to remove a judge with whom they disagree.

5.1 Judicial qualifications

Table A compares the judicial qualifications for constitutional court judges in Egypt, Germany, Iraq, Italy, South Africa and Turkey. Four of these six (Germany, Italy, South Africa and Turkey) appoint constitutional court judges for a defined, nonrenewable term. Egypt and Iraq do not specify a term length; Egypt sets a mandatory retirement age for judges, while Iraq only states the minimum and maximum ages a judge must be when appointed. Most countries also provide fairly specific professional qualifications that judges must hold to be appointed, including a high level of achievement as a judge, lawyer or academic. These professional qualifications limit the range of individuals that may be selected as a constitutional court judge. Iraq and South Africa are less specific regarding the qualifications required, which creates a broader pool of potential court nominees.

Country	Age restrictions	Professional/legal
		experience
Egypt ¹⁵⁴	No younger than 45; mandatory retirement at 70.	experienceJudges may be selected fromthe following groups:Members of the SupremeCourt;155Judiciary: Current or formermembers of the judicialbodies holding the rank ofcounsellor or its equivalentfor at least five consecutiveyears;Academics: Current or formerlaw professors who have heldthe position of a professor atan Egyptian university for atleast eight consecutive years;Lawyers: Must havepractised before the Court ofCassation or the HighAdministrative Court for at
Germany ¹⁵⁶	No younger than 40; mandatory retirement at 68; judges may only serve one 12- year, non-renewable term.	least ten consecutive years. Judges must meet the basic qualifications for judicial office: successful completion of legal studies and a subsequent period of preparatory training. FCC judges may not simultaneously hold office in the legislative or executive branch and must not have any other professional occupation except professor of law. Six of the 16 FCC members must be selected from among judges sitting on the highest ordinary and administrative courts (career judges).

Table A. Age and experience requirements for constitutional courts

Iraq ¹⁵⁷	No younger than 28 or older than 45.	Degree from Iraq's Judicial Institute and three years' legal experience, or appointment by presidential order and ten years legal experience.
Italy ¹⁵⁸	No specific age requirements, but judges may only serve one nine-year, non-renewable term.	Must be a judge on one of Italy's higher courts (ordinary or administrative), a full professor of law or a lawyer with 20 years' experience in practice. Judges may not simultaneously be members of Parliament or a Regional Council.
South Africa ¹⁵⁹	No specific age requirements, but judges may only serve one non-renewable 12-year term, and must retire at 70. In certain cases a judge may remain on the bench for 15 years, or until he or she reaches the age of 75, if necessary to fulfil the legal requirement of 15 years of active service. ¹⁶⁰	Judges must be 'appropriately qualified' and 'fit and proper'. The Court must always have at least four members who were judges at the time they were appointed.
Turkey ¹⁶¹	No younger than 45; mandatory retirement at 65. Judges may only serve one 12- year, non-renewable term.	Judges may be selected from the following groups: <i>Academics</i> : Must be an associate professor or professor. <i>Lawyers</i> : Must have practiced for at least 20 years. <i>Government officials</i> : Must have completed higher education and worked in public service for 20 years. <i>Judges and prosecutors</i> : Must have at least 20 years of work experience.

5.2 Removal procedures

All six of the countries reviewed in this report give the judiciary a key role in determining whether or not a constitutional court judge may be removed from the bench. In Egypt, Germany and Italy, the court itself must vote in favour of removing a judge. Iraq and South Africa both require that several different branches of government, including the judiciary, agree upon a judge's removal before it can be carried out. Turkey makes removal automatic for judges convicted of certain offences, but requires a vote of the Constitutional Court to remove a judge due to incapacity. Notably, five of the six countries reviewed here ensure that the power to remove a judge is either divided among several different political actors or is not granted to the same actor that is in charge of the appointments process, thus reducing the risk that any one political actor can capture the Court. Iraq's removal procedure is very similar to its appointments procedure, but because three different political actors must approve a judge's removal from the FSC, the risk is somewhat mitigated that a judge could be removed for political reasons rather than reasons related to his or her competence or fitness for service.

5.2.1 Egypt

The Supreme Constitutional Court is responsible for disciplining its members. At the request of the Chief Justice, a committee will investigate a judge accused of being untrustworthy or derelict in his or her duties. During the investigation, the member is put on mandatory leave with salary. After a hearing, the General Assembly of the Court decides whether the member must resign. Their decision is final and irrevocable.¹⁶²

5.2.2 Germany

The Federal Constitutional Court controls the discipline and removal of its judges. According to Section 105 of the FCC Act, FCC judges may be involuntarily dismissed only if they are permanently unfit for service, have been sentenced without appeal because of a 'dishonourable act' or to over six months' imprisonment, or if they have committed a 'gross breach of duty'. A dismissal requires a two-thirds majority vote of the FCC's two chambers sitting in plenary, after which the Federal President formally dismisses the judge. No FCC judge has been removed to date.¹⁶³

5.2.3 Iraq

Under the Transitional Administrative Law, no judge may be removed unless he or she is convicted of a crime involving moral turpitude or corruption or suffers permanent incapacity (TAL Article 47). To remove a justice, the Higher Juridical Council must recommend removal, the Council of Ministers must decide on removal and the Presidency Council (now the President) must approve the removal (TAL Article 47). If a judge is accused of a crime involving moral turpitude or corruption, he or she shall be suspended from work until the case is adjudicated (TAL Article 47). Iraq's 2005 Constitution also limits judicial removal; however, the exact causes for removal and discipline, as well as the removal procedures, are deferred to implementing legislation, which has not yet been drafted (Article 97).

5.2.4 Italy

The Italian Constitutional Court controls the process of removing its judges. No judge may be removed except by a vote of two thirds of the Court itself. Judges may only be removed due to incapacity to perform their duties or for gross misconduct.¹⁶⁴

5.2.5 South Africa

South Africa requires a two-step process to remove a Constitutional Court judge. First, the Judicial Service Commission must find that the judge suffers from incapacity, is grossly incompetent or is guilty of gross misconduct. Second, the JSC's ruling must be referred to the National Assembly; a judge's removal requires a two-thirds majority vote. Once both of these steps have been completed, the President of South Africa must remove the judge (Article 177). This process has not been used to date.

5.2.6 Turkey

Turkey's Constitution stipulates that a judge's membership on the Turkish Constitutional Court will terminate automatically if he or she is convicted of an offence requiring dismissal. Judges may also be removed by an absolute majority vote of the members of the TCC if it is definitely established that he or she is unable to perform his or her duties on account of ill health (Article 147).

6 Analysis of the June 2013 draft Constitution of Tunisia

Several countries in the MENA region are in the process of drafting new constitutions, or substantially reforming existing constitutions. The process in Tunisia is among the most successful so far, and the Constituent Assembly's discussions regarding judicial design are some of the most advanced debates taking place in the region. The draft Constitution issued by Tunisia's Constituent Assembly in June 2013 incorporates elements of the multi-constituency model as well as the legislative supermajority model into its Constitutional Court appointments process.

6.1 Appointments procedure and qualifications

Draft Constitution of the Republic of Tunisia (June 2013)

Article 115: The Constitutional Court is composed of 12 members having no less than 15 years of high expertise, two-thirds of whom must be legal specialists.

The President of the Republic, the Speaker of the Chamber of Deputies, the Prime Minister, and the Supreme Judicial Council shall each nominate six candidates, two-thirds of whom must be legal specialists.

Adopting one-half from each nominating party, the Chamber of Deputies shall elect twelve members by a two-thirds majority. The elected members' mandate shall be for one-term period lasting for nine years.

In the event that the required majority is not reached, the remaining candidates shall, with the same majority required, stand for election again. In the event of failure to reach the required majority, other candidates shall be nominated and the election process shall be repeated following the same method.

One-third of the members of the Constitutional Court shall be renewed every threeyear period. Any vacancies in the hierarchy of the Court shall be filled by virtue of the means adopted during appointment.

The members of the Court shall, from amongst the members, elect a President and a Vice President of the Court.

The June 2013 draft Constitution prescribes a two-step process for Constitutional Court appointments. In the first step, the President, the Speaker of the Chamber of Deputies, the Prime Minister and the Supreme Judicial Council compile separate lists of candidates. This ensures the involvement of the two main public office holders in the executive branch in the proposed semi-presidential system, as well as representatives of the legislative branch and the judiciary in the selection of judges, which promotes political investment in the Court by a wide range of political actors and other constituencies (e.g. the judiciary). In addition, the qualifications that two thirds of each list (four of the six judges on each list) must be legal specialists and that all candidates

have at least 15 years of 'high expertise' will reduce the risk of nominations of candidates based on political loyalties rather than legal expertise.

The second step of the appointments process charges the Chamber of Deputies with electing the Court's judges from the four lists of candidates. The Chamber must elect three judges from each list of six candidates, which guarantees that each of the political actors empowered to propose candidates will play a role in shaping the Court. Furthermore, judges must be elected by a two-thirds supermajority of the Chamber of Deputies, which encourages the different political parties represented in the Chamber to work together to reach compromises on candidates. In particular, it makes it very unlikely that a governing coalition would be able to choose its preferred candidates over the objections of opposition parties.

Overall, the appointments model set out in the June 2013 draft Constitution ensures the involvement of a wide range of political actors in the judicial selection process, thus promoting a strong sense of political investment in the Constitutional Court. It should be noted that requiring a legislative supermajority to elect a candidate can be difficult, and could potentially result in lengthy delays in appointing judges, as shown in the discussions of the German FCC (Section 4.1) and the Italian Constitutional Court (Section 4.4.2). In the June 2013 draft, even if a candidate does not obtain a two-thirds supermajority vote in the first round of voting, he or she must obtain the same supermajority in the second round of voting in order to be appointed. A better option may be to lower the supermajority required in the second round of voting in order to reduce the risk of legislative deadlock over Court appointments.

6.2 Removal procedure

Draft Constitution of the Republic of Tunisia (June 2013)

Article 104: No judge may be transferred without his consent, no judge may be dismissed, and no judge may be suspended, deposed, or subjected to a disciplinary punishment except in such cases and in accordance with the guarantees provided for by the law and by virtue of a justified decision issued by the Supreme Judicial Council.

Tunisia's June 2013 draft Constitution defers the question of removal of Constitutional Court judges to legislation, apart from indicating that the Supreme Judicial Council will play a role in removal proceedings. This is a worrisome omission. As discussed in Section 3.2, the power to remove a judge can have just as much impact on the Court's independence as the appointment power. The Constitution should clearly state the reasons for which a judge may be removed (e.g. misconduct or incapacity) and the procedure for removing a judge. Moreover, the powers of appointment and removal should not be accorded to the same actor(s). For example, as shown in Section 5.2,

some countries only permit the removal of a constitutional court judge if the court itself votes in favour of removal. Some countries also require that two different political bodies (e.g. the judicial council and the legislature) approve the removal of a judge.

7 Conclusion

Many countries in the MENA region are currently considering a range of important questions regarding the formation of a constitutional court and the appointment mechanism for its judges. The Arab Spring sparked a regional debate over constitutional reforms, providing a unique opportunity to create a strong judicial institution that can help promote the rule of law and hold all political actors accountable to the constitution. At present, there are two regional trends emerging. Countries such as Tunisia have proposed a procedure for appointing constitutional court judges that will involve many different political actors, thus fostering a broad sense of political investment in the court and helping to protect its independence. This sense of political investment will provide an incentive for all political actors to continue supporting the court even when they are on the losing side of its decisions.

In contrast, Jordan, Morocco and Syria have all granted the executive branch an enormous amount of power over constitutional court appointments. If court judges fear that angering the executive may cost them their positions, their decisions may be influenced more by the need to please the executive than by the law's requirements. Without establishing procedures and rules that will allow a constitutional court to withstand political pressure, the court will serve as mere window dressing for rulers who wish to give the appearance of respect for the rule of law without creating real checks on their power.

The appointments process should be designed to strike an appropriate balance between (1) the need to protect the constitutional court's independence and insulate it from political interference and (2) its need to be responsive to the democratic society in which it operates. Establishing the professional qualifications that judges must hold in order to be appointed to the court, and providing safeguards against the removal of judges for political reasons, are further measures that should be taken to protect the court's independence. With these principles in mind, we offer the following general recommendations for countries in the MENA region:

- Encourage a broad degree of political investment in the constitutional court by involving a wide range of political actors in the appointments process. The judicial council and multi-constituency models both provide for the inclusion of a large number of political interests, including political parties, civil society organizations, and judicial and legal organizations, as well as academics, in the appointments process.
- Do not grant any one political actor the power to appoint a majority of the constitutional court's members. This will protect the court from undue pressure from politicians, and will help avoid the perception that it is biased toward any particular political actor or party. For this reason, the judiciary-executive model is

not generally recommended for countries in the MENA region, because it gives the executive too much power over court appointments and excludes many other political actors from the process.

- Exercise caution when involving the legislature in constitutional court appointments. Many of the countries studied in this report, including Germany, South Africa, Italy and Turkey, give the legislature a role in selecting some of the court's members. Caution is required in deciding exactly how large a role to give the legislature, and in setting the rules for how the legislature will make its appointments. Countries with a history of weak political parties, or if there is likely to be a great deal of instability in terms of which parties are represented in the legislature, should not adopt an appointments procedure that requires a supermajority vote of the legislature to make court appointments, such as the legislative supermajority model. This model requires political parties to cooperate on constitutional court appointments over time based on relationships of trust and reciprocity. It is difficult for political parties to develop such practices if the party system is weak. In such situations, parties in the legislature will likely find it hard to reach a broad consensus on a given nominee because of the lack of trust, which can result in significant delays in filling vacancies. However, there are other ways to involve the legislature in court appointments; for example, South Africa includes members of the legislature on its Judicial Service Commission.
- Require candidates for the constitutional court to meet a high professional standard. Establishing specific qualifications for judges regarding their level of education and professional achievement has two benefits. First, it ensures that judges appointed to the court will have the knowledge and skills necessary to adjudicate complex constitutional issues. Second, it encourages the court's independence by limiting the pool from which political actors may nominate judges. A political party or president will find it harder to pack the court with loyal supporters if nominees must meet specific requirements to be considered eligible.
- Clearly state the grounds upon which a judge may be removed, and the procedure for removal, in the constitution. Enumerating the reasons why a judge may be removed, such as incapacity or a criminal conviction, helps protect judges from attack by political actors by ensuring that they cannot be removed for purely political reasons. Enshrining the removal procedure in the constitution offers further protection by making it difficult to alter. Furthermore, the power to remove a judge should not be given to the same political actor that appoints the judges, in order to reduce the susceptibility of judges to political pressure.

References

- African National Congress, 'Statement of the National Executive Committee on the Occasion of the 93rd Anniversary of the ANC', 8 January 2005, available at http://www.anc.org.za/show.php?id=55>
- Al-Ali, Zaid, The Struggle for Iraq's Future: How Corruption, Incompetence and Sectarianism Have Undermined Democracy (New Haven, Conn.: Yale University Press, 2014)
- Albrecht, Holger, Egypt's 2012 Constitution: Devil in the Details, Not in Religion (Washington, D.C.: United States Institute of Peace, 2013)
- Al-Saedi, Raid Juhi, 'Regime Change and the Restoration of the Rule of Law in Iraq', International Law Studies, 86/3 (2010), pp. 3-18
- Andrews, Penelope, 'The South African Judicial Appointments Process', Osgoode Hall Law Journal, 44/3 (2006), pp. 565–72
- Andrews, Penelope, 'The Judiciary in South Africa: Independence or Illusion?' in Adam Dodek and Lorne Sossin (eds), *Judicial Independence in Context* (Toronto: Irwin Law, 2010)
- Arato, Andrew, 'The Constitutional Reform Proposal of the Turkish Government: The Return of Majority Imposition', *Constellations*, 17/2 (2010), pp. 345–50
- Arato, Andrew and Goldfarb, Jeffrey, 'Politically Weighted Courts in Turkey', *Deliberately Considered*, 24 September 2010, available at http://www.deliberatelyconsidered.com/2010/09/politically-weighted-courts-inturkey-bad-news-for-constituents/>
- Associated Press, 'Egyptian Judges Back Out of Reform Conference with Morsi', 15 May 2013, available at http://www.cbsnews.com/8301-202_162-57584751/egyptian-judges-back-out-of-reform-conference-with-morsi/
- Baldassarre, Antonio, 'Structure and Organization of the Constitutional Court of Italy', St. Louis University Law Journal, 40/3 (1996), pp. 649–58
- Bâli, Aslı, 'Courts and Constitutional Transition: Lessons from the Turkish Case', International Journal of Constitutional Law, 11/3 (2013), pp. 666–701
- Bâli, Aslı, 'The Perils of Judicial Independence: Constitutional Transition and the Turkish Example', *Virginia Journal of International Law*, 52/2 (2012), pp. 235–320
- Barakat, Luna, 'Jordan: Constitutional Law Court Newly Established in Jordan', U.S. Library of Congress, available at

<http://www.loc.gov/lawweb/servlet/lloc_news?disp3_l205403416_text>

- BBC News, 'Egypt's Islamist Parties Win Elections to Parliament', 21 January 2012, available at http://www.bbc.co.uk/news/world-middle-east-16665748>
- BBC News, 'Q & A: Egypt's Supreme Court Rulings', 17 June 2012, available at http://www.bbc.co.uk/news/world-middle-east-18463887>

- Benda, Ernst and Klein, Eckart, Verfassungsprozessrecht. Ein Lehr- und Handbuch [Constitutional procedure. A textbook and handbook] (Heidelberg: C.F. Müller Verlag, 2001)
- Berning, Joey and Montesh, Moses, 'Countering Corruption in South Africa: The Rise and Fall of the Scorpions and Hawks', SA Crime Quarterly, 39 (March 2012), pp. 3–10, available at <http://www.issafrica.org/uploads/Berning_Montesh.pdf>
- Billing, Werner, Das Problem der Richterwahl zum Bundesverfassungsgericht [The problem of judicial elections for the Federal Constitutional Court] (Berlin: Duncker & Humblot, 1969)
- Boyle, Kevin and Sherif, Adel Omar (eds), *Human Rights and Democracy: The Role of the Supreme Constitutional Court of Egypt* (London: Kluwer Law International, 1996)
- Brown, Nathan, *The Rule of Law in the Arab World* (Cambridge: Cambridge University Press, 1997)
- Brown, Nathan, *Constitutions in a Nonconstitutional World* (Albany: State University of New York Press, 2002)
- Brown, Nathan, 'Why Did the Egyptian Regime Appoint a New Chief Justice?', *Foreign Policy*, 21 July 2009, available at <http://lynch.foreignpolicy.com/posts/2009/07/21/why_did_the_egyptian_regime_ appoint_a_new_chief_justice>
- Brown, Nathan, 'Cairo's Judicial Coup', *Foreign Policy*, 14 June 2012, available at http://mideast.foreignpolicy.com/posts/2012/06/14/cairo_s_judicial_coup
- Brown, Nathan, *Egypt's Judges in a Revolutionary Age* (Washington, D.C.: Carnegie Endowment for International Peace, 2012), available at http://carnegieendowment.org/files/egypt_judiciary.pdf>
- Brown, Nathan, 'Egypt's Judiciary between a Tea Ceremony and the WWE', Foreign Policy, 14 May 2013, available at http://mideast.foreignpolicy.com/posts/2013/05/14/egypts_judiciary_between_a_t ea_ceremony_and_the_wwe>
- Brown, Nathan, 'Quick Reactions to Egypt's New Draft Constitution', *I-CONnect Blog*, 22 August 2013, available at http://www.iconnectblog.com/2013/08/nathanbrown-quick-reactions-to-egypts-new-draft-constitution/>
- Brown, Nathan and Lombardi, Clark, 'The Supreme Constitutional Court of Egypt on Islamic Law, Veiling and Civil Rights: An Annotated Translation of Supreme Constitutional Court of Egypt Case No. 8 of Judicial Year 17 (May 18, 1996)', *American University International Law Review*, 21/3 (2006), pp. 437–60
- Coniglio, Robert, 'Methods of Judicial Decision-Making and the Rule of Law: The Case of Apartheid South Africa', *Boston University International Law Journal*, 30 (2012), pp. 497–527
- Corder, Hugh, 'Judicial Authority in a Changing South Africa', in Guy Canivet, Mads Andenas and Duncan Fairgrieve (eds), *Independence, Accountability and the Judiciary* (London: British Institute of International and Comparative Law, 2006)

- Corder, Hugh, 'Appointment, Discipline and Removal of Judges in South Africa', in H.P. Lee (ed.), *Judiciaries in Comparative Perspective* (Cambridge, Cambridge University Press, 2011)
- Currie, Iain, 'The Judiciary and Freedom of Speech in South Africa', in H. P. Lee (ed.), Judiciaries in Comparative Perspective (Cambridge: Cambridge University Press, 2011)
- De Vos, Pierre, 'On the Disbanding of the Scorpions', *Constitutionally Speaking*, 24 January 2008, available at http://constitutionallyspeaking.co.za/on-the-disbanding-of-the-scorpions/
- De Vos, Pierre, 'ANC Trying to 'Load' the JSC?' *Constitutionally Speaking*, 10 July 2009, available at http://constitutionallyspeaking.co.za/anc-trying-to-load-the-jsc/
- Diamond, Larry, Squandered Victory: The American Occupation and the Bungled Effort to Bring Democracy to Iraq (New York: Times Books, 2005)
- *Economist, The*, 'Injudicious', 22 October 2009, available at <hr/><hr/>http://www.economist.com/node/14710716>
- *Economist, The*, 'Morocco's Referendum: A Very Small Step', 7 July 2011, available at http://www.economist.com/node/18929381>
- *Economist, The*, 'Turkey's Constitutional Referendum: Erdogan Pulls it Off', 13 September 2010, available at <http://www.economist.com/blogs/newsbook/2010/09/turkeys_constitutional_refer endum>
- El Deeb, Sarah, 'Former Judge Challenges Egypt's Constitution', Associated Press, 8 January 2013, available at http://news.yahoo.com/former-judge-challenges-egypts-constitution-130519999.html
- El-Din, Gamal Essam, 'Egypt's Shura Council to Continue Exercising Legislative Powers', *Ahram Online*, 3 June 2013
- European Commission, 'Turkey 2010 Progress Report', 2010, available at http://ec.europa.eu/enlargement/pdf/key_documents/2010/package/tr_rapport_2010_en.pdf>
- Feldman, Noah and Martinez, Roman, 'Constitutional Politics and Text in the New Iraq: An Experiment in Islamic Democracy', *Fordham Law Review*, 75 (2006), pp. 883–920
- Ferreira, Emsie, 'South Africa: Minister Denies Manipulation of Judicial Service Commission', *South African Press Association*, 26 April 2012, available at http://allafrica.com/stories/201204270309.html
- Garoupa, Nuno and Ginsburg, Tom, 'Guarding the Guardians: Judicial Councils and Judicial Independence', American Journal of Comparative Law, 57 (2009), pp. 201– 32

- Geck, Wilhelm Karl, *Wahl und Amtsrecht der Bundesverfassungsrichter* [Election and laws regulating the office of Federal Constitutional Court judges] (Baden-Baden: Nomos Verlagsgesellschaft, 1986)
- Gibson, James, 'The Evolving Legitimacy of the South African Constitutional Court', in Francois Du Bois and Antje Du Bois-Pedain (eds), *Justice and Reconciliation in Post-Apartheid South Africa* (Cambridge: Cambridge University Press, 2008)
- Gibson, James and Caldeira, Gregory, 'Defenders of Democracy? Legitimacy, Popular Acceptance, and the South African Constitutional Court', *Journal of Politics*, 65/1 (2003), pp. 1–30
- Ginsburg, Tom, Judicial Review in New Democracies: Constitutional Courts in Asian Cases (Cambridge: Cambridge University Press, 2003)
- Gloppen, Siri, South Africa: The Battle over the Constitution (Aldershot: Ashgate Dartmouth, 1997)
- Greene, Richard Allen, 'Ahmed Shafik: Egypt's "Counter-revolutionary Candidate", CNN, 16 June 2012, available at

<http://www.cnn.com/2012/06/15/world/meast/egypt-election-shafik>

- Grote, Rainer, 'Models of Institutional Control: The Experience of Islamic Countries', in Rainer Grote and Tilmann Röder (eds), *Constitutionalism in Islamic Countries: Between Upheaval and Continuity* (Oxford: Oxford University Press, 2012)
- Hamad, Mahmoud, 'The Politics of Judicial Selection in Egypt', in Kate Malleson and Peter Russell (eds), *Appointing Judges in an Age of Judicial Power* (Toronto: University of Toronto Press, 2006)
- Hamoudi, Haider, 'Decision 88: Balance of Power under the Iraq Constitution', JURIST, 18 March 2011, available at <http://jurist.org/forum/2011/03/decision-88-checks-and-balances-in-iraq.php>
- Hamoudi, Haider, 'The Will of the (Iraqi) People', *Utah Law Review*, 1 (2011), pp. 45–61
- Hamoudi, Haider, Negotiating in Civil Conflict: Constitutional Construction and Imperfect Bargaining in Iraq (Chicago: University of Chicago Press, 2013)
- Harding, Andrew, Leyland, Peter and Groppi, Tania, 'Constitutional Courts: Forms, Functions and Practice in Comparative Perspective', in Andrew Harding and Peter Leyland (eds), *Constitutional Courts: A Comparative Study* (London: Wildy, Simmonds & Hill, 2009)
- Hedling, Nora, 'The Design of the Judicial Branch', in Markus Böckenförde, Nora Hedling, Winluck Wahiu (eds), A Practical Guide to Constitution Building (Stockholm: International IDEA, 2011), available at <http://www.idea.int/publications/pgcb/>
- Hirschl, Ran, Towards Juristocracy: The Origins and Consequences of the New Constitutionalism (Cambridge: Harvard University Press, 2004)
- Hönnige, Christoph, Verfassungsgericht, Regierung und Opposition. Die vergleichende Analyse eines Spannungsdreiecks [Constitutional Court, government and opposition.

The comparative analysis of a tense triangle] (Wiesbaden: VS Verlag für Sozialwissenschaften, 2007)

- Hooper, John, 'Italian Court Waters down Berlusconi Immunity Law', *The Guardian*, 13 January 2011, available at http://www.guardian.co.uk/world/2011/jan/13/italian-court-berlusconi-immunity-law
- Hurriyet Daily News, 'Court Clears Way for Gül to Re-run Top Post', 16 June 2012, available at ">http://www.hurriyetdailynews.com/court-clears-way-for-gul-to-re-run-top-post.aspx?pageID=238&nID=23315&NewsCatID=338>">http://www.hurriyetdailynews.com/court-clears-way-for-gul-to-re-run-top-post.aspx?pageID=238&nID=23315&NewsCatID=338>">http://www.hurriyetdailynews.com/court-clears-way-for-gul-to-re-run-top-post.aspx?pageID=238&nID=23315&NewsCatID=338>">http://www.hurriyetdailynews.com/court-clears-way-for-gul-to-re-run-top-post.aspx?pageID=238&nID=23315&NewsCatID=338>">http://www.hurriyetdailynews.com/court-clears-way-for-gul-to-re-run-top-post.aspx?pageID=238&nID=23315&NewsCatID=338>">http://www.hurriyetdailynews.com/court-clears-way-for-gul-to-re-run-top-post.aspx?pageID=238&nID=23315&NewsCatID=338>">http://www.hurriyetdailynews.com/court-clears-way-for-gul-to-re-run-top-post.aspx?pageID=238&nID=23315&NewsCatID=338>">http://www.hurriyetdailynews.com/court-clears-way-for-gul-to-re-run-top-post.aspx?pageID=238&nID=23315&NewsCatID=338>">http://www.hurriyetdailynews.com/court-clears-way-for-gul-to-re-run-top-post.aspx?pageID=238&nID=23315&NewsCatID=338>">http://www.hurriyetdailynews.com/court-clears-way-for-gul-to-re-run-top-post.aspx?pageID=238&nID=23315&NewsCatID=338>">http://www.hurriyetdailynews.com/court-clears-way-for-gul-to-re-run-top-post.aspx?pageID=238&nID=
- Hussain, Majid Salman, A Critical Study of Constitutional and Judicial Development in Kuwait, University of Newcastle upon Tyne Ph.D thesis (June 2010), available at https://theses.ncl.ac.uk/dspace/bitstream/10443/1164/1/Hussain11.pdf>
- Jordan Times, 'Profiles of Constitutional Court Members,' available at http://jordantimes.com/uploads/downloads/76228d5eeeb39b0ed4d2040e0984ad1 f8c4f3960.pdf>
- International Crisis Group, 'Trial by Error: Justice in Post-Qadhafi Libya', 17 April 2013, available at http://www.crisisgroup.org/en/regions/middle-east-north-africa/north-africa/libya/140-trial-by-error-justice-in-post-qadhafi-libya.aspx
- Issacharoff, Samuel, 'Constitutional Courts and Democratic Hedging', *Georgetown Law Journal*, 99 (2000), pp. 961–1012
- Kirkpatrick, David, 'Blow to Transition as Court Dissolves Egypt's Parliament', New York Times, 14 June 2012, available at <http://www.nytimes.com/2012/06/15/world/middleeast/new-politicalshowdown-in-egypt-as-court-invalidates-parliament.html?pagewanted=all>
- Klug, Heinz, 'South Africa's Constitutional Court: Enabling Democracy and Promoting Law in the Transition from Apartheid', in Andrew Harding and Peter Leyland (eds), *Constitutional Courts: A Comparative Study* (London: Wildy, Simmonds & Hill, 2009)
- Klug, Heinz, The Constitution of South Africa: A Contextual Analysis (Oxford: Hart Publishing, 2010)
- Kogacioglu, Dicle, 'Progress, Unity and Democracy: Dissolving Political Parties in Turkey', *Law & Society Review*, 38 (2004), pp. 433-62
- Kommers, Donald, Judicial Politics in West Germany: A Study of the Federal Constitutional Court (London: Sage Publications, 1976)
- Kommers, Donald, *The Constitutional Jurisprudence of the Federal Republic of Germany* (2nd ed.) (Durham, N.C.: Duke University Press, 1997)
- Kommers, Donald, 'Autonomy versus Accountability: The German Judiciary', in Peter Russell and David O'Brien (eds), *Judicial Independence in the Age of Democracy* (Charlottesville: University Press of Virginia, 2001)
- Kranenpohl, Uwe, Hinter dem Schleier des Beratungsgeheimnisses: der Willensbildungs- und Entscheidungsprozess des Bundesverfassungsgerichts [Behind the veil of secrecy of

deliberations of the court: the opinion-shaping and decision-making process of the Federal Constitutional Court] (Wiesbaden: VS Verlag für Sozialwissenschaften, 2010)

- Kruger, Franz, 'You Said it, Mantashe', *Mail & Guardian*, 14 October 2008, available at <<u>http://www.mg.co.za/article/2008-10-14-you-said-it-mantashe></u>
- Landfried, Christine, 'Die Wahl der Bundesverfassungsrichter und ihre Folgen für die Legitimität der Verfassungsgerichtsbarkeit' [The election of Federal Constitutional Court judges and its effect on the legitimacy of the constitutional jurisdiction], in Robert van Ooyen and Martin Möllers (eds), Das Bundesverfassungsgericht im politischen System [The Federal Constitutional Court in the political system] (Wiesbaden: VS Verlag für Sozialwissenschaften, 2006)
- Lembcke, Oliver, 'Das Bundesverfassungsgericht und die Regierung Adenauer vom Streit um den Status zur Anerkennung der Autorität', [The Federal Constitutional Court and the Adenauer government – from the debate about the status to the acknowledgement of its authority], in Robert van Ooyen and Martin Möllers (eds), *Das Bundesverfassungsgericht im politischen System* [The Federal Constitutional Court in the political system] (Wiesbaden: VS Verlag für Sozialwissenschaften, 2006)
- Macdonald, Roderick, and Kong, Hoi, 'Judicial Independence as a Constitutional Virtue', in Michel Rosenfeld and András Sajó (eds), *The Oxford Handbook of Comparative Constitutional Law* (Oxford: Oxford University Press, 2012)
- Madhuku, Lovemore, 'Constitutional Protection of the Independence of the Judiciary: A Survey of the Position in Southern Africa', *Journal of African Law*, 46/2 (2002), pp. 232–45
- Mahmoud, Medhat, 'Judicial System in Iraq: A Review of the Legislation Regulating Judicial Affairs in Iraq' (2004), available at
- <http://gjpi.org/wp-content/uploads/2009/01/mahmood_2004.pdf>
- Malleson, Kate, 'Assessing the Performance of the Judicial Service Commission', *South African Law Journal*, 116/1 (1999), pp. 36–49
- Mandel, Michael, 'Legal Politics Italian Style', in C. Neal Tate and Torbjorn Vallinder (eds), *The Global Expansion of Judicial Power* (New York: NYU Press, 1997)

Max Planck Institute, 'Bahrain', January 2013, available at <http://www.mpil.de/en/pub/service/globaler_wissenstransfer/constitutional_refor m_in_arab/bahrain.cfm>

McGreal, Chris, 'Mandela Attacks Mbeki's AIDS policy', *The Guardian*, 17 February 2002, available at

<http://www.theguardian.com/world/2002/feb/18/aids.nelsonmandela>

McKaiser, Eusebius, 'Democracy and its Malcontents', *New York Times*, 8 March 2012, available at <http://latitude.blogs.nytimes.com/2012/03/08/south-africas-governing-party-resents-the-constitutional-courts-fierce-independence/>

- Mofokeng, Moffet, Matlala, George, and Hawker, Dianne, 'Moseneke in New Clash with Zuma', *The Sunday Independent*, 14 May 2012, available at <http://www.iol.co.za/sundayindependent/moseneke-in-new-clash-with-zuma-1.1296282#.UYCRvivSO5w>
- Molloy, Connor, 'Why the Reduction in SCC Judges?' *Daily News Egypt*, 24 December 2012, available at http://www.dailynewsegypt.com/2012/12/24/why-the-reduction-in-scc-justices/
- Moustafa, Tamir, 'Law versus the State: The Judicialization of Politics in Egypt', Law & Social Inquiry, 28 (2003), pp. 883–930
- Moustafa, Tamir, The Struggle for Constitutional Power: Law, Politics, and Economic Development in Egypt (Cambridge: Cambridge University Press, 2007)
- Msomi, S'Thembiso, 'Battle of the Constitution: Changes on ANC Wish List', *Times Live*, 4 March 2012, available at http://www.timeslive.co.za/politics/2012/03/04/battle-of-the-constitution-changes-on-anc-wish-list
- Nicolauß, Karlheiz, 'Der Parlamentarische Rat und das Bundesverfassungsgericht' [The Parliamentary Council and the Federal Constitutional Court], in Robert van Ooyen and Martin Möllers (eds), *Das Bundesverfassungsgericht im Politischen System* [The Federal Constitutional Court in the political system] (Wiesbaden: VS Verlag für Sozialwissenschaften, 2006)
- Ottaway, Marina, *The New Moroccan Constitution: Real Change or More of the Same?* (Washington, D.C.: Carnegie Endowment for International Peace, 2011), available at http://carnegieendowment.org/2011/06/20/new-moroccan-constitution-real-change-or-more-of-same/51
- Özpek, Burak Bilgehan, 'Constitution-making in Turkey after the 2011 Elections', *Turkish Studies*, 13 (2012), pp. 153–67
- Parliament of the Republic of South Africa, 'State of Parties in the National Assembly', available at http://www.parliament.gov.za/live/content.php?Item_ID=324>
- Parliament of the Republic of South Africa, 'State of the Parties in the National Council of Provinces', available at

<http://www.parliament.gov.za/live/content.php?Item_ID=403>

- Pasquino, Pasquale, and Ferejohn, John, 'Constitutional Adjudication: Lessons from Europe', *Texas Law Review*, 82/7 (2004), pp. 1671–704
- Pasquino, Pasquale, 'The Debates of the Italian Constituent Assembly', in Pasquale Pasquino and Francesca Billi (eds), *The Political Origins of Constitutional Courts* (Rome: Fondazione Adriano Olivetti, 2009), available at <http://www.fondazioneadrianolivetti.it/_images/pubblicazioni/collana/031411115 24510.pdf>
- Pellegrina, Lucia, and Garoupa, Nuno, 'Choosing Between the Government and the Regions: An Empirical Analysis of the Italian Constitutional Court Decisions', *Illinois Law, Behavior and Social Science Research Papers Series*, 52 (2013), available at

<http://www.astrid-online.it/--l-attuaz/Studi--ric/Dalla-

Pellegrina_Garoupa_EuropeanJournal-PolResearch_4_2013.pdf>

- Peri, Antonina, 'Judicial Independence v. Judicial Accountability: Judicial Selection Models for Constitutional Courts: A Comparative Analysis', *Comparative Law Review*, 3/1 (2012)
- Pieper, Stefan Ulrich, Verfassungsrichterwahlen. Die Besetzung der Richterbank des Bundesverfassungsgerichts und die Besetzung des Europäischen Gerichtshofs sowie des Europäischen Gerichtshofs für Menschenrechte und des Internationalen Gerichtshofs mit deutschen Kandidaten [Elections of constitutional judges. The appointment of the judges' bench of the Federal Constitutional Court and the appointment of German candidates to the European Court of Human Rights and to the International Court of Justice] (Berlin: Duncker & Humblot, 1998)
- Posner, Michael, and McKay, Fiona, 'The Iraqi Justice System: Challenges in Responding to Iraq's Past Abuses of Human Rights', *The Judges' Journal*, 42 (2003), pp. 14–16
- Quintal, Angela, 'Judges to be Given Chance to Discuss Reforms', *Independent Online*, 31 July 2006, available at http://www.iol.co.za/news/politics/judges-to-be-given-chance-to-discuss-reforms-1.287402#. UboZ5vmzfTo>
- Radwan, Tarek, 'Egypt's Return of the Judiciary,' *Foreign Policy*, 18 October 2012, available at <http://mideast.foreignpolicy.com/posts/2012/10/18/egypt_s_return_of_the_judici

ary>

- Ramphele, Mamphela, and Dawood, Zohra, 'South African Courts at Risk', *Project Syndicate*, 5 June 2012, available at http://www.project-syndicate.org/commentary/south-african-courts-at-risk
- Rawoot, Ilham, 'Mogoeng's Assault on Women's Rights', *Mail & Guardian*, 26 August 2011, available at ">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles>">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles">http://mg.co.za/article/2011-08-26-mogoengs-marital-rape-judgments-raise-hackles"
- Roberts, Justin, 'Iraq's Burgeoning Judiciary: Current Issues and the Best Way Forward', *Humanities and Social Sciences*, 4/1 (2008), pp. 1–8
- Roux, Theunis, The Politics of Principle: The First South African Constitutional Court, 1995–2005 (Cambridge: Cambridge University Press, 2013)
- Royal Hashemite Court of Jordan, 'Jordan Enacts New Political Parties, Constitutional Court Laws', 6 June 2012, available at <http://www.kingabdullah.jo/index.php/en_US/news/view/id/10139/videoDisplay/ 1.html>
- Sadah, Ali Abel, 'Iraqi Judicial Reforms Include Removal of Chief Justice,' *Al Monitor*, 18 February 2013, available at

<http://www.al-monitor.com/pulse/originals/2013/02/judiciary-changes-iraq.html> Sadurski, Wojciech, 'Twenty Years after the Transition: Constitutional Review in

Central and Eastern Europe', Sydney Law School Research Paper No. 09/69, 23

July 2009, available at

<http://papers.ssrn.com/sol3/papers.cfm?abstract_id=1437843>

- Schlaich, Klaus and Korioth, Stefan, Das Bundesverfassungsgericht. Stellung, Verfahren, Entscheidungen. Ein Studienbuch [The Federal Constitutional Court. Position, procedure and decisions. A study book] (Munich: C.H. Beck Verlag, 2012)
- Schoenberger, Christoph, 'The Establishment of Judicial Review in Postwar Germany', in Pasquale Pasquino and Francesca Billi (eds), *The Political Origins of Constitutional Courts* (Rome: Fondazione Adriano Olivetti, 2009), available at http://www.fondazioneadrianolivetti.it/_images/pubblicazioni/collana/03141111524510.pdf>
- Seibert-Fohr, Anja, 'Judicial Independence in Germany', in Anja Seibert-Fohr (ed.), Judicial Independence in Transition (Heidelberg: Springer, 2012)
- Shambayati, Hooten and Kirdis, Esen, 'In Pursuit of "Contemporary Civilization": Judicial Empowerment in Turkey', *Political Research Quarterly*, 62 (2009), pp. 767– 80
- Sly, Liz, 'Iraq's High Court Orders Parliament to Get Back to Work', *Los Angeles Times*, 25 October 2010, available at

http://articles.latimes.com/2010/oct/25/world/la-fg-iraq-court-ruling-20101025> Smith, Janet, 'Chief Justice: How Zuma Got it Wrong', *Saturday Star*, 22 August 2011,

- available at <http://www.iol.co.za/saturday-star/chief-justice-how-zuma-got-itwrong-1.1122867#.UYCNySvSO5w>
- Spitz, Richard and Chaskalson, Matthew, *The Politics of Transition: A Hidden History of South Africa's Negotiated Settlement* (Oxford: Hart Publishing, 2000)
- Squires, Nick, 'Silvio Berlusconi to Reform Italian Courts to Stop "Witch Hunt", *The Telegraph*, 16 October 2009, available at http://www.telegraph.co.uk/news/worldnews/europe/italy/6349011/Silvio-Berlusconi-to-reform-Italian-courts-to-stop-witch-hunt.html
- Stilt, Kristen, "Islam is the Solution": Constitutional Visions of The Egyptian Muslim Brotherhood', *Texas International Law Journal*, 46 (2010), pp. 73–108
- Stone, Setumo, 'ANC Still has Review of Judiciary's Powers in its Sights', Business Day Live, 14 January 2013, available at http://www.bdlive.co.za/national/law/2013/01/14/news-analysis-anc-still-has-review-of-judiciarys-powers-in-its-sights
- Stone Sweet, Alec, 'Constitutional Courts', in Michel Rosenfeld and András Sajó (eds), The Oxford Handbook of Comparative Constitutional Law (Oxford: Oxford University Press, 2012)
- Süddeutsche Zeitung, 'CDU Verhindert Wechsel am Bundesverfassungsgericht' [CDU hampers replacement at the Federal Constitutional Court], 17 May 2010, available at <http://www.sueddeutsche.de/politik/bundesverfassungsgericht-cdu-verhindert-wechsel-am-verfassungsgericht-1.327900-2>

- Trumbull, Charles, and Martin, Julie, 'Elections and Government Formation in Iraq: An Analysis of the Judiciary's Role', *Vanderbilt Journal of Transnational Law*, 44 (2011), pp. 331–88
- Vanberg, Georg, *The Politics of Constitutional Review in Germany* (Cambridge: Cambridge University Press, 2005)
- Venice Commission (Council of Europe), 'List of Constitutional Courts', available at http://www.venice.coe.int/webforms/courts/>
- Visser, Reidar, 'Iraqiyya Challenges the Jurisdiction of the Federal Supreme Court', Iraq and Gulf Analysis, 31 March 2010, available at http://gulfanalysis.wordpress.com/2010/03/31/iraqiyya-challenges-the-jurisdiction-of-the-federal-supreme-court/>
- Visser, Reidar, 'The Law on the Federal Supreme Court: The Balance of Islamic and Career Judges', *Iraq and Gulf Analysis*, 6 February 2011, available at <http://gulfanalysis.wordpress.com/2011/02/06/the-law-on-the-federal-supremecourt-the-balance-of-islamic-and-career-judges/>
- Voigt, Rüdiger, 'Das Bundesverfassungsgericht in rechtspolitologischer Sicht' [The Federal Constitutional Court from a political-legal point of view], in Robert van Ooyen and Martin Möllers (eds), Das Bundesverfassungsgericht im Politischen System [The Federal Constitutional Court in the political system] (Wiesbaden: VS Verlag für Sozialwissenschaften, 2006)
- Volcansek, Mary, 'Political Power and Judicial Review in Italy', Comparative Political Studies, 26/4 (1994), pp. 492–509
- Volcansek, Mary, Constitutional Politics in Italy (New York: St. Martin's Press, 2000)
- Volcansek, Mary, 'Appointing Judges the European Way', Fordham Urban Law Journal, 34 (2007), pp. 363–85
- Volcansek, Mary, 'Bargaining Constitutional Design in Italy: Judicial Review as Political Insurance', *West European Politics*, 33 (2010), pp. 280–96
- Wesson, Murray, and du Plessis, Max, 'The Transformation of the Judiciary: Fifteen Year Policy Review', Presidency of the Republic of South Africa, available at <http://www.thepresidency.gov.za/docs/reports/15year_review/jcps/transformation _judiciary.pdf>
- Willoweit, Dietmar, Deutsche Verfassungsgeschichte. Vom Frankenreich bis zur Wiedervereinigung Deutschlands [German constitutional history. From the Kingdom of the Franks to the German Reunification] (Munich: C.H. Beck Verlag, 2009)
- Wing, Joel, 'Iraq's Federal Court Returns Chief Justice to Top Position,' Musings on Iraq, 24 September 2013, available at http://musingsoniraq.blogspot.com/2013/09/iraqs-federal-court-returnschief.html>

Zebley, Julia, 'Italy PM Berlusconi Pledges Judicial Reform while Awaiting Trial', *JURIST*, 20 February 2011, available at <http://jurist.org/paperchase/2011/02/italy-pm-berlusconi-pledges-judicialreform-while-awaiting-trial.php>

Zille, Helen, 'Zille Disappointed over Zuma's Final JSC Choices', *Politics Web*, 19 July 2009, available at

<http://www.politicsweb.co.za/politicsweb/view/politicsweb/en/page71646?oid=13 6773&sn=Detail&pid=71646>

Appendix: selected constitutional and legal provisions on constitutional courts

Egypt

Constitution of the Arab Republic of Egypt (11 Sept. 1971, as amended 22 May 1980, 25 May 2005, 26 March 2007)¹⁶⁵

Article 174:

The Supreme Constitutional Court shall be an independent, self-standing judiciary body, in the Arab Republic of Egypt, seated in Cairo.

Article 175:

The Supreme Constitutional Court shall exclusively undertake the judicial control of the constitutionality of the laws and regulations, and shall undertake in the manner prescribed by the law the interpretation of legislative texts. The law shall determine the other competencies of the court, and regulate the procedure to be followed before it.

Article 176:

The law shall regulate the manner of the formation of the Supreme Constitutional Court, and define requirements to be satisfied by its members, rights and immunities.

Article 177:

Members of the Supreme Constitutional Court shall not be removed from office. The Court shall call to account its members, in the manner prescribed by the law.

Constitution of the Arab Republic of Egypt (26 Dec. 2012 (suspended))¹⁶⁶

Section 4: The Supreme Constitutional Court

Article 175: Mandate and procedures

The Supreme Constitutional Court is an independent judicial body. It is based in Cairo. It is exclusively competent to decide on the constitutionality of laws and regulations. The law defines the Court's other competencies and regulates the procedures that are to be followed before the Court.

Article 176: Composition

The Supreme Constitutional Court is made up of a president and ten members. The law determines the judicial or other bodies and associations that nominate them, the manner in which they are to be appointed, and the requirements to be satisfied by them. Appointments take place by a decree from the President of the Republic.

Article 177: Constitutionality of electoral laws

The President of the Republic or the Speaker of the Council of Representatives present draft laws governing presidential, legislative or local elections before the Supreme Constitutional Court, to determine their compliance with the Constitution prior to dissemination. The Court reaches a decision in this regard within 45 days from the date the matter is presented before it; otherwise, the proposed law is considered approved.

If the Court deems one or more parts of the text non-compliant with the provisions of the Constitution, its decision is implemented.

The laws referred to in the first paragraph are not subject to the subsequent control stipulated in Article 175 of the Constitution.

Law No. 48 of 1979 (Law Governing the Operations of the Supreme Constitutional Court)¹⁶⁷

Article 3:

The Court shall be formed of the Chief Justice and a sufficient number of members. Judgements and decisions of the court, shall be entered by a quorum of seven of its members.

The Chief Justice or the next senior member, shall preside over the sessions of the court. Where the office of the Chief Justice is vacant, or in the case of his absence or impairment, all competencies attributed thereto shall be carried out in accordance with descending seniority of the members of the Court.

Article 4:

Nominees for membership of the court must meet all the requirements for the general judicial service specified in the law on the judicial power, and must be not less than forty-five calendar years of age.

They may be chosen from among the following groups:

- (a) Members of the Supreme Court.
- (b) Current or former members of the judicial bodies holding the rank of a counsellor or its equivalent for at least five consecutive years.
- (c) Current or former law professors who have held the position of a professor at an Egyptian university for at least eight consecutive years.
- (d) Attorneys-at-law who have practiced before the court of cassation, or the high administrative court for at least ten consecutive years.

Article 5:

The President of the Republic appoints the Chief Justice of the Court by a presidential decree. Members of the court are also appointed by a presidential decree after consulting with the Supreme Council of the Judicial Bodies; from among two candidates, one is chosen by the general assembly of the Court, and the other by the Chief Justice. At least two thirds of the appointees to the bench must be chosen from the other judicial bodies. The presidential decree that appoints a member shall indicate his position and seniority.

Article 26:

The Supreme Constitutional Court is empowered to provide the definitive interpretation of laws, enacted by the legislature, and presidential decrees with the force of law issued in accordance with the constitution if, during the course of their application, they arise divergent points of view, and they have an importance that necessitates uniform interpretation.

Germany

Grundgesetz Für die Bundesrepulik Deutschland [Basic Law] (23 May 1949, as last amended July 21, 2010)¹⁶⁸

Article 92: Court organization

The judicial power shall be vested in the judges; it shall be exercised by the Federal Constitutional Court, by the federal courts provided for in this Basic Law, and by the courts of the Länder.

Article 94: Composition of the Federal Constitutional Court

(1) The Federal Constitutional Court shall consist of federal judges and other members. Half the members of the Federal Constitutional Court shall be elected by the Bundestag and half by the Bundesrat. They may not be members of the Bundestag, of the Bundesrat, of the Federal Government, or of any of the corresponding bodies of a Land.

(2) The organisation and procedure of the Federal Constitutional Court shall be regulated by a federal law, which shall specify in which instances its decisions shall have the force of law. The law may require that all other legal remedies be exhausted before a constitutional complaint may be filed, and may provide for a separate proceeding to determine whether the complaint will be accepted for decision.

Federal Constitutional Court Act (Bundesverfassungsgerichtsgesetz in der Fassung der Bekanntmachung vom 11. August 1993) (BGBl. I S. 1473)¹⁶⁹

§ 2

(1) The Federal Constitutional Court shall consist of two panels.

(2) Eight judges shall be elected to each panel.

(3) Three judges of each panel shall be elected from among the judges of the supreme Federal courts of justice. Only judges who have served at least three years with a supreme Federal court of justice should be elected.

§3

(1) The judges must have reached the age of 40, be eligible for election to the Bundestag, and have stated in writing that they are willing to become a member of the Federal Constitutional Court.

(2) They must be qualified to exercise the functions of a judge pursuant to the Judges Act.

(3) They may not be members of the Bundestag, the Bundesrat, the Federal Government, nor of any of the corresponding organs of a Land. On their appointment they shall cease to be members of such organs.

(4) The functions of a judge shall preclude any other professional occupation save that of a lecturer of law at a German institution of higher education. The functions of a Judge of the Federal Constitutional Court shall take precedence over the functions of such lecturer.

§ 4

(1) The term of office of the judges shall be twelve years, not extending beyond retirement age.

(2) Immediate or subsequent re-election of judges shall not be permissible.

(3) Retirement age shall be the end of the month in which a judge reaches the age of 68.

(4) Upon expiration of his term of office a judge shall continue to perform his functions until a successor is appointed.

§ 5

(1) Half of the judges of each panel shall be elected by the Bundestag and the other half by the Bundesrat. Of those to be selected from among the judges of the supreme Federal courts of justice one shall be elected by one of the electoral organs and two by the other, and of the remaining judges three shall be elected by one organ and two by the other.

(2) A judge shall be elected at the earliest three months before the expiration of his predecessor's term of office or, if the Bundestag is dissolved at the time, within one month of the first meeting of the Bundestag.

(3) If a judge relinquishes his office prematurely, his successor shall be elected within one month by the same Federal organ as that which elected his predecessor.

§6

(1) The judges to be elected by the Bundestag shall be elected indirectly.

(2) The Bundestag shall, by proportional representation, elect a twelve-man electoral committee for the Federal Constitutional Court judges. Each parliamentary group may propose candidates for the committee. The number of candidates elected on each list shall be calculated from the total number of votes cast for each list in accordance with the d'Hondt method. The members shall be elected in the sequence in which their names appear on the list. If a member of the electoral committee retires or is unable to perform his functions, he shall be replaced by the next member on the same list.

(3) The eldest member of the electoral committee shall immediately with one week's notice call a meeting of the committee to elect the judges and shall chair the meeting, which shall continue until all of them have been elected.

(4) The members of the electoral committee are obliged to maintain secrecy about the personal circumstances of candidates which become known to them as a result of their activities in the committee as well as about discussions hereon in the committee and the voting.

(5) To be elected, a judge shall require at least eight votes.

§ 7

The judges to be elected by the Bundesrat shall be elected with two thirds of the votes of the Bundesrat.

§105

(1) The Federal Constitutional Court may authorize the Federal President to

1. retire a Judge of the Federal Constitutional Court because of permanent unfitness for service;

2. dismiss a Judge of the Federal Constitutional Court if he has been sentenced without appeal because of a dishonourable act or to over six months' imprisonment or, if he has committed a gross breach of duty, so that his remaining in office is ruled out.

(2) The plenum of the Federal Constitutional Court shall decide on the institution of proceedings pursuant to paragraph 1 above.

(3) The General Procedural Provisions and the provisions of Articles 56 (1) and 55 (1),(2), (4) to (6) above shall apply mutatis mutandis.

(4) Authorization pursuant to paragraph 1 above shall require the consent of two thirds of the members of the Court.

(5) After institution of proceedings pursuant to paragraph 2 above the plenum of the Federal Constitutional Court may temporarily remove the judge from office. The same shall apply if principal proceedings have been instituted against the judge because of a misdemeanour. Temporary removal from office shall require the consent of two thirds of the members of the Court.

(6) Upon dismissal pursuant to paragraph 1 (2) above the judge shall forfeit all claims arising from his office.

Iraq

Law of Administration for the State of Iraq in the Transitional Period (8 Mar. 2004)¹⁷⁰

Article 44:

- (A) A court called the Federal Supreme Court shall be constituted by the law in Iraq.
- (B) The jurisdiction of the Federal Supreme Court shall be as follows:
- (1) Original and exclusive jurisdiction in legal proceedings between the Iraqi Transitional Government and the regional governments, governorate and municipal administrations, and local administrations.
- (2) Original and exclusive jurisdiction, on the basis of a complaint from a claimant or a referral from another court, to review claims that a law, regulation, or directive issued by the federal or regional governments, the governorate or municipal administrations, or local administrations is inconsistent with this Law.
- (3) Ordinary appellate jurisdiction of the Federal Supreme Court shall be defined by federal law.
- (C) Should the Federal Supreme Court rule that a challenged law, regulation, directive, or measure is inconsistent with this Law, it shall be deemed null and void.
- (D) The Federal Supreme Court shall create and publish regulations regarding the procedures required to bring claims and to permit attorneys to practice before it. It shall take its decisions by simple majority, except decisions with regard to the proceedings stipulated in Article 44(B)(1), which must be by a two-thirds majority. Decisions shall be binding. The Court shall have full powers to enforce its decisions, including the power to issue citations for contempt of court and the measures that flow from this.
- (E) The Federal Supreme Court shall consist of nine members. The Higher Juridical Council shall, in consultation with the regional judicial councils, initially nominate no less than eighteen and up to twenty-seven individuals to fill the initial vacancies

in the aforementioned Court. It will follow the same procedure thereafter, nominating three members for each subsequent vacancy that occurs by reason of death, resignation, or removal. The Presidency Council shall appoint the members of this Court and name one of them as its Presiding Judge. In the event an appointment is rejected, the Higher Juridical Council shall nominate a new group of three candidates.

Article 45:

A Higher Juridical Council shall be established and assume the role of the Council of Judges. The Higher Juridical Council shall supervise the federal judiciary and shall administer its budget. This Council shall be composed of the Presiding Judge of the Federal Supreme Court, the presiding judge and deputy presiding judges of the federal Court of Cassation, the presiding judges of the federal Courts of Appeal, and the presiding judge and two deputy presiding judges of each regional court of cassation. The Presiding Judge of the Federal Supreme Court shall preside over the Higher Juridical Council. In his absence, the presiding judge of the federal Court of Cassation shall preside over the Council.

Constitution of the Republic of Iraq (15 Oct. 2005)¹⁷¹

Article 92:

First: The Federal Supreme Court is an independent judicial body, financially and administratively.

Second: The Federal Supreme Court shall be made up of a number of judges, experts in Islamic jurisprudence, and legal scholars, whose number, the method of their selection, and the work of the Court shall be determined by a law enacted by a two-thirds majority of the members of the Council of Representatives.

Article 97:

Judges may not be removed except in cases specified by law. Such law will determine the particular provisions related to them and shall regulate their disciplinary measures.

Italy

Costituzione della Repubblica Italiana (27 Dec. 1947, as amended 4 Dec. 1992, title 7, sec. 1)¹⁷² Article 135:

The Constitutional Court shall be composed of fifteen judges, a third nominated by the President of the Republic, a third by Parliament in joint sitting and a third by the ordinary and administrative supreme Courts.

The judges of the Constitutional Courts shall be chosen from among judges, including those retired, of the ordinary and administrative higher Courts, university professors of law and lawyers with at least twenty years practice.

Judges of the Constitutional Court shall be appointed for nine years, beginning in each case from the day of their swearing in, and they may not be re-appointed.

At the expiry of their term, the constitutional judges shall leave office and the exercise of the functions thereof.

The Court shall elect from among its members, in accordance with the rules established by law, a President, who shall remain in office for three years and may be re-elected, respecting in all cases the expiry term for constitutional judges.

The office of constitutional judge shall be incompatible with membership of Parliament, of a Regional Council, the practice of the legal profession, and with every appointment and office indicated by law.

In impeachment procedures against the President of the Republic, in addition to the ordinary judges of the Court, there shall also be sixteen members chosen by lot from among a list of citizens having the qualification necessary for election to the Senate, which the Parliament prepares every nine years through election using the same procedures as those followed in appointing ordinary judges.

Article 137:

A constitutional law shall establish the conditions, forms, terms for proposing judgements on constitutional legitimacy, and guarantees on the independence of constitutional judges.

Ordinary laws shall establish the other provisions necessary for the constitution and the functioning of the Court. No appeals are allowed against the decision of the Constitutional Court.

Constitutional Law No. 1 of 9 February 1948: Provisions governing the review of constitutionality and guaranteeing the independence of the Constitutional Court¹⁷³

Section 3:

Judges of the Constitutional Court may only be removed or suspended from office by a decision of the same Court on the grounds of physical or civil incapacity or gross misconduct in the exercise of their office.

While in office, judges of the Constitutional Court shall enjoy the same immunities as members of both Houses of Parliament provided by Article 68(2) of the Constitution. The authorisation referred to in that Article shall be issued by the Constitutional Court.

Constitutional Law No. 1 of 11 March 1953: Supplementary Constitutional Provisions regarding the Constitutional Court

Section 7:

The judges of the Constitutional Court may be removed or suspended from office under Section 3 of Constitutional Law No. 1 of 9 February 1948 but only after a resolution to this effect has been adopted by the Constitutional Court with a two-thirds majority vote of the judges present.

Section 8:

Any judge of the Constitutional Court who fails to perform his functions for six consecutive months shall cease to hold office.

Constitutional Law No. 2 of 22 November 1967: Amendment to Article 135 of the Constitution and provisions relating to the Constitutional Court

Section 3:

The judges of the Constitutional Court appointed by Parliament are elected at a joint sitting of both Houses, by secret ballot, with a majority consisting of two thirds of the members of the Assembly. After the third ballot a majority consisting of three fifths of the members of the Assembly is sufficient.

Section 4:

The judges of the Constitutional Court appointed by the highest ordinary and administrative courts following the procedures laid down by statute shall be elected by majority vote, on condition that they obtain a number of votes in excess of one half of the number of the members of the panel.

If this majority is not obtained on the first ballot, on the following day a second ballot shall be held between the candidates obtaining the largest number of votes in the previous day's ballot, with twice as many candidates as there are positions to be filled. The candidates obtaining the highest number of votes are elected. In the event of a tie, the oldest candidate shall be elected or entered for a tie-breaking ballot.

South Africa

Constitution of the Republic of South Africa (1996)¹⁷⁴

Article 165: Judicial authority

- 1. The judicial authority of the Republic is vested in the courts.
- 2. The courts are independent and subject only to the Constitution and the law, which they must apply impartially and without fear, favour or prejudice.
- 3. No person or organ of state may interfere with the functioning of the courts.
- 4. Organs of state, through legislative and other measures, must assist and protect the courts to ensure the independence, impartiality, dignity, accessibility and effectiveness of the courts.

5. An order or decision issued by a court binds all persons to whom and organs of state to which it applies.

Article 167: Constitutional Court

1. The Constitutional Court consists of the Chief Justice of South Africa, the Deputy Chief Justice and nine other judges.

[Subs. (1) substituted by s. 11 of the Constitution Sixth Amendment Act of 2001]

- 2. A matter before the Constitutional Court must be heard by at least eight judges.
- 3. The Constitutional Court
- a. is the highest court of the Republic; and
- b. may decide
- 1. constitutional matters; and
- any other matter, if the Constitutional Court grants leave to appeal on the grounds that the matter raises an arguable point of law of general public importance which ought to be considered by that Court;
- c. makes the final decision whether a matter is within its jurisdiction.

[Subs. (3) substituted by s. 3 of the Constitution Seventeenth Amendment Act of 2013]

- 4. Only the Constitutional Court may:
- a. decide disputes between organs of state in the national or provincial sphere concerning the constitutional status, powers or functions of any of those organs of state;
- b. decide on the constitutionality of any parliamentary or provincial Bill, but may do so only in the circumstances anticipated in section 79 or 121;
- c. decide applications envisaged in section 80 or 122;
- d. decide on the constitutionality of any amendment to the Constitution;
- e. decide that Parliament or the President has failed to fulfil a constitutional obligation; or
- f. certify a provincial constitution in terms of section 144.
- 5. The Constitutional Court makes the final decision whether an Act of Parliament, a provincial Act or conduct of the President is constitutional, and must confirm any order of invalidity made by the Supreme Court of Appeal, the High Court of South Africa, or a court of similar status, before that order has any force.

[Subs. (5) substituted by s. 3 of the Constitution Seventeenth Amendment Act of 2013]

- 6. National legislation or the rules of the Constitutional Court must allow a person, when it is in the interests of justice and with leave of the Constitutional Court
- a. to bring a matter directly to the Constitutional Court; or
- b. to appeal directly to the Constitutional Court from any other court.
- 7. A constitutional matter includes any issue involving the interpretation, protection or enforcement of the Constitution.

Article 174: Appointment of judicial officers

- 1. Any appropriately qualified woman or man who is a fit and proper person may be appointed as a judicial officer. Any person to be appointed to the Constitutional Court must also be a South African citizen.
- 2. The need for the judiciary to reflect broadly the racial and gender composition of South Africa must be considered when judicial officers are appointed.
- 3. The President as head of the national executive, after consulting the Judicial Service Commission and the leaders of parties represented in the National Assembly, appoints the Chief Justice and the Deputy Chief Justice and, after consulting the Judicial Service Commission, appoints the President and Deputy President of the Supreme Court of Appeal.

[Subs. (3) substituted by s. 13 of the Constitution Sixth Amendment Act of 2001]

- 4. The other judges of the Constitutional Court are appointed by the President, as head of the national executive, after consulting the Chief Justice and the leaders of parties represented in the National Assembly, in accordance with the following procedure:
- a. The Judicial Service Commission must prepare a list of nominees with three names more than the number of appointments to be made, and submit the list to the President.
- b. The President may make appointments from the list, and must advise the Judicial Service Commission, with reasons, if any of the nominees are unacceptable and any appointment remains to be made.

- c. The Judicial Service Commission must supplement the list with further nominees and the President must make the remaining appointments from the supplemented list.
- [Subs. (4) substituted by s. 13 of the Constitution Sixth Amendment Act of 2001]
- 5. At all times, at least four members of the Constitutional Court must be persons who were judges at the time they were appointed to the Constitutional Court.
- 6. The President must appoint the judges of all other courts on the advice of the Judicial Service Commission.
- [...]

Article 177: Removal

- 1. A judge may be removed from office only if
- a. the Judicial Service Commission finds that the judge suffers from an incapacity, is grossly incompetent or is guilty of gross misconduct; and
- b. the National Assembly calls for that judge to be removed, by a resolution adopted with a supporting vote of at least two thirds of its members.
- 2. The President must remove a judge from office upon adoption of a resolution calling for that judge to be removed.
- 3. The President, on the advice of the Judicial Service Commission, may suspend a judge who is the subject of a procedure in terms of subsection (1).

Article 178: Judicial Service Commission

- 1. There is a Judicial Service Commission consisting of
- a. the Chief Justice, who presides at meetings of the Commission;
- b. the President of the Supreme Court of Appeal;

[Para. (b) substituted by s. 16 (a) of the Constitution Sixth Amendment Act of 2001]

- c. one Judge President designated by the Judges President;
- d. the Cabinet member responsible for the administration of justice, or an alternate designated by that Cabinet member;
- e. two practising advocates nominated from within the advocates' profession to represent the profession as a whole, and appointed by the President;

- f. two practising attorneys nominated from within the attorneys' profession to represent the profession as a whole, and appointed by the President;
- g. one teacher of law designated by teachers of law at South African universities;
- h. six persons designated by the National Assembly from among its members, at least three of whom must be members of opposition parties represented in the Assembly;
- i. four permanent delegates to the National Council of Provinces designated together by the Council with a supporting vote of at least six provinces;
- j. four persons designated by the President as head of the national executive, after consulting the leaders of all the parties in the National Assembly; and
- k. when considering matters relating to a specific Division of the High Court of South Africa, the Judge President of that Division and the Premier of the province concerned, or an alternate designated by each of them.

[Para. (k) substituted by s. 2 of the Constitution Second Amendment Act of 1998, s. 16 of the Constitution Sixth Amendment Act of 2001 and s. 10 of the Constitution Seventeenth Amendment Act of 2013]

[...]

Turkey

Constitution of the Republic of Turkey (7 Nov. 1982 (prior to 2010 amendment))¹⁷⁵

Article 146 – The Constitutional Court - Organization

The Constitutional Court shall be composed of eleven regular and four substitute members.

The President of the Republic shall appoint two regular and two substitute members from the High Court of Appeals, two regular and one substitute member from the Council of State, and one member each from the Military High Court of Appeals, the High Military Administrative Court and the Audit Court, three candidates being nominated for each vacant office by the Plenary Assemblies of each court from among their respective presidents and members, by an absolute majority of the total number of members; the President of the Republic shall also appoint one member from a list of three candidates nominated by the Higher Education Council from among members of the teaching staff of institutions of higher education who are not members of the Council, and three members and one substitute member from among senior administrative officers and lawyers.

To qualify for appointments as regular or substitute members of the Constitutional Court, members of the teaching staff of institutions of higher education, senior administrative officers and lawyers shall be required to be over the age of forty and to have completed their higher education, or to have served at least fifteen years as a member of the teaching staff of institutions of higher education or to have worked actually at least fifteen years in public service or to have practiced as a lawyer for at least fifteen years.

The Constitutional Court shall elect a President and Deputy President from among its regular members for a term of four years by secret ballot and by an absolute majority of the total number of members. They may be re-elected at the end of their term of office. The members of the Constitutional Courts shall not assume other official and private functions, besides their main functions.

Article 147: Termination of Membership

The members of the Constitutional Court shall retire on reaching the age of sixty-five. Membership in the Constitutional Court shall terminate automatically if a member is convicted of an offence requiring his dismissal from the judicial profession; it shall terminate by a decision of an absolute majority of the total number of members of the Constitutional Court if it is definitely established that he is unable to perform his duties on account of ill health.

Constitution of the Republic of Turkey (7 Nov. 1982, as amended 12 Sept. 2010)¹⁷⁶

Article 146- Constitutional Court formation:

The Constitutional Court shall be composed of seventeen members.

The Grand National Assembly of Turkey shall elect, by secret ballot, two members from among three candidates to be nominated by and from among the president and members of the Court of Accounts, for each vacant position, and one member from among three candidates nominated by the heads of the bar associations from among self-employed lawyers. In this election to be held in the Grand National Assembly of Turkey, for each vacant position, two-thirds majority of the total number of members shall be required for the first ballot, and absolute majority of total number of members shall be required for the second ballot. If an absolute majority cannot be obtained in the second ballot, a third ballot shall be held between the two candidates who have received the greatest number of votes in the second ballot; the member who receives the greatest number of votes in the third ballot shall be elected.

The President of the Republic shall appoint three members from High Court of Appeals, two members from Council of State, one member from the High Military Court of Appeals, and one member from the High Military Administrative Court from among three candidates to be nominated, for each vacant position, by their respective general assemblies, from among their presidents and members; three members, at least two of whom being law graduates, from among three candidates to be nominated for each vacant position by the Council of Higher Education from among members of the teaching staff who are not members of the Council, in the fields of law, economics and political sciences; four members from among high level executives, self-employed lawyers, first category judges and public prosecutors or rapporteurs of the Constitutional Court.

In the elections to be held in the respective general assemblies of the High Court of Appeals, Council of State, High Military Court of Appeals, High Military Administrative Court, the Court of Accounts and the Council of Higher Education for nominating candidates for membership of the Constitutional Court, three persons obtaining the greatest number of votes shall be considered to be nominated for each vacant position. In the elections to be held for the three candidates nominated by the heads of bar associations from among self-employed lawyers, three persons obtaining the greatest number of votes shall be considered to be nominated.

To qualify for appointments as members of the Constitutional Court, members of the teaching staff shall be required to possess the title of professor or associate professor;

lawyers shall be required to have practiced as a lawyer for at least twenty years; high level executives shall be required to have completed higher education and to have worked for at least twenty years in public service, and first category judges and public prosecutors with at least twenty years of work experience including their period of candidacy, provided that they all shall be over the age of forty-five.

The Constitutional Court shall elect a president and two deputy presidents from among its members for a term of four years by secret ballot and by an absolute majority of the total number of its members. Those whose term of office ends may be re-elected. The members of the Constitutional Court shall not assume other official and private duties, apart from their fundamental duties.

Article 147: Term of office of the members and termination of membership

The members of the Constitutional Court shall be elected for a term of twelve years. A member shall not be re-elected. The members of the Constitutional Court shall retire when they are over the age of sixty-five. The appointment of the members to another office whose term of office expires prior to their mandatory age of retirement and matters regarding their personnel status shall be laid down in law.

Membership in the Constitutional Court shall terminate automatically if a member is convicted of an offence requiring his/her dismissal from the judicial profession, and by a decision of an absolute majority of the total number of members of the Constitutional Court if it is definitely established that he/she is unable to perform his/her duties on account of ill-health.

Endnotes

¹ Grote, 'Models of Institutional Control', p. 221; Venice Commission, 'List of Constitutional Courts'.

² Iraq's first constitution established the Iraqi High Court, which was composed of nine justices, five of whom, including the president, came from the Senate; the remaining four were senior judges. Brown, *Constitutions in a Non-Constitutional World*, pp. 148–9, 219 note 12.

³ Article 44 of the TAL and Law No. 30/2005.

⁴ International Crisis Group, 'Trial by Error', pp. 2, 9–15.

⁵ Constitution of Tunisia, 1959, Articles 66, 72–5; draft Constitution of Tunisia, June

2013, Article 115; Tunisian Decree Law No. 2011-14, 23 March 2011, Article 2.

⁶ Royal Hashemite Court of Jordan, 'Jordan Enacts New Political Parties'; Barakat,

'Jordan: Constitutional Law Court Newly Established in Jordan'.

⁷ Constitution of Morocco, 1996, Articles 78, 80, 81; Constitution of Morocco, 2011,

Article 130; Economist, 'Morocco's Referendum: A Very Small Step'.

⁸ Max Planck Institute, 'Bahrain'; Bahrain Decree Law 38-2012.

⁹ Brown, *Constitutions in a Nonconstitutional World*, pp. 148–51; Kuwait Law No. 14 of 1973.

¹⁰ Stone Sweet, 'Constitutional Courts', p. 817.

¹¹ Issacharoff, 'Democratic Hedging', pp. 963–4.

¹² Brown, The Rule of Law in the Arab World, p. 9.

¹³ Macdonald and Kong, 'Judicial Independence as a Constitutional Virtue', p. 845;

Moustafa, The Struggle for Constitutional Power, pp. 4-6.

¹⁴ Macdonald and Kong, 'Judicial Independence as a Constitutional Virtue', pp. 844–5.

¹⁵ See Ginsburg, Judicial Review in New Democracies.

¹⁶ Kommers, *Constitutional Jurisprudence*, pp. 186–91.

¹⁷ Kirkpatrick, 'Blow to Transition'; Moustafa, *The Struggle for Constitutional Power*, pp. 162–3.

¹⁸ Moustafa, *The Struggle for Constitutional Power*, 103; *BBC News*, 'Q & A: Egypt's Supreme Court Rulings'; Greene, 'Ahmed Shafik'.

¹⁹ Issacharoff, 'Democratic Hedging', pp. 996–8.

²⁰ Justice Alliance of South Africa v. President of Republic of South Africa (2011) CCT 53/11.

²¹ Iraqi Federal Supreme Court Decision 43/2010; Hamoudi, 'Decision 88'.

²² Italian Constitutional Court Judgement No. 1/1956; Volcansek, 'Political Power and Judicial Review in Italy', p. 495; Kommers, *The Constitutional Jurisprudence*, p. 568 note 40; BVerfGE 3, p. 58 [Beamtenverhältnisse Case].

²³ Issacharoff, 'Democratic Hedging', pp. 994–5; Certification Decision of the Constitution of the Republic of South Africa (1996) CCT 23/96.

- ²⁴ Harding, Leyland, and Groppi, 'Constitutional Courts', p. 15.
- ²⁵ Hedling, 'The Design of the Judicial Branch', p. 2.
- ²⁶ Brown, *Constitutions in a Nonconstitutional World*, p. 151; Hussain, *A Critical Study*, pp. 346–7.

²⁷ Ottaway, 'The New Moroccan Constitution'; Constitution of Morocco, 2011, article 130.

²⁸ Brown, Constitutions in a Nonconstitutional World, p. 150.

²⁹ Constitution of Jordan, 2011, Article 58; *Jordan Times*, 'Profiles of Constitutional Court Members'.

³⁰ Moustafa, *The Struggle for Constitutional Power*, *pp.* 198–9; El Deeb, 'Former Judge Challenges Egypt's Constitution'; Albrecht, 'Egypt's 2012 Constitution', p. 4.

³¹ Harding, Leyland and Groppi, 'Constitutional Courts', p. 17. Regarding simple majority requirements, see, for example, the Polish Constitutional Tribunal, where appointment power is exclusively vested in Parliament but the legislature decides by simple majority. In Poland, the Sejm (lower house of Parliament) appoints judges by simple majority in the presence of at least half of the total number of deputies (Constitutional Tribunal Act of Poland, 1997, Article 5, para. 4). The arrangement is beneficial in the sense that it allows for faster appointments to be made. The main weakness of this type of system is that it can result in extreme politicization of the appointment process, in which the current parliamentary majority largely controls nominations and the role of the opposition is essentially meaningless. Although the opposition has the power to nominate candidates, those candidates have little chance of success. See Sadurski, 'Twenty Years after the Transition', pp. 5–6.

³² Schoenberger, 'The Establishment of Judicial Review', pp. 8–9; Kommers, *The Constitutional Jurisprudence*, pp. 6–8.

³³ Geck, *Wahl und Amtsrecht*, pp. 13–14; Kommers, *Judicial Politics in West Germany*, pp. 79–82; Kommers, *The Constitutional Jurisprudence*, pp. 8–9.

³⁴ Nicolauß, 'Der Parlamentarische Rat', p. 128; Pieper, *Verfassungsrichterwahlen*, p. 37; Kommers, *Judicial Politics in West Germany*, pp. 79–80.

³⁵ Vanberg, *The Politics of Constitutional Review in Germany*, p. 63; Kommers, *Judicial Politics in West Germany*, p. 88.

³⁶ Kommers, *The Constitutional Jurisprudence*, pp. 16–18.

³⁷ Kommers, 'Autonomy versus Accountability', pp. 148–9; Geck, *Wahl und Amtsrecht*, p. 22.

³⁸ Kommers, *Judicial Politics in West Germany*, p. 114; Kommers, 'Autonomy versus Accountability', p. 150.

³⁹ Kommers, *Judicial Politics in West Germany*, p. 117; Kommers, *The Constitutional Jurisprudence*, p. 22.

⁴⁰ Kommers, 'Autonomy versus Accountability', p. 149; Volcansek, 'Appointing Judges,' p. 378; Hönnige, *Verfassungsgericht*, p. 170. Die LINKE was founded in 2007 when two other parties, the WASG and PDS, merged. It has not yet formed part of a governing coalition on the federal level. However, Die LINKE is represented in the federal Parliament and in many state Parliaments, especially in Eastern Germany.

⁴¹ Schlaich and Korioth, *Das Bundesverfassungsgericht*, p. 30; Pasquino and Ferejohn, 'Constitutional Adjudication', p. 1678.

⁴² Adenauer once called the Court 'Germany's dictator' (Willoweit, *Deutsche Verfassungsgeschichte*, p. 368), whereas Dehler claimed that one of its judgements was void (Lembcke, 'Das Bundesverfassungsgericht', pp. 155–7). Kommers, *Judicial Politics in West Germany*, p. 128; BVerfGE 1, p. 281; BVerfGE 2, p. 143; BVerfGE 2, p. 79; BVerfGE 12, p. 205.

⁴³ Vanberg, *The Politics of Constitutional Review in Germany*, pp. 67–8.
 ⁴⁴ Ibid. pp. 67–70.

⁴⁵ Ibid. pp. 70–71; Kommers, Judicial Politics in West Germany, p. 283.

⁴⁶ Vanberg, *The Politics of Constitutional Review in Germany*, pp. 71–2; Kommers, *Judicial Politics in West Germany*, pp. 283–4.

⁴⁷ Vanberg, *The Politics of Constitutional Review in Germany*, pp. 71–3; Kommers, *Judicial Politics in West Germany*, p. 284.

⁴⁸ Vanberg, *The Politics of Constitutional Review in Germany*, pp. 73–6; Kommers, *Judicial Politics in West Germany*, p. 286.

⁴⁹ Kommers, 'Autonomy versus Accountability', p. 148; Kommers, *Judicial Politics in West Germany*, pp. 132–5, 157, note 22.

⁵⁰ Kommers, 'Autonomy versus Accountability', pp. 148–9; Kommers, *Judicial Politics in West Germany*, pp. 135–8, Lembcke, 'Das Bundesverfassungsgericht', pp. 158–9; Billing, *Das Problem*, pp. 195–206, 214.

⁵¹ For examples of controversial individual rights cases, see the 'Sitzblockaden' case (BVerfGE 73, p. 206), in which the FCC ruled in favour of individuals' rights in sit-in demonstrations. In the 'Soldaten-sind-Mörder' case (BVerfGE 93, p. 266) the FCC upheld a constitutional complaint that challenged a law limiting the freedom of expression in criticizing soldiers. Similarly, in the 'Kruzifix' case (BVerfGE 93, p. 1) the FCC found that the attachment of a cross or crucifix in the classrooms of compulsory state schools violates the fundamental rights enumerated in Article 4, para. 1 of the Basic Law. Benda and Klein, Verfassungsprozessrecht, p. 5; Voigt, 'Das Bundesverfassungsgericht', p. 71. ⁵² Pieper, Verfassungsrichterwahlen, p. 13; Hönnige, Verfassungsgericht, p. 172. For a press review mentioning these two exceptional cases, see Süddeutsche Zeitung, 'CDU verhindert Wechsel am Bundesverfassungsgericht'. ⁵³ Kranenpohl, *Hinter dem Schleier*, p. 28. ⁵⁴ Landfried, 'Die Wahl der Bundesverfassungsrichter', p. 233; Benda and Klein, Verfassungsprozessrecht, pp. 57-8; BVerfG, 2 BvC 2/10, 19.6.2012. ⁵⁵ Seibert-Fohr, 'Judicial Independence in Germany', pp. 516–17. ⁵⁶ Garoupa and Ginsburg, 'Guarding the Guardians', p. 106. ⁵⁷ Wesson and du Plessis, The Transformation of the Judiciary', pp. 3–4; Corder, 'Appointment, Discipline and Removal', p. 98; Coniglio, 'Methods of Judicial Decision-Making', p. 504; Currie, 'The Judiciary and Freedom of Speech in South Africa', pp. 221–2; Constitution of South Africa, 1983, Article 34. ⁵⁸ Wesson and du Plessis, The Transformation of the Judiciary', pp. 4–5. ⁵⁹ Spitz and Chaskalson, *The Politics of Transition*, p. 191 [quoted in text]. ⁶⁰ Andrews, 'The Judiciary in South Africa', p. 478 [quoted in text]; Spitz and Chaskalson, The Politics of Transition, pp. 201-9. ⁶¹ Spitz and Chaskalson, *The Politics of Transition*, pp. 201–5. ⁶² Gloppen, South Africa: The Battle over the Constitution, p. 229; Malleson, 'Assessing the Performance', p. 38; Spitz and Chaskalson, The Politics of Transition, pp. 201-7. ⁶³ Malleson, 'Assessing the Performance', pp. 38–9; Spitz and Chaskalson, The Politics of Transition, p. 208. Note that there have been some amendments to the appointment system, but by and large the process remains the same. ⁶⁴ Act 9 of 1994 (Judicial Service Commission); Judges' Remuneration and Conditions of Employment Act, Act No. 47 of 2001 (South Africa). ⁶⁵ There are permanent and special delegates to the National Council of Provinces (Constitution of South Africa, 1996, Articles 60–2). Andrews, 'The Judiciary in South Africa', p. 480 [quoted in text]; Corder, 'Judicial Authority in a Changing South Africa', p. 197. Attorneys are generalists; they provide legal advice to clients on a variety of issues and can appear in lower courts. Advocates are specialists who can appear in all courts and are usually hired by attorneys when their specialty (i.e. in tax litigation) is required.

⁶⁶ Andrews, 'The South African Judicial Appointments Process', p. 568 [quoted in text]; Andrews, 'The Judiciary in South Africa', p. 481.

⁶⁷ Andrews, 'The South African Judicial Appointments Process', pp. 567–8 [quoted in text].

⁷⁰ Andrews, 'The Judiciary in South Africa', pp. 483–5.

⁷¹ Minister of Health and Others v. Treatment Action Campaign and Others (2002) CCT 8/02.

⁷² Ibid.; McGreal, 'Mandela Attacks Mbeki's AIDS Policy'.

⁷³ Hugh Glenister v. President of the Republic of South Africa and Others (2011) CCT 48/10; Berning and Montesh, 'Countering Corruption in South Africa'; De Vos, 'On the Disbanding of the Scorpions'.

⁷⁴ Hugh Glenister v. President of the Republic of South Africa and Others (2011) CCT 48/10.

⁷⁵ See Corder, 'Appointment, Discipline and Removal', p. 116, on the role played by civil society and the media in protecting judicial independence; Malleson, 'Assessing the Performance', p. 39; African National Congress, 'Statement of the National Executive Committee'; Kruger, 'You Said it, Mantashe'; Quintal, 'Judges to be Given Chance to Discuss Reforms'; McKaiser, 'Democracy and its Malcontents'; Msomi, 'Battle of the Constitution'; Stone, 'ANC Still has Review of Judiciary's Powers in its Sights'.

⁷⁶ For more information on support for the Court within South Africa, and the Court's complex relationship with the political branches, see Roux, *The Politics of Principle*; Gibson and Caldeira, 'Defenders of Democracy'; Gibson, 'The Evolving Legitimacy of the South African Constitutional Court'.

⁷⁷ Parliament of the Republic of South Africa, 'State of Parties in the National Assembly', 'State of Parties in the National Council of Provinces'.

⁷⁸ Malleson, 'Assessing the Performance', pp. 38–9.

⁷⁹ Zille, 'Zille Disappointed over Zuma's Final JSC Choices'; De Vos, 'ANC Rrying to "load" the JSC'; Corder, 'Appointment, Discipline and Removal', p. 100, note 19.
 ⁸⁰ Justice Alliance of South Africa v. President of Republic of South Africa (2011) CCT 53/11.

⁸¹ For other news articles referring to the Mogoeng controversy, see Rawoot,

'Mogoeng's Assault on Women's Rights'; Mofokeng, Matlala and Hawker, 'Moseneke in New Clash with Zuma'; Smith, 'Chief Justice: How Zuma Got it Wrong'; Klug, *The Constitution of South Africa*, pp. 248–9.

⁸² Malleson, 'Assessing the Performance', pp. 40–1, 45.

⁶⁸ Madhuku, 'Constitutional Protection of the Independence of the Judiciary', p. 236; Corder, 'Judicial Authority in a Changing South Africa', p. 198.

⁶⁹ Constitution of South Africa, 1996, Article 174; Corder, 'Judicial Authority in a Changing South Africa', p. 198.

⁸³ Wesson and du Plessis, 'The Transformation of the Judiciary', p. 7; Malleson,

'Assessing the Performance', pp. 44–7, referring to the 'Guidelines for Questioning Candidates for Nomination to the Constitutional Court' (1994).

⁸⁴ Klug, 'South Africa's Constitutional Court', p. 285; Ramphele and Dawood, 'South African Courts at Risk'; Ferreira, 'South Africa: Minister Denies Manipulation of Judicial Service Commission'.

⁸⁵ Hamad, 'The Politics of Judicial Selection in Egypt', p. 267 [quoting Moustafa in text]; Moustafa, 'Law versus the State', pp. 885–93.

⁸⁶ Moustafa, 'Law versus the State', pp. 893–4.

⁸⁷ Moustafa, *The Struggle for Constitutional Power*, pp. 198–201; Hamad, 'The Politics of Judicial Selection in Egypt', pp. 271–4.

⁸⁸ Brown, *Egypt's Judges in a Revolutionary Age*, p. 12.

⁸⁹ Associated Press, 'Egyptian Judges Back Out of Reform Conference with Morsi'.

⁹⁰ Brown, *Egypt's Judges in a Revolutionary Age*, p. 12; Brown, 'Quick Reactions'; Brown and Lombardi, 'The Supreme Constitutional Court', p. 441 note 7.

⁹¹ Moustafa, *The Struggle for Constitutional Power*, pp. 140–2, 182–7; Hamad, 'The Politics of Judicial Selection in Egypt', p. 272.

⁹² Moustafa, *The Struggle for Constitutional Power*, pp. 104–5 [quoted in text], p. 181.

⁹³ Hamad, 'The Politics of Judicial Selection in Egypt', pp. 272–3 [quoted in text];

Moustafa, The Struggle for Constitutional Power, pp. 188-91.

⁹⁴ Moustafa, *The Struggle for Constitutional Power*, pp. 198–9.

⁹⁵ Moustafa, The Struggle for Constitutional Power, pp. 199–201.

⁹⁶ Moustafa, *The Struggle for Constitutional Power*, p. 199; Hamad, 'The Politics of Judicial Selection in Egypt', p. 273.

⁹⁷ Moustafa, The Struggle for Constitutional Power, pp. 205–8.

⁹⁸ Moustafa, *The Struggle for Constitutional Power*, pp. 205–8.

⁹⁹ Moustafa, *The Struggle for Constitutional Power*, pp. 210–13 [quoted in text]; Brown,

'Why Did the Egyptian Regime Appoint a New Chief Justice?'; Stilt, 'Islam is the Solution', p. 83.

¹⁰⁰ Radwan, 'Egypt's Return of the Judiciary'; Brown, *Egypt's Judges in a Revolutionary Age*, p. 12.

¹⁰¹ Kirkpatrick, 'Blow to Transition'; *BBC News*, 'Egypt's Islamist Parties Win Elections to Parliament'.

¹⁰² Brown, 'Egypt's Judiciary'; El-Din, 'Egypt's Shura Council'; Brown, 'Cairo's Judicial Coup'.

¹⁰³ Molloy, 'Why the Reduction in SCC Justices?'.

¹⁰⁴ Mahmoud, 'Judicial System in Iraq', pp. 10–11.

¹⁰⁵ Judicial Service Act 27/1945; Judicial Service Act No. 58/1956; Judicature Act No. 26/1963; Mahmoud, 'Judicial System in Iraq', pp. 18–21; Posner and McKay, 'The Iraqi Justice System', p. 14.

¹⁰⁶ Posner and McKay, 'The Iraqi Justice System', p. 15; Mahmoud, 'Judicial System in Iraq', pp. 21–2.

¹⁰⁷ Diamond, *Squandered Victory*, p. 149; Posner and McKay, 'The Iraqi Justice System', p. 15.

¹⁰⁸ Diamond, *Squandered Victory*, pp. 149–50; Roberts, 'Iraq's Burgeoning Judiciary', p.3.

¹⁰⁹ Al-Ali, *The Struggle for Iraq's Future*; Diamond, *Squandered Victory*, pp. 142–3.

¹¹⁰ Diamond, *Squandered Victory*, pp. 148–50; Feldman and Martinez, 'Constitutional Politics and Text in the New Iraq', p. 917.

¹¹¹ Feldman and Martinez, 'Constitutional Politics and Text in the New Iraq', 917; Al-Ali, *The Struggle for Iraq's Future*.

¹¹² Al-Ali, The Struggle for Iraq's Future.

¹¹³ Ibid.

¹¹⁴ Feldman and Martinez, 'Constitutional Politics and Text in the New Iraq', p. 918;

Hamoudi, Negotiating in Civil Conflict, pp. 187–9; Al-Ali, The Struggle for Iraq's Future; Sadah, 'Iraqi Judicial Reforms'; Wing, 'Iraq's Federal Court'.

¹¹⁵ Sadah, 'Iraqi Judicial Reforms'; Wing, 'Iraq's Federal Court'.

¹¹⁶ Feldman and Martinez, 'Constitutional Politics and Text in the New Iraq', p. 917; Diamond, *Squandered Victory*, pp. 149–50.

¹¹⁷ Hamoudi, *Negotiating in Civil Conflict*, p. 187; Visser, 'The Law on the Federal Supreme Court'.

¹¹⁸ Hamoudi, 'The Will of the (Iraqi) People', pp. 46–51.

¹¹⁹ FSC Decision No. 25/2010; Trumbull and Martin, 'Elections and Government Formation in Iraq', pp. 369–70.

¹²⁰ Trumbull and Martin, 'Elections and Government Formation in Iraq', pp. 370–1; Visser, 'Iraqiyya Challenges the Jurisdiction'; Hamoudi, 'The Will of the (Iraqi) People', pp. 48–50.

¹²¹ Sly, 'Iraq's High Court'.

¹²² FSC Decision No. 88/2010.

¹²³ Visser, 'The Law on the Federal Supreme Court'.

¹²⁴ Bâli, 'Courts and Constitutional Transition', p. 669; Shambayati and Kirdis, 'In Pursuit of "Contemporary Civilization", pp. 771–3.

¹²⁵ Shambayati and Kirdis, 'In Pursuit of "Contemporary Civilization", pp. 773–5; Bâli, 'Courts and Constitutional Transition', pp. 670–1. The actions of the Turkish military

align with the explanation offered by Ran Hirschl for the choice by elite actors to constitutionalize certain rights and political principles and establish judicial review, as a means of preserving their own power and values. See Hirschl, *Towards Juristocracy*. ¹²⁶ Bâli, 'Courts and Constitutional Transition', p. 694.

¹²⁷ Kogacioglu, 'Progress, Unity and Democracy', p. 440. See also Articles 68 and 69 of the 1982 Turkish Constitution:

Article 68, para. 4: 'The statutes and programs, as well as the activities of political parties shall not be contrary to the independence of the State, its indivisible integrity with its territory and nation, human rights, the principles of equality and rule of law, sovereignty of the nation, the principles of the democratic and secular republic; they shall not aim to promote or establish class or group dictatorship or dictatorship of any kind, nor shall they incite citizens to crime.'

Article 69, para. 4, 5: 'The dissolution of political parties shall be decided finally by the Constitutional Court after the filing of a suit by the office of the Chief Public Prosecutor of the High Court of Appeals. The permanent dissolution of a political party shall be decided when it is established that the statute and program of the political party violate the provisions of the fourth paragraph of Article 68.'

¹²⁸ Shambayati and Kirdis, 'In Pursuit of "Contemporary Civilization", p. 776; Bâli,'Courts and Constitutional Transition', p. 672.

¹²⁹ See Bâli, 'Courts and Constitutional Transition', pp. 674–7 for a full account of the TCC case and the Court's reasoning.

¹³⁰ Bâli, 'Perils of Judicial Independence', pp. 250–2.

¹³¹ Özpek, 'Constitution-making in Turkey after the 2011 Elections', pp. 153–5.

¹³² Bâli, 'Courts and Constitutional Transition', p. 694.

¹³³ Bâli, 'Perils of Judicial Independence', pp. 303–4.

¹³⁴ Arato and Goldfarb, 'Politically Weighted Courts in Turkey'; Arato, 'The Constitutional Reform Proposal of the Turkish Government'; Bâli, 'Perils of Judicial Independence', pp. 296–302; *The Economist*, 'Turkey's Constitutional Referendum: Erdogan Pulls it Off'.

¹³⁵ Bâli, 'Courts and Constitutional Transition', p. 694.

¹³⁶ Hurriyet Daily News, 'Court Clears Way for Gül to Re-run Top Post' [quoted in

text]; Arato and Goldfarb, 'Politically Weighted Courts in Turkey'; Özpek,

'Constitution-making in Turkey after the 2011 Elections', p. 155.

¹³⁷ Volcansek, *Constitutional Politics*, pp. 14–15.

¹³⁸ Pasquino, 'The Debates of the Italian Constituent Assembly,' p. 109; Volcansek, *Constitutional Politics*, pp. 15–17; Volcansek, 'Bargaining Constitutional Design', p. 286.
¹³⁹ Volcansek, 'Bargaining Constitutional Design', p. 285; Mandel, 'Legal Politics Italian Style', p. 266.

¹⁴⁰ Volcansek, *Constitutional Politics*, pp. 17–18 [quoted in text]; Volcansek, 'Bargaining Constitutional Design', p. 288; Pasquino, 'The Debates of the Italian Constituent Assembly,' p. 111.

¹⁴¹ The term limit was originally 12 years. When asked why they decreased the length of the term, former President of the Italian Constitutional Court Antonio Baldassarre replied, 'Ironically, I may say that many people have expectations of becoming justices in the Court so nine years is better than twelve years. I almost joke, but just almost.' See Baldassarre, 'Structure and Organization of the Constitutional Court of Italy', p. 658.
¹⁴² Law No. 87, 11 March 1953; Constitutional Law No. 2, 22 Nov. 1967; Volcansek, *Constitutional Politics*, p. 21; Mandel, 'Legal Politics Italian Style', p. 266.

¹⁴³ Constitutional Law No. 2, 22 Nov. 1967; Volcansek, 'Bargaining Constitutional Design', pp. 291–2.

¹⁴⁴ Volcansek, *Constitutional Politics*, pp. 22–3; Volcansek, 'Bargaining Constitutional Design', p. 288.

¹⁴⁵ Volcansek, *Constitutional Politics*, pp. 21–3 [quoted in text]; Pasquino and Ferejohn, 'Constitutional Adjudication', pp. 1681–2.

¹⁴⁶ Pasquino and Ferejohn, 'Constitutional Adjudication', p. 1682; Volcansek, *Constitutional Politics*, p. 23; Pellegrina and Garoupa, 'Choosing Between the Government and the Regions', pp. 3–4.

¹⁴⁷ Mandel, 'Legal Politics Italian Style', p. 267; Volcansek, 'Political Power', pp. 499– 505.

¹⁴⁸ Judgement No. 1/1956 (Constitutional Court of Italy); Volcansek, *Constitutional Politics*, p. 19 [quoted in text]; Volcansek, 'Political Power', p. 495; Mandel, 'Legal Politics Italian Style', p. 267.

¹⁴⁹ Volcansek, 'Political Power', p. 507.

¹⁵⁰ Peri, 'Judicial Independence v. Judicial Accountability', pp. 17–18.

¹⁵¹ The first Constitutional Court ruling was No. 24/2004, which struck down Law No. 140 of 20 June 2003. The second was ruling No. 262/2009, which struck down Law No. 124 of 23 July 2008. Hooper, 'Italian Court Waters down Berlusconi Immunity Law'; Squires, 'Silvio Berlusconi to Reform Italian Courts to Stop "witch Hunt"; *The Economist*, 'Injudicious'; Zebley, 'Italy PM Berlusconi Pledges Judicial Reform while Awaiting Trial'. ¹⁵² Sadurski, 'Twenty Years After the Transition', p. 4; Volcansek, *Constitutional Politics*,

p. 24: 'The pluralistic composition of the Court resulting from the party formulas has likely moderated some individual inclinations, and shared interests in maintaining intracourt harmony and limiting workloads emerge as goals that can transcend ideological divisions.'

¹⁵³ Bâli, 'Perils of Judicial Independence', pp. 301–3; European Commission, 'Turkey 2010 Progress Report', p. 75.

¹⁵⁴ Law 48 of 1979 Governing the Operations of the Supreme Constitutional Court of Egypt (SCC Law); *Associated Press*, 'Egyptian Judges Back Out of Reform Conference with Morsi'.

¹⁵⁵ 'Members of the Supreme Court' refers to members of the judicial body created in 1969, which the Supreme Constitutional Court replaced in 1979. See Moustafa, *The Struggle for Constitutional Power*, pp. 65–78.

¹⁵⁶ Federal Constitutional Court Act (1951) (Germany) § 2–4.

¹⁵⁷ Iraq has not yet specified the qualifications for judges serving on the FSC. The data in this table is taken from Iraq's Judicial Organization Law (Law No. 160 of 1979), Article 36, which sets out the requirements for judges throughout the judiciary, and from al-Saedi, 'Regime Change', pp. 6–7.

¹⁵⁸ Constitution of Italy, 1948, Article 135.

¹⁵⁹ Constitution of South Africa, 1996, Articles 174, 176.

¹⁶⁰ Judges' Remuneration and Conditions of Employment Act, Act No. 47 of 2001 (South Africa).

¹⁶¹ Constitution of Turkey (amended 2010), Articles 146–7.

¹⁶² Law 48/1979, Article 19.

¹⁶³ Seibert-Fohr, 'Judicial Independence in Germany', p. 513.

¹⁶⁴ Constitutional Law No. 1 § 7, 11 March 1953; Constitutional Law No. 1 § 3, 9 February 1948.

¹⁶⁵ Translation available online:

http://www.sis.gov.eg/En/Templates/Articles/tmpArticles.aspx?CatID=208.

¹⁶⁶ Translation available at <http://www.constitutionnet.org/vl/item/new-constitutionarab-republic-egypt-approved-30-nov-2012>.

¹⁶⁷ Translation by Awad al-Murr, reprinted in Boyle and Sherif, *Human Rights and Democracy*.

¹⁶⁸ Translation available at <http://www.gesetze-im-internet.de/englisch_gg/>.

¹⁶⁹ Translation available at <http://www.iuscomp.org/gla/statutes/BVerfGG.htm#2>.

¹⁷⁰ Available at <http://www.refworld.org/docid/45263d612.html>

¹⁷¹ Translation available at <http://www.refworld.org/docid/454f50804.html>

<http://www.senato.it/documenti/repository/istituzione/costituzione_inglese.pdf>

¹⁷³ The three pieces of supplemental legislation listed here are available at <<u>http://www.codices.coe.int/NXT/gateway.dll/CODICES/laws/eng/eur/ita?fn=docum</u>ent-frame.htm\$f=templates\$3.0>.

¹⁷⁴ Available at <http://www.constitutionalcourt.org.za/site/theconstitution/english-2013.pdf>.

¹⁷⁵ Translation available at <http://www.ilo.org/wcmsp5/groups/public/---ed_protect/---protrav/---ilo_aids/documents/legaldocument/wcms_127495.pdf>.

¹⁷⁶ Translation available at <http://global.tbmm.gov.tr/docs/constitution_en.pdf>.

¹⁷² Translation available at

The Arab region is experiencing an unprecedented moment of constitutional transition. Many countries in the region are reforming their systems of constitutional judicial review as a way of signaling the government's commitment to the rule of law, including by establishing constitutional courts. This report analyses the critical question of how to select the members of a constitutional court. It presents four models with a particular emphasis on how effectively those models foster a sense of political investment in the court on the part of multiple political actors. The report also presents a comparison of the qualifications required for appointment to the constitutional court, and the rules for removing constitutional court judges.

The Center for Constitutional Transitions at NYU Law and the International Institute for Democracy and Electoral Assistance are publishing a series of research reports on issues in constitutional design that have arisen in the Arab region. The reports address constitutional court appointment mechanisms, semi-presidentialism as power-sharing, political party finance regulation, anti-corruption efforts, decentralization in unitary states, and ownership and management of oil and gas resources. These reports are designed for use in support of constitution building activities in the Arab region.

> The Center for Constitutional Transitions at NYU Law 139 MacDougal Street New York, NY 10012, USA Email: constitutional.transitions@nyu.edu Website: constitutionaltransitions.org

International IDEA Strömsborg SE - 103 34 Stockholm Sweden Tel: +46 8 698 37 00 Fax: +46 8 20 24 22 Email: info@idea.int Website: www.idea.int